UC Berkeley UC Berkeley Electronic Theses and Dissertations

Title

Dualities and Topological Field Theories from Twisted Geometries

Permalink

https://escholarship.org/uc/item/5t69807z

Author Markov, Ruza

Publication Date 2012

Peer reviewed|Thesis/dissertation

Dualities and Topological Field Theories from Twisted Geometries by

Ruza Markov

A dissertation submitted in partial satisfaction of the

requirements for the degree of

Doctor of Philosophy

in

Physics

in the

Graduate Division

of the

University of California, Berkeley

Committee in charge:

Professor Ori J. Ganor, Chair Professor Petr Horava Professor Michael Hutchings

Spring 2012

Dualities and Topological Field Theories from Twisted Geometries

Copyright 2012 by Ruza Markov

Abstract

Dualities and Topological Field Theories from Twisted Geometries

by

Ruza Markov

Doctor of Philosophy in Physics

University of California, Berkeley

Professor Ori J. Ganor, Chair

I will present three studies of string theory on twisted geometries.

In the first calculation included in this dissertation we use gauge/gravity duality to study the Coulomb branch of an unusual type of nonlocal field theory, called Puff Field Theory. On the gravity side, this theory is given in terms of D3-branes in type IIB string theory with a geometric twist. While the field theory description, available in the IR limit, is a deformation of Yang-Mills gauge theory by an order seven operator which we here compute.

In the rest of this disertation we explore $\mathcal{N} = 4$ super Yang-Mills (SYM) theory compactied on a circle with S-duality and R-symmetry twists that preserve $\mathcal{N} = 6$ supersymmetry in 2 + 1D. It was shown that abelian theory on a flat manifold gives Chern-Simons theory in the low-energy limit and here we are interested in the nonabelian counterpart. To that end, we introduce external static supersymmetric quark and anti-quark sources into the theory and calculate the Witten Index of the resulting Hilbert space of ground states on a two-torus. Using these results we compute the action of simple Wilson loops on the Hilbert space of ground states without sources. In some cases we find disagreement between our results for the Wilson loop eigenvalues and previous conjectures about a connection with Chern-Simons theory.

The last result discussed in this dissertation demonstrates a connection between gravitational Chern-Simons theory and $\mathcal{N} = 4$ four-dimensional SYM theory compactified on a circle twisted by S-duality where the remaining three-manifold is not flat starting with the explicit geometric realization of S-duality in terms of (2, 0) theory.

To the universe, for being so wonderful, and twisted.

Contents

Li	st of Figures	v
Li	st of Tables	vii
Ι	Insights from Geometric Twists	1
1	An Introduction to Twists	2
2	Examples2.1Melvin twist2.2S-Duality twist	4 5 7
3	Organization	11
II a 4	A Study of the Coulomb branch of Puff Field theory in Melvin Universe Setup	13
5	Result	17
6	Appendix	19

Duality Twist		28
7	Introduction	29
8	Review of the S-duality twist and its type-IIA dual8.1Definition of Tr-S8.2Type-IIA dual8.3 \mathbb{Z}_k symmetries8.4Is Tr-S a topological theory?	33 33 34 37 38
9	Static charges and their type-IIA dual description9.1Charges as endpoints of type-IIB strings	42 43 45 46 47 51 54 55 57 58 59
10	U(1) gauge group 10.1 A quark and anti-quark pair $(m = 1)$	61 63 64 66
11	U(n) gauge group11.1 Examples of U(2) sectors and states11.2 Counting fermionic zero modes11.3 The Witten Index — results11.4 Wilson loop operators and their eigenvalues11.5 Consistency checks11.6 Comparison with Chern-Simons theory	69 69 74 81 84 85 86
12	2 Discussion	95
13	 Appendices 13.1 Supersymmetry and fermionic zero modes — details	98 98 103 103

iii

13.2.2 A generating function for the number of sectors and Fibonacci	
numbers \ldots	104
13.3 An alternative set of boundary conditions using tilted D3-branes	108

IV Abelian Self-duality and Gravitational Chern-Simons Theory 111

14	Motivation 14.1 A Review of (2, 0) Theory 14.2 (2, 0) Theory on a Twisted Geometry	112 113 115
15	Calculation15.1 Low-Energy Expansion of Gravitational Chern-Simons Theory15.2 Field theory loop calculation	117 117 118
16	Interpretation	122
Bi	bliography	124

List of Figures

2.1	Melvin twist in 3 dimensions. a) The construction introduces a rotation in the z-plane by an angle θ for every loop completed in the compact ζ direction. b) A twisted string state in this geometry feels a force toward the origin of the z-plane	5
9.1	External quark and anti-quark sources are realized as endpoints of fundamental strings. At low-energy, the strings are described by free $1 + 1D$ fields $X^{\mu}_{\mathfrak{c}}(x_9, t), \psi_{\mathfrak{c}}(x_9, t)$ and the low-energy modes of the compact interacting system of D3-branes and charges are described by	
	periodic variables $\mathbf{p}^{a}(t), \mathbf{q}^{\alpha}(t)$.	44
9.2	The type-IIA configuration following the U-duality transformation of	
	Table 8.1. The m pairs of type-IIB open strings become m continuous	
	D2-branes. The n D3-branes become n fundamental strings, which in	
	the presence of the D2-branes can break up into open strings. At least	
	one pair of open strings must be attached to each D2-brane. In this	
	example $m = 4$ and $n = 2$.	47
9.3	(a) A fundamental string (F1) bound to the D2-brane. The D2-brane	
	wraps the compact direction x_{10} (the vertical direction) and extends	
	indefinitely in direction x_9 (not shown in the picture). The fundamen-	
	tal string is at $x_9 = 0$ and extends in direction x_3 (perpendicular to	
	the plane of the drawing). Because of the S-duality twist, which in the	
	type-IIA picture translates to a rotation, the fundamental string's end-	
	point z'' can be different from its starting point z' . (b) Configurations	
	of the D2-brane with the two endpoints z', z'' of the string marked as	
	oppositely charged points. As the D2-brane changes its x_1 -position.	
	the positions of the charges change accordingly.	48
		-0

9.4	A variable $\mathfrak{p}^{\mathfrak{a}}$ ($\mathfrak{a} = 1, \ldots, m$) is associated with each D2-brane, ordered in the direction of increasing x_3 . Let every string break at every brane into a total of $m+1$ open segments. The constant x_{10} coordinate of each of these segments is denoted by $\mathfrak{q}^{\mathfrak{i}\mathfrak{a}'}$ with $\mathfrak{i} = 1, \ldots, n$ and $\mathfrak{a}' = 0, \ldots, m$ in the direction of increasing x_3 . However, the strings actually break only at positions marked with a \bullet and the main contribution to the action comes from these break points. The strings are coincident, but we separated them in the picture for clarity.	50
9.5	(a) A junction of two open fundamental strings, one starting and one ending on a D2-brane. (b) The D2-brane wraps direction $10 \equiv \natural$ and the configuration can be deformed so that it lifts to a smooth holomor-	
9.6	phic curve in M-theory	53 56
10.1	Illustration of fermionic zero modes for $n = 1$ and $m = 2$ case. The fundamental string breaks at every D2-brane intersection	65
11.111.211.3	Fermionic zero modes on the D2-brane and F1-strings for (a) Sectors 1 and 2, and (b) Sector 3 of the $m = 1$ cases listed in §11.1. In (a), we have trivial identifications $\psi_{20} = \psi_{21}$ and $\lambda_{10} = \lambda_{11}$ An example of sectors without fermionic zero modes. Only one string is attached to each D2-brane	76 79 80
13.1 13.2	The above decomposes the $\sigma = 1$ class of sectors (with $(m + 1)$ D2- branes) in a particular way. The circles on the last two D2-branes indicate various possibilities for the last two columns of \mathfrak{B} , giving pos- sibly different sectors	107 108
15.1	One-loop diagram computed in twisted (2, 0) theory with vertices $h_{\phi\chi}T^{\phi\chi}$ and $h_{\dot{\phi}\dot{\chi}}T^{\dot{\phi}\dot{\chi}}$ where $h_{\phi\chi}$ are deviations from flat metric and $T^{\phi\chi}$ is the low-energy-momentum tensor.	120

List of Tables

8.1	The sequence of dualities from n D3-branes in type-IIB to n funda- mental strings in type-IIA. A direction that the corresponding brane or string wraps with periodic boundary conditions is represented by $-$, a direction that the object wraps with twisted boundary conditions is represented by \div , and a dimension that doesn't exist in the particular string theory is denoted by \times . All the branes in the table are at the origin of directions 4,,9.	35
9.1	Open D2-branes are the U-duals of the type-IIB strings. Here $-$ denotes a direction that the brane wraps, \div denotes a direction that the brane/string wraps with a twist, and \vdash denote a direction in which the brane extends but with endpoints (see §9.9).	46
11.1	The contribution to the Witten Index of the sector with given k and n , and a permutation σ in the conjugacy class that corresponds to the partition $n = n_1 + \cdots + n_n$	92
11.2	The Witten Index as a function of k, n , and m . The behavior of the Witten Index as a function of the number of quark and anti-quark pairs m allows us to calculate the eigenvalues λ_l of the operator $\mathcal{W}^{\dagger}\mathcal{W}$ and their multiplicities N_l . They are listed in the last column as	02
11.3	$\lambda_{1(N_1)}, \lambda_{2(N_2)}, \dots$ The results of [1] regarding the equivalence of the Hilbert spaces of Tr-S and Chern-Simons theory on T^2 as representations of the mapping class group SL(2, Z) together with \mathcal{U}, \mathcal{V} . The notation $U(n)_{k',k''}$ corresponds to a Chern-Simons theory where the $U(1)$ part is at level k'and the SU(n) part is at level k''. One of the sectors $(4 = 2 + 2)$ for $n = 4and k = 1 could not be matched with a Chern-Simons theory and istherefore written explicitly as \mathcal{H}_{(2,2)}. It also appears in the 5 = 2+2+1$	93
	decomposition of the $n = 5$ theory	93

- 11.4 The n = 2 results of [1], sector by sector. Each Hilbert space of a Tr-S sector is equivalent, as a representation of the mapping class group $SL(2,\mathbb{Z})$ and \mathcal{U}, \mathcal{V} , to a corresponding Hilbert space of Chern-Simons theory. The notation $U(2)_{k',k''}$ corresponds to a Chern-Simons theory where the U(1) part is at level k' and the SU(2) part is at level k''. 94
- 13.1 Open strings end on the original n D3 branes and additional 2m D3-branes. Appropriate low onergy background fields on the NS5 brane worldvol

Acknowledgments

I want to thank my advisor, Ori Ganor, for being very kind and supportive throughout our work together. And especially for taking me on, as his student, in a difficult moment in my life. Only we truly know I would not be receiving my PhD if it were not for him. It was so wonderful to work with someone so excited about our research and so patient to explain things over and over again. Ori, thank you, I have learned so much from you.

I would also like to thank all my other wonderful teachers. I am very lucky to have had more amazing teachers than I could fit the names of on this page, so let me spare you such a list. I will remember you and all the things you taught me as long as I possibly can. Knowledge is what I value the most in life — thank you for teaching me.

I am very grateful to all my collaborators on the projects that made it into this dissertation, in part or in full. I am extending my most sincere thanks to Yoon Pyo Hong, Hai Siong Tan, Shannon McCurdy and Nesty Torres Chicon — working with you was so much more satisfying than working alone, thank you.

I want to thank all my close friends. Thank you for accompanying me in the happiness and in the sadness. Thank you for being there. I hope you know how much you mean to me.

I want to thank all the people I dated and without whose love I would have felt so lost during gradschool. I am so happy to have shared this life with you.

I want to thank all my roommates while apologizing for how difficult it was to dig me out from under my books at times. I know I was not always a fun person to live with so thank you for sticking with me.

I also want to thank the Langton Labs community for providing me with another home whenever I needed it and with a shop for all the projects that my Space Ship would not contain. Your science minded people were my other family during gradschool and I am so grateful for having them.

And last, though certainly not least, I want to thank my real family, well, mostly for giving me space when I wanted to work and for having a healthy sense of humor about this physics I do and don't know how to tell them about.

Part I

Insights from Geometric Twists

Chapter 1

An Introduction to Twists

This thesis will present three studies of string theory on twisted geometries that can be constructed as orbifolds, yielding new and interesting results. Thus I introduce this concept first, as an attempt of setting a common theme for my dissertation.

Starting with a manifold M with a discrete symmetry group G we can construct an orbifold M/G via modding out by this group [2, 3, 4]. This means that each point $x \in M$ is identified with its orbit under the action of the group (we think of x and y = gx as the same point for every $g \in G$, as well as their tangent spaces and so on). If M has no fixed points under the action of G, the quotient space M/G is a smooth manifold. On the other hand, fixed points result in orbifold singularities rendering M/G not a manifold. In certain cases it is known how to repair the singularities by removing these points and gluing in manifolds with the appropriate boundary. This is called blowing up and resolving the singularities.

For a simple example, one can consider the real line, \mathbb{R} , with coordinate x. It has an infinite discrete symmetry translating $x \to x + 2\pi R$ with no fixed points so that \mathbb{R}/\mathbb{Z} is a smooth manifold — the circle. There is also a discrete reflection symmetry $x \to -x$ of the real line. This symmetry, however, has a fixed point x = 0 so that the orbifold \mathbb{R}/\mathbb{Z}_2 is the half line starting at an orbifold singularity.

Many mathematical properties follow from the definition of the orbifold. For example, the fundamental group of an orbifold is G/F where F is the subgroup generated by elements that have a fixed point.

Orbifolds are interesting in the context of string theory, since they provide a space of string compactifications that are richer than tori but still have exactly solvable sigma models [5, 6]. We can construct a string theory on the orbifold M/G by gauging a symmetry group G. To do this we must know the action of G both on the space-time and on the gauge degrees of freedom. The symmetry group, in fact, does not have to have a geometric interpretation but it can act only on the internal degrees of freedom.

The orbifold construction indicates that we should keep the states and operators invariant under the symmetry group. But that is not the whole story for a closed string theory. A string theory on an orbifold also includes additional string states with periodicity condition twisted by G. This means that for an element g of Gwe include a twisted sector of strings closed only up to the action of g. Thus the procedure of modding out by G both reduces the number of states because the states must be invariant under G, but it also increases the number of states because of these novel twisted sectors (which correspond to conjugacy classes of G).

Such twisted sectors would not appear in the quantum mechanics of a point particle on an orbifold which is well described only by G-invariant wave functions. But twisted closed string states are allowed (they also must be projected onto a G-invariant subspace) and are actually required for modular invariance.

Strings propagate smoothly over the orbifold singularities of classical geometry. But the strings in twisted sectors are localized at the fixed points as the aperiodicity forbids any center of mass coordinate or momentum. If the fixed points of G are isolated, expansion about strings sitting at different fixed points leads to disjoint sectors in the string propagation. States in the twisted sectors have another interesting feature, we will return to later — their quanta of momentum are commonly non-integral.

The orbifold construction results in breaking of the supersymmetries and gauge symmetries that do not commute with G which is a phenomenologically interesting consequence. Also, considering a theory with a tachyon (a state with a negative mass) on an orbifold sometimes results in a tachyon-free spectrum.

Chapter 2 Examples

Let me start by presenting two examples of simple geometric twists and mentioning some of the properties of the theories on these geometries. The examples will introduce concepts and ideas that the rest of my thesis describes. This chapter contains the background needed for understanding these geometric constructions and their importance for goals of this dissertation. The organization of the dissertation content is given in the following Chapter 3.

The first geometry discussed here is called a Melvin twist and the resulting Melvin Universe is an exact solution of Einstein gravity coupled with gauge fields [7]. It describes a consistent gravitational backreaction to a uniform magnetic field. To construct a Melvin twisted theory of interest, on the gravity side, we start by embedding this twisted geometry into string theory and adding D3-branes to the Melvin background. The gauge theory dual to this setup is Puff Field Theory (PFT). In the dual gravity description of PFT in addition to the usual Ramond-Ramond flux along the D3-branes, there is a strong five-form flux. This RR-field is responsible for Lorentz violation and non-locality in PFT. The preserved symmetry in this case includes spacial rotations and thus makes PFT of particular phenomenological interest.

Another twist this dissertation will focus on is S-duality of four-dimensional super Yang-Mills (SYM) gauge theory in a geometric realization which is available at a selfdual value of gauge coupling constant. As S-duality is then a symmetry of the theory we can compactify the four-manifold on a circle and require that the state of our system after making one loop around this circle is S-dual to its initial configuration. Shrinking this circle to zero size leaves us with a three-dimensional theory that has the S-duality operator at every point and enables us to study its effects. Topological Chern-Simons theory arises as the low-energy limit of the gauge theory on this geometry. And if the leftover three-manifold is curved we also find the gravitational Chern-Simons contribution.

2.1 Melvin twist

The simplest example of the Melvin twist is in three dimensions. This geometry can be thought of as the discrete orbifold $(\mathbb{C} \times \mathbb{R})/\mathbb{Z}$ where, for a $(z, \zeta) \in \mathbb{C} \times \mathbb{R}$, the identification generated by element of \mathbb{Z} is given by

$$\begin{array}{rcl} z & \to & e^{i\theta}z, \\ \zeta & \to & \zeta + 2\pi R. \end{array}$$

$$\tag{2.1}$$

The resulting manifold is smooth, with no singularities, as \mathbb{Z} is freely acting.

Although construction (2.1) does not have a fixed point one can think of the origin of the z-plane as the the deformation near which the twisted states of the orbifold are localized. To see this, consider a low-energy state of the w-twisted sector that winds w times as in part b) of Figure 2.1. The endpoints of such a state are separated by an angle $w\theta$ in the z-plane and a classical string placed a distance r from its origin has energy

$$\alpha'^2 M(r)^2 = (Rw)^2 + \left(2rw\sin\left(\frac{\theta}{2}\right)\right)^2.$$
(2.2)

The wave-functions of such winding states falls off exponentially as $r \to \infty$ and thus strings in this geometry are localized near the origin of the z-plane.



Figure 2.1: Melvin twist in 3 dimensions. a) The construction introduces a rotation in the z-plane by an angle θ for every loop completed in the compact ζ direction. b) A twisted string state in this geometry feels a force toward the origin of the z-plane.

The metric describing this twist in three dimensions can be written in cylindrical coordinates (where z-plane is parametrized by (r, ϕ)) as

$$ds^{2} = -dt^{2} + dr^{2} + r^{2} \left(d\phi + \frac{\theta}{2\pi R} d\zeta \right)^{2} + d\zeta^{2}.$$
 (2.3)

It is obvious that for $\theta = 0$, this twisted space reduces to flat 3+1 dimensions. As well as that there is a natural string theory realization of this Melvin twist, simply, by embedding metric (2.3) into supergravity. Concretely, let us consider type IIA string

theory on this geometry. Then T-dualizing along the ζ -direction results in type IIB theory on a space-time with a background of magnetic fields and scalars

$$ds^{2} = -dt^{2} + dr^{2} + \frac{1}{1 + \left(\frac{\theta r}{2\pi R}\right)^{2}} (r^{2}d\phi^{2} + d\zeta^{2})$$

$$B = \frac{\frac{\theta r}{2\pi R}r^{2}}{1 + \left(\frac{\theta r}{2\pi R}\right)^{2}} d\phi \wedge d\zeta$$

$$e^{\Phi} = \sqrt{\frac{1}{1 + \left(\frac{\theta r}{2\pi R}\right)^{2}}}.$$
(2.4)

This is a Melvin Universe supported by the flux of the NS-NS B-field. Its global geometry is that of a teardrop. Resulting background is special from the point of view of string theory, because its world-sheet sigma model is exactly solvable [8]. This is a simple consequence of it being dual to a flat space with periodic identifications.

This simple Malvin solution in particular is not supersymmetric. However, it is possible to repeat this twist in more than one plane, obtaining a more general Melvin configuration, in such a way that some fraction of supersymmetry is preserved. Further details about more general Melvin twists that preserve some supersymmetry will be discussed in Part II of this dissertation.

In general, Melvin twists, also called TsT transformations, are constructions that rely on an a geometric $U(1) \times U(1)$ isometry of a theory [9]. (In the three-dimensional example above these isometries were the U(1) along the compact ζ -direction and a U(1) for rotation in the z-plane.) A TsT transformation consists of a T-duality, followed by an SL(2, Z) transformation of the complex structure of the dual (a shift) and then another T-duality. The twist could thus be though of as an SL(2, Z) transformation of the Kähler structure of the original theory.

In string theory, applying Melvin twists to Dp-brane backgrounds gives rise to different field theories depending on the direction of the brane. If both of the U(1)isometries are along the D-brane, one gets a noncommutative field theory [10, 11, 12]. Taking only one U(1) transverse to the brane, the result is a dipole field theory [13, 14, 15]. While if both U(1)'s are chosen to be transverse to the brane, we have the Lunin and Maldacena construction [16] also giving the Puff Field Theory [17, 18] which is studied in Part II.

Although these theories were described in terms of metrics on twisted geometries they have another formulation obtained from gauge/gravity duality. On the field theory side, these models can be constructed instead by starting with a Lagrangian and adding a Lorentz violating term which is an IR-irrelevant local operator so that the low-energy behavior is unaffected. The question that then arises is if the UVcompletion exists. Without other corrections a term of conformal dimension $\Delta > 4$ would result in a theory that is not UV-complete. However, the above examples are of UV-complete theories with a local Lorentz violating deformations of super YangMills (SYM) theory. There, UV-completeness is maintained because, in addition to the leading deformation operator, the Lagrangian has an infinite number of nonrenormalizable local terms, which sum up to renormalizable nonlocal interactions.

For SYM on a space of noncommutative geometry the low-energy deformation operator is a two-form of conformal dimension $\Delta = 6$, which breaks the Lorentz group to $SO(2) \times SO(1, 1)$. In the example of the dipole theory, this operator is a spacetime vector of dimension $\Delta = 5$. It breaks the Lorentz group to SO(2, 1). Puff Field theory, in the IR limit, can be approximated by $\mathcal{N} = 4$ SYM deformed by a dimension $\Delta = 7$ operator. This construction leaves unbroken the spatial subgroup of the Lorentz group $SO(3) \subset SO(1,3)$ and thus is of phenomenological interest for constructing field theory in agreement with the Friedmann-Lematre-Robertson-Walker type cosmology which, on large scale, breaks Lorentz invariance to rotational invariance.

2.2 S-Duality twist

A duality is a mathematical equivalence of two theoretical descriptions of a physical systems which, on the first sight, seem different. Such an equivalence often arises when a single quantum theory has distinct classical limits. Probably the most famous example is that of the particle-wave duality. This is simply the fact that quantum field theory has one limit described by classical field theory and another described by classical particle mechanics. Another well known example is Kramers-Wannier duality of the two dimensional square lattice Ising model at a high-temperature with that at a low-temperature [19].

S-duality is a duality under which the coupling constant of a quantum theory changes nontrivially. Important examples include the $SL(2,\mathbb{Z})$ self-dualities of IIB string theory and of four-dimensional $\mathcal{N} = 4$ supersymmetric Yang-Mills (SYM) theory which, in a simplified sense, are just the electric-magnetic duality.

Indeed, S-duality is a generalization of the 19^{th} century electric-magnetic symmetry which can be easily seen from the vacuum Maxwell equations

$$\vec{\nabla} \times \vec{E} = -\frac{\partial \vec{B}}{\partial t}, \qquad \vec{\nabla} \cdot \vec{E} = 0,$$

$$\vec{\nabla} \times \vec{B} = \frac{\partial \vec{E}}{\partial t}, \qquad \vec{\nabla} \cdot \vec{B} = 0.$$
 (2.5)

These equations have an invariance under $\vec{E} \to \vec{B}$ and $\vec{B} \to -\vec{E}$ while the structure constant $\alpha = e^2/4\pi\hbar c$ transforms as $\alpha \to 1/\alpha$. Alike historic electric-magnetic duality which exchanges electric charges and magnetic monopoles, in a higher dimensional space-time, S-duality automatically exchanges the electric and magnetic eigenbranes.

In 1976, [20] showed that in gauge theories electric charge takes values in the weight lattice of the gauge group, while magnetic charge takes values in the weight

lattice of a dual group. This result motivated the Montonen-Olive electric-magnetic duality conjecture [21] according to which a specific gauge theory based on a given gauge group is equivalent to a similar theory with the coupling constant inverted and the gauge group replaced by its dual. The duality is even more natural in a supersymmetric theory [22] and, once θ -angle is included, this \mathbb{Z}_2 symmetry can be extended to a full $SL(2,\mathbb{Z})$ transformation [23, 24, 25] as described bellow.

It is simple to see S-duality arising in abelian gauge theory on a four-manifold M_4 with one-form gauge field A, being the connection on the U(1) bundle, so that the curvature is F = dA. This two form can be written in terms of electric and magnetic fields as $F = dt \wedge d\vec{x} \cdot \vec{E} + \frac{1}{2}d\vec{x} \cdot d\vec{x} \times \vec{B}$ whence Maxwell's vacuum equations read dF = *dF = 0.

Starting with the Minkowski space action

$$S = -\frac{1}{4g_{\rm YM}^2} \int_{M_4} F \wedge *F \tag{2.6}$$

we can perform a Legendre transformation with respect to F and implement Bianchi identity dF = 0 while introducing a dual connection A_D :

$$S = -\frac{1}{4g_{\rm YM}^2} \int_{M_4} F \wedge *F + \frac{1}{2} A_D \wedge dF.$$
 (2.7)

Integrating F out or, equivalently, solving the equation of motion for F gives $dA_D = 1/g_{\text{YM}}^2 * F \equiv F_D$ and the action rewritten in terms of this dual field strength

$$S_D = -\frac{g_{\rm YM}^2}{4} \int_{M_4} F_D \wedge *F_D \tag{2.8}$$

has inverse coupling constant. This argument shows the \mathbb{Z}_2 summery of abelian gauge theory taking $g_{\rm YM} \rightarrow -1/g_{\rm YM}$.

Now, let us assume M_4 to be a compact and oriented spin manifold, as eventually we will want to include fermions. The third homotopy group

$$\frac{1}{(2\pi)^2} \int_{M_4} F \wedge F \tag{2.9}$$

is an even integer. Therefore, we see a classical symmetry of the four-dimensional abelian gauge theory after addition of θ -term to the action

$$S_{\theta} = \frac{\theta}{8\pi^2} \int_{M_4} F \wedge F \tag{2.10}$$

as $e^{iS_{\theta}}$ is invariant under $\theta \to \theta + 2\pi$.

The full Yang-Mills gauge theory action including the θ -term is

$$S = -\frac{1}{4g_{\rm YM}^2} \int_{M_4} F \wedge *F + \frac{i\theta}{8\pi^2} \int_{M_4} F \wedge F.$$

$$(2.11)$$

Where the θ -angle can be combined with the gauge coupling to form the full complex coupling

$$\tau \equiv \frac{4\pi i}{g_{\rm YM}^2} + \frac{\theta}{2\pi} \,. \tag{2.12}$$

Since θ -angle is periodic, $\tau \to \tau + 1$ is a symmetry of a combined theory as well as $\tau \to -1/\tau$. A generic element $g \in SL(2,\mathbb{Z})$ acts on the complex coupling

$$g = \begin{pmatrix} \mathbf{a} & \mathbf{b} \\ \mathbf{c} & \mathbf{d} \end{pmatrix}, \qquad \tau \to \frac{a\tau + b}{c\tau + d}$$
 (2.13)

by implementing both of these symmetries as follows

$$T = \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix}, \qquad \theta \to \theta + 2\pi$$
$$S = \begin{pmatrix} 0 & 1 \\ -1 & 0 \end{pmatrix}, \qquad g_{\rm YM} \to -1/g_{\rm YM}. \qquad (2.14)$$

The elements S and T are a common choice of generators of $SL(2, \mathbb{Z})$, where S is the one that inverts the coupling constant.

We can make Yang-Mills gauge theory supersymmetric by coupling it to massless spin-1/2 particles in the adjoin representation of the gauge group. The number of physical fermionic modes described by such a spinor field depends on the number of spacetime dimensions D and the type of spinor it is but it is always a power of two. However, the number of physical modes described by the gauge field A is D-2, corresponding to all the possible transverse polarizations. This shows that when D = 3, 4, 6 or 10 we can have the same number of physical modes carried by the spinor as by the gauge field which permits an extra symmetry between the gauge bosons and the fermions, called supersymmetry. My dissertation will mostly focus on D = 4 case where the largest possible number of supersymmetries is $\mathcal{N} = 4$.

The generators of supersymmetry transformations are fermionic and are called supercharges. For $\mathcal{N} = 4$ supersymmetry there are 16 separate supercharges, $Q_{\alpha a}$ and $\bar{Q}^a_{\dot{\alpha}}$, where $\alpha, \dot{\alpha} = 1, 2$ and $a = 1, \ldots, 4$ are the spinor and R-symmetry indices, respectively. The R-symmetry in this case is a global $SU(4) \simeq SO(6)$ that transforms supercharges, as well as the fields of this theory, into each other. It commutes with the gauge symmetry.

We can learn new facts about the $SL(2,\mathbb{Z})$ S-duality of $\mathcal{N} = 4$ super Yang-Mills (SYM) theory in four dimensions by studying a circle compactification of the theory with unconventional boundary conditions as follows. Realizing the circle as the segment $[0, 2\pi R]$ with endpoints 0 and $2\pi R$ identied, we require the configuration at $2\pi R$ to be an S-dual of that at 0. We will refer to this kind of boundary conditions as an S-twist and study the effect it has on the action

$$S = \int_{M_3 \times [0,2\pi R)} \mathcal{L}_{SYM} d^4 x + \int_{M_3} A(0) \wedge dA(2\pi R).$$
 (2.15)

After taking the $R \to 0$ limit we are left with a three-manifold that has the S-twist at every point. We call the resulting low-energy theory on M_3 Tr-S.

The S-twist is only an option if it is a symmetry of the theory which happens when the coupling, τ , is self-dual. This occurs at the values of τ for which there exists an element $s \in \mathrm{SL}(2,\mathbb{Z})$, other than the identity I, or -I, that leaves τ invariant. A self-dual theory can be compactified on an S^1 with an s-twist. The 2+1D low-energy limit of this setting has been studied [26, 1] and it was shown that abelian SYM on a flat $M_3 = T^2 \times \mathbb{R}$ gives Chern-Simons theory. The self-dual coupling constant $\tau = i$ is invariant a $\mathbb{Z}_4 \subset \mathrm{SL}(2,\mathbb{Z})$ including $S: \tau \to -1/\tau$ and leads to Chern-Simons at level k = 2. While for $\tau = e^{\frac{\pi i}{3}}$ which is invariant under a $\mathbb{Z}_6 \subset \mathrm{SL}(2,\mathbb{Z})$ we can get levels k = 1, 3.

There is a more geometric realization of the S-twist that I would like to mention now and elaborate on in Part IV of my thesis. Six-dimensional (2, 0) theory of the worldvolume of M5-branes is probably the most natural setting for defining this twist. $\mathcal{N} = 4$ super Yang-Mills theory with coupling constant τ is the low-energy limit of (2, 0) theory compactified on a two-torus, with τ being the complex structure parameter of the torus, where S-duality can be realized as the $SL(2,\mathbb{Z})$ group of the diffeomorphisms of the T^2 .

Now, Tr-S is viewed as the effective low-energy description of the (2, 0) theory formulated on $M_3 \times [(\mathbb{C} \times S^1)/\mathbb{Z}_q]$. Here $(\mathbb{C} \times S^1)/\mathbb{Z}_q$ is the discrete orbifold of $\mathbb{C} \times S^1$, parameterized by (z, ζ) with $0 \leq \zeta < 2\pi Rq$, by \mathbb{Z}_q that is generated by the freely acting $(z, \zeta) \rightarrow (e^{2\pi i q} z, \zeta + 2\pi R)$. Where q = 4 corresponds to $\tau = i$ while q = 6 for $\tau = e^{\frac{\pi i}{3}}$, as above. It is also obvious that we can add an R-symmetry transformation of order q to this \mathbb{Z}_q action. In this description, Tr-S theory is reminiscent of the Melvin twist background described in §2.1.

Chapter 3

Organization

My dissertation will elaborate on these two examples of geometric twists while presenting details of the calculations I worked on. The rest of the document is organized as follows. Part II focuses on the particular Melvin twist that results in Puff Field Theory (PFT) as sketched in §2.1. I will explain further details of this geometry while attempting to explore the Coulomb branch of this field theory. The calculation presented is a project I did with Ori J. Ganor and Shannon McCurdy.

Part III gives a long work on putting static charges into Tr-S theory setup as described in §2.2 which was done with Ori J. Ganor, Yoon Pyo Hong and Hai Siong Tan. This part of my thesis can also be found in [27]. We are interested in the nonabelian Tr-S theory and, in particular, want to show that this theory has no propagating degrees of freedom, and thus is topological, by studying its Witten Index. It is worth note that there are other possible tests of our conjecture that have not been explored here. For example, one could study its Bogomol'nyi-Prasad-Sommerfield (BPS) states — massive representations of the supersymmetry algebra. While momentum is carried by BPS states, if this theory is topological there should not be any propagating degrees of freedom, so one could look for BPS states that carry nonzero momentum along the two-torus the nonabelian SYM is compactified on to disprove our conjecture.

A topological quantum field theory (TQFT) computes topological invariants properties invariant under homeomorphisms. Correlation functions of a TQFT do not depend on the metric of spacetime. We distinguish two basic kinds of TQFTs. Witten-type, or cohomological, TQFTs have explicit metric dependence in the Lagrangian but the partition function and some correlation functions can me shown to be diffeomorphism invariant. One example of this can be found in the Wess-Zumino-Witten (WZW) model [28, 29].

On the other hand, in a Schwartz-type TQFT the path integral measure as well as all the observables are explicitly independent of the metric. In, particular, TQFT found when studying abelian $\mathcal{N} = 4$ SYM on $M_3 \times S^1$ where the circle has an S-duality twist is Chern-Simons theory on M_3 . Chern-Simons theory is a Schwartz-type TQFT and can be specified by a choice of a gauge group and a level k. Although Chern-Simons action is gauge dependent the partition function is not if $k \in \mathbb{Z}$ and gauge field strength vanishes on ∂M_3 . The expectation values of Chern-Simons observables are related to knot invariants [30] and can be calculated as Wilson lines at a fixed point on the circle in Tr-S theory.

Even though we are unsure if nonabelian Tr-S theory is topological, its abelian counterpart is understood better [1]. Compactification of U(1) SYM on S^1 with boundary conditions that are twisted by an S-duality (together with an R-symmetry twist to eliminate zero modes and preserve supersymmetry), leads to a 2 + 1D lowenergy theory that is pure Chern-Simons theory at a level k determined by the SYM coupling constant τ . However, we know that Chern-Simons theory on a manifold with curvature is not quite invariant but requires a gravitational Chern-Simons counterterm [30]. Also, [31] shows that the partition function of Chern-Simons theory is not a modular invariant but transforms as a modular form with weight quadratic in curvature. These results suggest that an S-duality twist should be accompanied with a gravitational Chern-Simons term.

In Part IV, I compare low-energy gravitational Chern-Simons theory with a oneloop amplitude calculation in (2, 0) theory compactified on a two-torus to get $\mathcal{N} = 4$ SYM gauge theory, and then again on a circle with an S-duality twist inserted. This calculation was done in collaboration with Ori J. Ganor and Nesty R. Torres-Chicon.

Part II

A Study of the Coulomb branch of Puff Field theory in a Melvin Universe

Chapter 4

Setup

Puff Field Theory (PFT) arises from an embedding of N D-branes in a particular Melvin Universe, and is an example of a non-local, Lorentz-violating field theory in the decoupling limit. PFT is a gauge theory dual to a string theory with D3-branes in the Melvin background with strong Ramond-Ramond fields.

On the gravity side, PFT can be constructed by lifting type IIA supergravity with N D0-branes to M-theory and introducing Melvin twists in three transverse planes before reducing back to type IIA theory. This gives D0-branes in a Melvin universe [7] supported by an Ramond-Ramond two-form flux. Then T-dualizing along three untwisted directions results in a simple supergravity dual to PFT: N D3-branes in type IIB theory supported by an RR five-form field strength background.

Our particular starting configuration has N D0-branes in type IIA string theory compactified on a three-torus T^3 $(S^1 \times S^1 \times S^1)$ in directions x, y and z. Let the leftover \mathbb{R}^6 directions be parametrized as three (r_i, ϕ_i) planes with $i = 1, \ldots, 3$. We proceed by lifting this setup up to eleven dimensions to get a Kaluza-Klein particle with N units of momentum along the new compact ζ -direction in flat M-theory with $\zeta \sim \zeta + 2\pi R$. The metric is parametrized as follows

$$ds^{2} = -dt^{2} + dx^{2} + dy^{2} + dz^{2} + \sum_{i=1}^{3} \left(dr_{i}^{2} + r_{i}^{2} d\phi_{i}^{2} \right) + d\zeta^{2}$$

We then twist in each of the three (r_i, ϕ_i) planes with deformation parameters β_i

$$ds^{2} = -dt^{2} + dx^{2} + dy^{2} + dz^{2} + \sum_{i=1}^{3} \left(dr_{i}^{2} + r_{i}^{2} (d\phi_{i} + \beta_{i} d\zeta)^{2} \right) + d\zeta^{2}.$$

This is equivalent to doing three separate twists as in the simple three-dimensional example in §2.1.

After reducing on the M-theory ζ -circle back to type IIA, we perform a TsT twist. In this case, the twist is T-duality in direction x, followed by an S-duality, followed by T-dualities in directions y and z, which results in a system of N coincident D3-branes embedded in a Melvin universe with RR four-form potential in the background of type IIB theory.

This IIB supergravity dual solution the has metric

$$ds^{2} = \frac{R^{2}}{r^{2}} \left(K^{-\frac{1}{2}} \left[dx^{2} + dy^{2} + dz^{2} - \left(dt - \frac{4\pi N}{r^{2}} \beta \psi \right)^{2} \right] + K^{\frac{1}{2}} ds_{\mathbb{R}^{6}}^{2} \right)$$
(4.1)

and the RR four-form potential

$$C'_{4} = \frac{\pi N}{r^{4}} K^{-1} \left[dt \wedge dx \wedge dy \wedge dz - \frac{\beta}{r^{2}} \psi \wedge dx \wedge dy \wedge dz \right]$$
(4.2)

with $K = 1 + \left(\frac{4\pi N}{r^2}\right)^2 \frac{\beta^2}{r^2}$ and constant $R^4 = 4\pi g_s N \alpha'^2$, while $g_{\text{IIB}} = 2\pi g_{\text{YM}}^2$. The appropriate decoupling limit was taken for the parameters β , N and R as one sends α' to zero in order to decouple the gauge theory on the D-branes from the rest of the string theory modes.

Here x, y, and z are static coordinates on N coincident D3-branes while six transverse coordinates of \mathbb{R}^6 can be chosen to be radial direction r and coordinates on S^5 so that $ds_{\mathbb{R}^6}^2 = dr^2 + r^2 d\Omega_5^2$. If we think of the five-sphere as the principal U(1)-bundle, or a Hopf fibration of S^1 , over the base $\mathbb{C}P^2$, we can write the metric $d\Omega_5^2 = ds_{\mathbb{C}P^2}^2 + \psi^2$ where ψ is a one-form such that $d\psi$ is a representative of the second de-Rham cohomology $H^2(\mathbb{C}P^2, \mathbb{R})$. Considering the transverse directions as a Hopf fibration is convenient as in these coordinates the Melvin twist is along the direction of the fiber S^1 .

It will also be useful to parametrize the S^5 using six angular coordinates which we will denote \vec{n} with components n_I where $I = 1, \ldots, 6$. Locally, in these angular coordinates, one-form ψ can be written as $\beta \psi = \vec{n}^T \varepsilon d\vec{n}$. Rewriting the metric and the RR four-form in the angular coordinates

$$ds^{2} = \frac{R^{2}}{r^{2}} \left(K^{\frac{1}{2}} \left[dx^{2} + dy^{2} + dz^{2} - \left(dt - \frac{4\pi N}{r^{2}} \vec{n}^{T} \varepsilon d\vec{n} \right)^{2} \right] + K^{-\frac{1}{2}} ds_{\mathbb{R}^{6}}^{2} \right)$$

$$C'_{4} = \frac{\pi N}{r^{4}} K^{-1} \left[dt \wedge dx \wedge dy \wedge dz - \frac{1}{r^{2}} \vec{n}^{T} \varepsilon d\vec{n} \wedge dx \wedge dy \wedge dz \right]$$
(4.3)

where $|\vec{n}|^2 = \sum n_I^2 = 1$ and $d\Omega_5^2 = \sum dn_I^2$.

The Melvin twist that we modded out by is a simultaneous translation in the z-direction and a rotation of the S^5 , where the rotation is determined by the constant matrix $\varepsilon \in so(6)$. In this calculation we have chosen the background where ε is the

following:

$$\varepsilon = \begin{pmatrix} 0 & \beta_1 & 0 & 0 & 0 & 0 \\ -\beta_1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & \beta_2 & 0 & 0 \\ 0 & 0 & -\beta_2 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & \beta_3 \\ 0 & 0 & 0 & 0 & -\beta_3 & 0 \end{pmatrix}; \qquad \beta = \beta_1 = \beta_2 = \beta_3.$$
(4.4)

Although described in terms of gravity above, on the gauge theory side this is a deformation of $\mathcal{N} = 4$ SYM which, with our choice $\beta = \beta_1 = \beta_2 = \beta_3$, breaks all supersymmetry. However, it is possible to construct backgrounds that preserve some fraction of supersymmetry by performing different Melvin twists [17]. Choosing $\beta_1 + \beta_2 + \beta_3 = 0$ to stay in a so(3) subgroup preserves $\mathcal{N} = 1$ supersymmetry. Additionally if we set, say, $\beta_3 = 0$ we are in a so(2) invariant subgroup and our gauge theory has $\mathcal{N} = 2$ supersymmetry. Preserving some supersymmetry is important as a generic background can be unstable.

Chapter 5

Result

Puff Field theory, in the IR limit, can be approximated to the leading order by Yang-Mills gauge theory which is not supersymmetric with our choice of the Melvin twist parameters. We study the Coulomb branch of the decoupled field theory which is flat in this limit. It is deformed by an order seven operator [18] whose bosonic part can be deduced from the Dirac-Born-Infeld (DBI) action and the Wess-Zumino-Witten (WZW) term.

Let the coordinates on the full spacetime be indexed with μ , ν (μ , $\nu = t$, \vec{x} , r, \vec{n}) while we use indices α , β for coordinates along the brane (α , $\beta = t$, \vec{x}) and indices m, n for the transverse directions. Let $G_{\alpha\beta}$ be the pull-back of the full space-time metric $g_{\mu\nu}$ to the D3-brane. If the gauge fields are A_{α} , with field strength F = dA, on the brane and scalars Φ are describing transverse motion of the brane the DBI action is

$$S_{DBI} = -T_3 \int_{M_4} dt d^3x \sqrt{-\det\left(G_{\alpha\beta} + B_{\alpha\beta} + kF_{\alpha\beta}\right)}.$$
(5.1)

While the WZW term is

$$S_{WZW} = \mu_3 \int_{M_4} (C \wedge e^{B+kF}).$$
 (5.2)

Here C is the pullback of the total RR potential to the D3-brane worldvolume, while μ_3 and T_3 are the D3-brane charge and tension. Constant $k = 2\pi \alpha'$ and we set the NS-NS two-form B = 0.

Changing the coordinates on S^5 again, we rescale the six angular denoted \vec{n} (with components n_I where $I = 1, \ldots, 6$, where $|\vec{n}|^2 = \sum n_I^2 = 1$ and $d\Omega_5^2 = \sum dn_I^2$). Take our new field $\vec{\phi}$ to be defined as

$$\phi^{I} = \frac{1}{r}n^{I}$$

$$\partial_{\alpha}\phi^{I} = -\frac{1}{r^{2}}\partial_{\alpha}rn^{I} + \frac{1}{r}\partial_{\alpha}n^{I}.$$

Expanding the DBI action with the WZW term to the lowest order in the deformation parameter $\varepsilon \sim \beta$ and keeping up to three derivatives we get the following order seven operator as the leading contribution deforming the Yang-Mills theory with gauge fields and fermions set to zero

$$\frac{N^2\beta}{2\pi}\frac{\varepsilon_{IJ}\varepsilon_{KL}}{\beta^2}\phi^K\epsilon^{0\alpha\beta\gamma}\partial_\alpha\phi^I\partial_\beta\phi^J\partial_\gamma\phi^L.$$

Further details of this expansions can be found in the following Appendix.

Chapter 6 Appendix

6.1 DBI action

The leading order low-energy effective action for the dynamics of the massless excitations on a D-brane (with gauge field A_{α} supported on the brane and scalars Φ^m describing transverse motion of the brane) is given by the Dirac-Born-Infeld action

$$S_{DBI} = -T_3 \int dt d^3x \sqrt{-\det\left(G_{\alpha\beta} + B_{\alpha\beta} + kF_{\alpha\beta}\right)}$$

where $G_{\alpha\beta}$ is the pull-back of the space-time metric $g_{\mu\nu}$ with coordinates Φ^{μ} (μ , $\nu = t, \vec{x}, r, \vec{n}$) to the brane, $F_{\alpha\beta}$ is the field strength of the gauge field A_{α} on the brane, and $B_{\alpha\beta} = 0$ is the NS-NS two-form, while constant $k = 2\pi\alpha'$. Let us also set $F_{\alpha\beta} = 0$ for simplicity of current discussion and reintroduce it in §6.1.1. The induced metric on the brane can be written as

$$G_{\alpha\beta} = g_{\mu\nu}\partial_{\alpha}\Phi^{\mu}\partial_{\beta}\Phi^{\nu}$$

in static coordinates. We use indices $(\alpha, \beta = t, \vec{x})$ for coordinates on the D3-brane worldvolume, and $(m, n = r, \vec{n})$ for transverse directions, so that

$$G_{\alpha\beta} = h_{\alpha\beta} + g_{\alpha n} \partial_{\beta} \Phi^{n} + g_{m\beta} \partial_{\alpha} \Phi^{m} + g_{mn} \partial_{\alpha} \Phi^{m} \partial_{\beta} \Phi^{n}$$

Where the full space-time metric can be writing in matrix form with coordinates in the order (t, \vec{x}, r, \vec{n}) , where for convenience we define $c_r = \frac{\sqrt{4\pi N}}{r}$

$$g_{\mu\nu} = \frac{R^2}{\sqrt{K}r^2} \begin{pmatrix} -1 & 0 & 0 & c_r^2 \vec{n}^T \varepsilon \\ 0 & \mathbb{1} & 0 & 0 \\ 0 & 0 & K & 0 \\ c_r^2 \varepsilon^T \vec{n} & 0 & 0 & (Kr^2 \mathbb{1} - c_r^4 \varepsilon^T \vec{n} \vec{n}^T \varepsilon) \end{pmatrix}$$

and the rotation matrix

$$\varepsilon = \begin{pmatrix} 0 & \beta & 0 & 0 & 0 & 0 \\ -\beta & 0 & 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & \beta & 0 & 0 \\ 0 & 0 & -\beta & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & \beta \\ 0 & 0 & 0 & 0 & -\beta & 0 \end{pmatrix} \in so(6)$$

with the deformation parameter β .

To calculate the DBI action we need

$$h_{\beta}^{\gamma} = h^{\gamma\alpha}G_{\alpha\beta} = \delta_{\beta}^{\gamma} + h^{\gamma\alpha}g_{\alpha n}\partial_{\beta}\Phi^{n} + h^{\gamma\alpha}g_{m\beta}\partial_{\alpha}\Phi^{m} + h^{\gamma\alpha}g_{mn}\partial_{\alpha}\Phi^{m}\partial_{\beta}\Phi^{n}.$$

The pull-back metric can be written as $h^{\gamma}_{\alpha} = \delta^{\gamma}_{\alpha} + \mathbb{M}^{\gamma}_{\alpha}$, where with static coordinates in the order (t, \vec{x}) in the matrix form

$$\mathbb{M} = \begin{pmatrix} h^{tt}(g_{rr}\partial_t r\partial_t r + \partial_t \vec{n}^T g_{\vec{n}\vec{n}}\partial_t \vec{n} + 2\partial_t \vec{n}^T g_{\vec{n}t}) & h^{tt}(g_{t\vec{n}}\partial_i \vec{n} + g_{rr}\partial_t r\partial_i r + \partial_t \vec{n}^T g_{\vec{n}\vec{n}}\partial_i \vec{n}) \\ h^{ik}(\partial_k \vec{n}^T g_{\vec{n}\mathbf{t}} + g_{rr}\partial_k r\partial_t r + \partial_k \vec{n}^T g_{\vec{n}\vec{n}}\partial_t \vec{n}) & h^{ik}(g_{rr}\partial_k r\partial_j r + \partial_k \vec{n}^T g_{\vec{n}\vec{n}}\partial_i \vec{n}) \end{pmatrix}$$

and

$$h_{\alpha\beta} = \frac{R^2}{\sqrt{K}r^2} \begin{pmatrix} -1 & 0\\ 0 & 1 \end{pmatrix}, \qquad h^{\alpha\beta} = \frac{\sqrt{K}r^2}{R^2} \begin{pmatrix} -1 & 0\\ 0 & 1 \end{pmatrix}.$$

Therefore

$$I_{DBI} = -T_3 \int_{M_4} dt d^3x \sqrt{-\det(h_{\alpha\beta})} \sqrt{\det(\mathbb{1} + \mathbb{M})}$$

where

$$\sqrt{-\det(h_{\alpha\beta})} = \frac{R^4}{Kr^4}.$$

Since

$$\ln \det(\mathbb{1} + \mathbb{M}) = \operatorname{tr} \ln(\mathbb{1} + \mathbb{M}) = \operatorname{tr}(\mathbb{M}) - \frac{1}{2}\operatorname{tr}(\mathbb{M}^2) + \frac{1}{3}\operatorname{tr}(\mathbb{M}^3) + \dots$$

we have

$$(\det(\mathbb{1} + \mathbb{M}))^{\frac{1}{2}} = (e^{\operatorname{tr}(\mathbb{M}) - \frac{1}{2}\operatorname{tr}(\mathbb{M}^{2}) + \dots})^{\frac{1}{2}}$$

= $1 + \frac{1}{2}\operatorname{tr}(\mathbb{M}) - \frac{1}{4}\operatorname{tr}(\mathbb{M}^{2}) + \frac{1}{6}\operatorname{tr}(\mathbb{M}^{3}) + \frac{1}{8}(\operatorname{tr}(\mathbb{M}))^{2}$
 $-\frac{1}{8}\operatorname{tr}(\mathbb{M})(\operatorname{tr}(\mathbb{M}))^{2} + \frac{1}{48}(\operatorname{tr}(\mathbb{M}))^{3} + \dots$

The DBI action, without the gauge fields, where as we expected to find an operator of order seven, we kept terms with up to three derivatives

$$\mathcal{L}_{DBI} = -T_3 \frac{R^4}{Kr^4} \Big(1 - c_r^2 \partial_t \vec{n}^T \varepsilon^T \vec{n} \\ + \frac{1}{2} K (-\partial_t r \partial_t r + \delta^{ij} \partial_i r \partial_j r) + \frac{1}{2} K r^2 (-\partial_t \vec{n}^T (\mathbb{1}) \partial_t \vec{n} + \delta^{ij} \partial_i \vec{n}^T (\mathbb{1}) \partial_j \vec{n}) \\ + (-c_r^2 \partial_t \vec{n}^T \varepsilon^T \vec{n}) \Big(\frac{1}{2} K r^2 (\partial_t \vec{n}^T) (\mathbb{1}) (\partial_t \vec{n}) - \frac{1}{2} K r^2 \delta^{ik} (\partial_k \vec{n}^T) (\mathbb{1}) (\partial_i \vec{n}) \Big) \\ + K (c_r^2 \vec{n}^T \varepsilon \partial_i \vec{n}) \delta^{ik} \partial_k r \partial_t r + \frac{1}{2} K (-c_r^2 \partial_t \vec{n}^T \varepsilon^T \vec{n}) (\partial_t r \partial_t r + \delta^{kj} \partial_k r \partial_j r)) + \dots \Big)$$

where the D3-brane tension, with conventions from [32], is

$$T_3 = \frac{1}{(2\pi)^3 g_s \alpha'^2} = \frac{N}{2\pi^2 R^4}.$$
(6.1)

6.1.1 Including Gauge Fields

Recall that the full DBI action is

$$I_{DBI} = -T_3 \int dt d^3x \sqrt{-\det(G_{\alpha\beta} + kF_{\alpha\beta})}$$

where we so far have only calculated

$$S_{DBI} = -T_3 \int dt d^3x \sqrt{-\det(G_{\alpha\beta})}$$

setting $F_{\alpha\beta} = 0$ in §6.1. Now we need to add

$$S_{DBI} = -T_3 \int dt d^3x \sqrt{-\det(G_{\alpha\beta})} \sqrt{\det(\mathbb{1} + kG^{\beta\gamma}F_{\gamma\beta})}.$$

And we know that

$$\det(G_{\alpha\beta} + kF_{\alpha\beta}) = \det(G_{\alpha\beta} + kF_{\alpha\beta})^T = \det(G_{\alpha\beta} - kF_{\alpha\beta})$$

which is an even function of k. Using matrix notation and denoting

$$\mathbb{N} = kG^{-1}F$$

we have

$$\sqrt{\det(\mathbb{1} + \mathbb{N})} = (\det(\mathbb{1} - \mathbb{N}^2))^{\frac{1}{4}} \\
\approx 1 - \frac{1}{4}\operatorname{tr}(\mathbb{N}^2) - \frac{1}{8}\operatorname{tr}(\mathbb{N}^4) + \frac{1}{32}(\operatorname{tr}(\mathbb{N}^2))^2 + \dots$$

$$\approx 1 + \frac{1}{4}k^2 F_{\alpha\beta}G^{\alpha\delta}G^{\beta\gamma}F_{\delta\gamma} - \frac{1}{8}k^2 (F_{\alpha\beta}G^{\alpha\delta}G^{\beta\gamma}F_{\delta\gamma})^2 + \frac{1}{32}k^4 + \dots$$

However, since we are keeping at most three derivatives for the leading order contribution, we actually only need:

$$\sqrt{\det(\mathbb{1}+\mathbb{N})} \approx 1 + \frac{1}{4}k^2 F_{\alpha\beta}G^{\alpha\delta}G^{\beta\gamma}F_{\delta\gamma}$$

where we only need $G^{\alpha\delta}$ up to single derivatives. This is

$$\begin{aligned} G_{\alpha\beta} &= h_{\alpha\beta} + \begin{pmatrix} 2\partial_t \vec{n}^T g_{\vec{n}t} & g_{t\vec{n}} \partial_i \vec{n} \\ \partial_i \vec{n}^T g_{\vec{n}t} & 0 \end{pmatrix} + g_{mn} \partial_\alpha \Phi^m \partial_\beta \Phi^n \\ G^{\alpha\delta} &\approx h^{\alpha\beta} + (\frac{\sqrt{K}r^2}{R^2})^2 \begin{pmatrix} -2\partial_t \vec{n}^T g_{\vec{n}t} & g_{t\vec{n}} \partial_i \vec{n} \\ \partial_i \vec{n}^T g_{\vec{n}t} & 0 \end{pmatrix} + \dots \\ &\approx h^{\alpha\beta} + (\frac{\sqrt{K}r^2}{R^2}) \tilde{h}^{\alpha\beta} + \dots \end{aligned}$$

and

$$\tilde{h}^{\alpha\beta} = \left(\frac{\sqrt{K}r^2}{R^2}\right) \begin{pmatrix} -2\partial_t \vec{n}^T g_{\vec{n}t} & g_{t\vec{n}}\partial_i \vec{n} \\ \partial_i \vec{n}^T g_{\vec{n}t} & 0 \end{pmatrix} = \begin{pmatrix} -2c_r^2 \partial_t \vec{n}^T \varepsilon^T \vec{n} & c_r^2 \vec{n}^T \varepsilon \partial_i \vec{n} \\ c_r^2 \partial_i \vec{n}^T \varepsilon^T \vec{n} & 0 \end{pmatrix}.$$

Therefore

$$\begin{split} \sqrt{\det(\mathbb{1}+\mathbb{N})} &\approx 1 + \frac{1}{4}k^2 F_{\alpha\beta}h^{\alpha\delta}h^{\beta\gamma}F_{\delta\gamma} + \frac{1}{2}k^2(\frac{\sqrt{Kr^2}}{R^2})F_{\alpha\beta}h^{\alpha\delta}\tilde{h}^{\beta\gamma}F_{\delta\gamma} \\ &\approx 1 + \frac{1}{4}k^2\frac{Kr^4}{R^4}F_{\alpha\beta}\eta^{\alpha\delta}\eta^{\beta\gamma}F_{\delta\gamma} + \frac{1}{2}k^2\frac{Kr^4}{R^4}F_{\alpha\beta}\eta^{\alpha\delta}\tilde{h}^{\beta\gamma}F_{\delta\gamma}. \end{split}$$

And the DBI action, including contributions from gauge fields, is

$$\begin{split} S_{DBI} &= -T_3 \int dt d^3x \sqrt{-\det(G_{\alpha\beta})} \sqrt{\det(\mathbb{1}+\mathbb{N})} \\ &= -T_3 \int dt d^3x \sqrt{-\det(h_{\alpha\beta})} \sqrt{\det(\mathbb{1}+\mathbb{M})} (\det(\mathbb{1}-\mathbb{N}^2))^{\frac{1}{4}} \\ &= -T_3 \int dt d^3x \frac{R^4}{r^4} \Big(1 - \frac{1}{2} (1 + c_r^2 \partial_t \vec{n}^T \varepsilon^T \vec{n}) \partial_t r \partial_t r + \frac{1}{2} \delta^{ij} (1 - c_r^2 \partial_t \vec{n}^T \varepsilon^T \vec{n}) \partial_i r \partial_j r \\ &+ \frac{1}{2} r^2 (-(1 + c_r^2 \partial_t \vec{n}^T \varepsilon^T \vec{n}) \partial_t \vec{n}^T (\mathbb{1}) \partial_t \vec{n} + (1 + c_r^2 \partial_t \vec{n}^T \varepsilon^T \vec{n}) \delta^{ij} \partial_i \vec{n}^T (\mathbb{1}) \partial_j \vec{n}) \\ &- c_r^2 \partial_t \vec{n}^T \varepsilon^T \vec{n} \\ &+ (c_r^2 \vec{n}^T \varepsilon \partial_i \vec{n}) \delta^{ik} \partial_k r \partial_t r \\ &+ \frac{1}{4} k^2 \frac{r^4}{R^4} (1 - c_r^2 \partial_t \vec{n}^T \varepsilon^T \vec{n}) F_{\alpha\beta} \eta^{\alpha\delta} \eta^{\beta\gamma} F_{\delta\gamma} + \frac{1}{2} k^2 \frac{r^4}{R^4} F_{\alpha\beta} \eta^{\alpha\delta} \tilde{h}^{\beta\gamma} F_{\delta\gamma} + \dots \Big). \end{split}$$

6.2 WZW term

We also need to add the terms that come from the Ramond-Ramond field C'_4 in the form of Wess-Zumino-Witten action which picks out the form of order four (as it is proportional to the worldvolume of D3-branes) from

$$S_{WZW} = \mu_3 \int_4 (C \wedge e^{B+kF}).$$

Here C is the total RR potential pulled back to the brane, μ_3 is D3-brane charge, and the NS-NS *B*-field is set to zero. Ignoring the correction due to the curvature tensor, which is of a higher order in derivatives

$$S_{WZW} = \mu_3 \int C'_4$$

as we only have a RR four-form potential. For us

$$C_4' = \frac{\pi N}{r^4} K^{-1} dt \wedge dx \wedge dy \wedge dz - \frac{4\pi N}{r^6} K^{-1} \vec{n}^T \varepsilon d\vec{n} \wedge dx \wedge dy \wedge dz$$

where

$$R^4 = 4\pi N g_s \alpha'^2.$$

This term in the Lagrangian is the pull-back of C'_4 to the brane in static coordinates

$$\mathcal{L}_{WZW} = \mu_3 \frac{\pi N}{r^4 K} (1 - c_r^2 \vec{n}^T \varepsilon \partial_t \vec{n})$$

while from above

$$\mathcal{L}_{DBI} = -\frac{1}{2\pi^2} \frac{N}{Kr^4} \Big(1 - c_r^2 \partial_t \vec{n}^T \varepsilon^T \vec{n} + \dots \Big).$$

Remembering that because $\partial_t(\vec{n}^T \varepsilon \vec{n}) = 0$

$$\vec{n}^T \varepsilon \partial_t \vec{n} = \partial_t \vec{n}^T \varepsilon^T \vec{n}$$

we see that these two contributions in DBI and WZW Lagrangians will cancel with D3-brane charge $\mu_3^{-1} = 2\pi^3$:

$$\mathcal{L}_{DBI} + \mathcal{L}_{WZW} = -T_3 \frac{R^4}{r^4} \Big(-\frac{1}{2} (1 + c_r^2 \partial_t \vec{n}^T \varepsilon^T \vec{n}) \partial_t r \partial_t r + \frac{1}{2} \delta^{ij} (1 - c_r^2 \partial_t \vec{n}^T \varepsilon^T \vec{n}) \partial_i r \partial_j r \\ + \frac{1}{2} r^2 (-(1 + c_r^2 \partial_t \vec{n}^T \varepsilon^T \vec{n}) \partial_t \vec{n}^T (\mathbb{1}) \partial_t \vec{n} + (1 + c_r^2 \partial_t \vec{n}^T \varepsilon^T \vec{n}) \delta^{ij} \partial_i \vec{n}^T (\mathbb{1}) \partial_j \vec{n}) \\ + (c_r^2 \vec{n}^T \varepsilon \partial_i \vec{n}) \delta^{ik} \partial_k r \partial_t r$$
$$+\frac{1}{4}k^2\frac{r^4}{R^4}(1-c_r^2\partial_t\vec{n}^T\varepsilon^T\vec{n})F_{\alpha\beta}\eta^{\alpha\delta}\eta^{\beta\gamma}F_{\delta\gamma}+\frac{1}{2}k^2\frac{r^4}{R^4}F_{\alpha\beta}\eta^{\alpha\delta}\tilde{h}^{\beta\gamma}F_{\delta\gamma}+\dots)$$

However, this RR potential C'_4 is not complete as it does not have a self-dual field strength $dC_4 = F_5 \neq *F_5$:

$$\begin{split} C_4' &= \frac{N}{2\pi^2 r^4 K} (dt - c_r^2 \vec{n}^T \varepsilon d\vec{n}) \wedge dx \wedge dy \wedge dz \\ dC_4' &= \frac{N}{2\pi^2 r^4 K} \frac{-4}{r} dr \wedge dt \wedge dx \wedge dy \wedge dz \\ &\quad -\frac{N}{2\pi^2 r^4 K} \frac{-6}{r} c_r^2 dr \wedge \vec{n}^T \varepsilon d\vec{n} \wedge dx \wedge dy \wedge dz \\ &\quad +\frac{N}{2\pi^2 r^4 K} \frac{6}{r K} (1 - K^{-1}) dr \wedge (dt - c_r^2 \vec{n}^T \varepsilon d\vec{n}) \wedge dx \wedge dy \wedge dz \\ -\frac{N}{2\pi^2 r^4 K} c_r^2 d\vec{n}^T \wedge \varepsilon d\vec{n} \wedge dx \wedge dy \wedge dz \\ F_5' &= dC_4' &= \frac{N}{2\pi^2 r^4 K} \frac{2}{r} \left(1 - \frac{3}{K}\right) dr \wedge dt \wedge dx \wedge dy \wedge dz \\ &\quad +\frac{N}{2\pi^2 r^4 K} \frac{6}{r K^2} dr \wedge c_r^2 \vec{n}^T \varepsilon d\vec{n} \wedge dx \wedge dy \wedge dz \\ &\quad -\frac{N}{2\pi^2 r^4 K} c_r^2 d\vec{n}^T \wedge \varepsilon d\vec{n} \wedge dx \wedge dy \wedge dz \end{split}$$

where I used that

$$dc_r^2 = -2\frac{4\pi N}{r^3} = -\frac{2}{r}c_r^2 dr$$

$$dK = -\frac{6}{r}\left(\frac{4\pi N\beta}{r^3}\right)^2 = -\frac{6}{r}(K-1)dr$$

$$dK^{-1} = -\frac{1}{K^2}dK = \frac{1}{K}\frac{6}{r}(1-K^{-1})dr.$$

Thus we must add to this calculation a WZW term for another four-form A_4 such that $dA_4 = *F'_5$ so that $d(C'_4 + A_4) = F'_5 + *F'_5 = *d(C'_4 + A_4)$.

6.2.1 Writing S^5 As a Hopf Fibration

Due to the symmetries of the problem it is convenient to parametrize the fivesphere in the directions transverse to the N coincident D3-branes as a Hopf fibration instead of using six angular coordinates we denoted \vec{n} . As explained in Chapter 4, we can think of the S^5 as the principal U(1)-bundle over the base $\mathbb{C}P^2$ and rewrite the metric $d\Omega_5^2 = \sum dn_I^2 = ds_{\mathbb{C}P^2}^2 + \psi^2$. Here ψ is a one-form such that $\omega = d\psi$ is a representative of the second de-Rham cohomology $H^2(\mathbb{C}P^2, \mathbb{R})$ and $\beta \psi = \vec{n}^T \varepsilon d\vec{n}$.

If coordinates on $\mathbb{C}P^2$ are v_2 , v_3 , φ_2 and φ_3 while ζ is the coordinate along fiber S^1

$$ds_{\mathbb{C}P^2}^2 = \frac{dv_2^2 + dv_3^2 - (v_2dv_3 - v_3dv_2)^2}{1 - v_2^2 - v_3^2} + v_2^2d\varphi_2^2 + v_3^2d\varphi_3^2 - (v_2^2d\varphi_2 + v_3^2d\varphi_3)^2$$

and

$$\psi = d\xi + \Theta = \frac{1}{\beta} \vec{n}^T \varepsilon d\vec{n}$$

where

$$\Theta = v_2^2 d\varphi_2 + v_3^2 d\varphi_3 \omega = d\Theta = d\psi = 2v_2 dv_2 \wedge d\varphi_2 + 2v_3 dv_3 \wedge d\varphi_3.$$

Then the metric on the S^5 can be rewritten as

$$ds_{S^5}^2 = \frac{dv_2^2 + dv_3^2 - (v_2dv_3 - v_3dv_2)^2}{1 - v_2^2 - v_3^2} + v_2^2 d\varphi_2^2 + v_3^2 d\varphi_3^2 + d\xi^2 + 2d\xi(v_2^2d\varphi_2 + v_3^2d\varphi_3).$$

The five-form field strength F_5^\prime in these coordinates is

$$\begin{split} F'_5 &= \frac{N}{2\pi^2 r^4 K} \frac{2}{r} \Big(1 - \frac{3}{K} \Big) dr \wedge dt \wedge dx \wedge dy \wedge dz \\ &+ \frac{N}{2\pi^2 r^4} \frac{6}{rK^2} c_r^2 \beta dr \wedge \psi \wedge dx \wedge dy \wedge dz \\ &- \frac{N}{2\pi^2 r^4 K} c_r^2 \beta \omega \wedge dx \wedge dy \wedge dz \\ &= \frac{N}{2\pi^2 r^4 K} \frac{2}{r} \Big(1 - \frac{3}{K} \Big) dr \wedge dt \wedge dx \wedge dy \wedge dz \\ &+ \frac{N}{2\pi^2 r^4} \frac{6}{rK^2} c_r^2 \beta dr \wedge (d\xi + \Theta) \wedge dx \wedge dy \wedge dz \\ &- \frac{N}{2\pi^2 r^4 K} c_r^2 \beta \omega \wedge dx \wedge dy \wedge dz. \end{split}$$

Recalling that the Hodge dual of a *m*-form η is an *n*-form $*\eta$, where m + n = D is the number of spacetime dimensions, and the components are

$$*\eta_{i_1\dots i_n} = \frac{1}{k!}\sqrt{|\det g|}g^{j_1k_1}\dots g^{j_mk_m}\eta_{j_1\dots j_m}\epsilon_{k_1\dots k_m i_1\dots i_n}$$

for the Hodge dual of our five-form field strength we get

$$* F_5' = \frac{N}{4\pi^2} \Big(\omega \wedge \omega \wedge d\xi + \frac{2\pi N\beta}{r^4} dt \wedge \omega \wedge \omega + \frac{8N\pi\beta}{r^5} dr \wedge dt \wedge \omega \wedge \psi \Big)$$

Where we are looking for A_4 such that, locally, $dA_4 = *F'_5$, which is exists because F

is closed¹. The correction four-form A_4 contains

$$A_4 = \frac{N}{4\pi^2} \left(\frac{\xi}{a} \omega \wedge \omega + \left(1 - \frac{1}{a} \right) \theta \wedge \omega \wedge d\xi + \frac{2N\pi\beta}{r^4} dt \wedge \omega \wedge \psi \right)$$

for some constant a.

There is only one term in the pull-back of A_4 that contains at most three derivatives of the fields

$$A_4|_{3derivatives} = \frac{N}{4\pi^2} \Big(\frac{2N\pi\beta}{r^4} dt \wedge \omega \wedge \psi \Big).$$

In terms of the \vec{n} coordinates on S^5

$$\psi = \frac{1}{\beta} \vec{n}^T \varepsilon d\vec{n}$$
$$\omega = \frac{1}{\beta} d\vec{n}^T \wedge \varepsilon d\vec{n}$$

we can rewrite it as

$$A_4|_{3derivatives} = \frac{N}{4\pi^2} \Big(\frac{2N\pi\beta}{r^4} \frac{\varepsilon_{IJ}\varepsilon_{KL}}{\beta^2} n^K \epsilon^{0\alpha\beta\gamma} \partial_\alpha n^I \partial_\beta n^J \partial_\gamma n^L dt \wedge dx \wedge dy \wedge dz \Big).$$

Then its pull back to the brane, up to an overall factor, looks like

$$P(A_4|_{3derivatives}) = \frac{N}{4\pi^2} \left(\frac{2N\pi\beta}{r^4} \frac{\varepsilon_{IJ}\varepsilon_{KL}}{\beta^2} n^K \epsilon^{0\alpha\beta\gamma} \partial_\alpha n^I \partial_\beta n^J \partial_\gamma n^L \right).$$

6.3 Stress-Energy Tensor

The most instructive way to rewrite all the contributions to the low-energy effective action is in terms of the stress-energy tensor of rescaled transverse fields. Let us define these new fields as

$$\phi^I = \frac{1}{r}n^I$$

¹ If F were not closed ([32] pg. 205)

$$\begin{array}{rcl} F & = & F_5' + *F_5' \\ dF & = & *J_m \\ d*F & = & *J_e(=*J_m = *J). \end{array}$$

From this we would have

 $* d * F'_5 = J.$

$$\partial_{\alpha}\phi^{I} = -\frac{1}{r^{2}}\partial_{\alpha}rn^{I} + \frac{1}{r}\partial_{\alpha}n^{I}$$

and remember that

$$\vec{n}^T \vec{n} = 1, \qquad \partial_\alpha(\vec{n}^T \vec{n}) = 0.$$

We have

$$\partial_{\alpha}\phi^{I}\delta_{IJ}\partial_{\beta}\phi^{J} = \frac{1}{r^{4}}\partial_{\alpha}r\partial_{\beta}r + \frac{1}{r^{2}}\partial_{\alpha}\vec{n}^{T}\partial_{\beta}\vec{n}$$

and define

$$J_{\alpha} = \delta \phi = \frac{1}{r^2} \vec{n}^T \varepsilon \partial_{\alpha} \vec{n}.$$

Then stress-energy tensor contributions from the Yang-Mills gauge field strength F = dA and transverse scalar fields $\vec{\phi}$ can be written as

$$(T_{YM})_{\alpha\beta} = \eta^{\gamma\nu} F_{\alpha\gamma} F_{\beta\nu} - \frac{1}{4} \eta_{\alpha\beta} F_{\gamma\nu} F^{\gamma\nu} (T_{\phi})_{\alpha\beta} = \partial_{\alpha} \phi^{I} \delta_{IJ} \partial_{\beta} \phi^{J} - \frac{1}{2} \eta_{\alpha\beta} \eta^{\gamma\delta} \partial_{\gamma} \phi^{I} \delta_{IJ} \partial_{\delta} \phi^{J}.$$

While the terms

$$\eta^{\delta\alpha} J_{\delta}(T_{\phi})_{\alpha t} = -J_t(T_{\phi})_{tt} + J_i(T_{\phi})_{it}$$

$$= -J_t(\frac{1}{2}(\partial_t \phi^I \delta_{IJ} \partial_t \phi^J + \partial_i \phi^I \delta_{IJ} \partial_i \phi^J) + J_i(\partial_i \phi^I \delta_{IJ} \partial_t \phi^J)$$

$$\eta^{\delta\alpha} J_{\delta}(T_{YM})_{\alpha t} = -J_t(T_{YM})_{tt} + J_i(T_{YM})_{it}$$

$$= -J_t(F_{ti}F_{ti} + \frac{1}{4}F_{\gamma\nu}F^{\gamma\nu}) + J_i(F_{ij}F_{tj}).$$

Adding the the contributions of DBI action and WZW term with A_4 correction gives the total Lagrangian density up to three derivatives

$$\mathcal{L}_{DBI} + \mathcal{L}_{WZW} + \mathcal{L}_{A_4} = \frac{N}{2\pi^2} \Big(\mathcal{L}_{\phi} + \frac{\pi}{g_s N} \mathcal{L}_{YM} \\ -(4\pi N)\eta^{\delta\alpha} J_{\delta} (T_{\phi} + T_{YM})_{\alpha t} \\ + \Big(N\pi\beta \frac{\varepsilon_{IJ} \varepsilon_{KL}}{\beta^2} \phi^K \epsilon^{0\alpha\beta\gamma} \partial_{\alpha} \phi^I \partial_{\beta} \phi^J \partial_{\gamma} \phi^L \Big) \dots \Big).$$

Part III

Static Charges in the Low-Energy Theory of the S-Duality Twist

Chapter 7

Introduction

For U(n) gauge group, S-duality of $\mathcal{N} = 4$ super Yang-Mills theory [21, 22, 33] asserts that the theory at the complex combination of coupling constant and θ angle

$$\tau \equiv \frac{4\pi i}{g_{\rm YM}^2} + \frac{\theta}{2\pi}$$

is equivalent to the same theory at complex coupling $-1/\tau$. The conjecture has passed many tests (see for instance [34, 35, 36, 37]) and is generally accepted as true, even though no proof exists. Over the years, much insight has been accumulating on the way S-duality works. Some notable breakthroughs include the geometric realization of S-duality in terms of the (2,0)-theory [38, 39], the connection with the geometric Langlands program [40], and the discovery [41, 42] of the role of certain strongly-coupled $2+1D \mathcal{N} = 4$ theories [43] as intertwiners between a supersymmetric boundary condition and its S-dual.

Another way to explore S-duality was recently examined in [26, 1]. There, an S-duality twist was introduced into a compactification of $\mathcal{N} = 4$ super Yang-Mills (SYM) on S^1 in a way that preserves $\mathcal{N} = 6$ supersymmetry in 2 + 1D. An S-duality twist is an unusual possible boundary condition that is permissible when the complex coupling constant is set to the self-dual value $\tau = i$. The S-duality twist is then achieved by inserting the transformation that realizes $\tau \to -1/\tau$ at some point along S^1 . We can then further compactify the remaining two spatial dimensions on, say, T^2 . The Hilbert space of ground states of this compactification, which was studied in [1], provides insight into the operator that realizes S-duality. We refer to the resulting three-dimensional low-energy theory as Tr-S, because correlation functions of Wilson loops $\langle W(C_1) \cdots W(C_l) \rangle_{\text{Tr-S}}$ in this theory can be interpreted as, roughly speaking, a regularized version of the trace $\operatorname{tr}((-1)^F SRW(C_1) \cdots W(C_l))$, where the trace is taken over the Hilbert space of $\mathcal{N} = 4 U(n)$ SYM at the self-dual coupling $\tau = i$, S is the S-duality operator, R is an appropriate SU(4) R-symmetry operator that is inserted in order to preserve $\mathcal{N} = 6$ supersymmetry in 2 + 1D, F is the fermion number (which is equivalent to a central element of the R-symmetry group), and $W(C_1), \ldots, W(C_l)$ are Wilson loop operators associated with the loops C_1, \ldots, C_l in \mathbb{R}^3 .

S-duality is part of a larger $SL(2,\mathbb{Z})$ group of dualities, and some of them can be used as twists as well. These arise for the special coupling $\tau = e^{\frac{i\pi}{3}}$, which is invariant under $\tau \to (\tau - 1)/\tau$ and $\tau \to -1/(\tau - 1)$, and the corresponding $SL(2,\mathbb{Z})$ elements can be used to twist the boundary conditions. Together with $\tau \to -1/\tau$, we thus have three $SL(2,\mathbb{Z})$ elements to explore as possible twists.¹ We denote a general $SL(2,\mathbb{Z})$ -element by

$$\mathbf{g} \equiv \begin{pmatrix} \mathbf{a} & \mathbf{b} \\ \mathbf{c} & \mathbf{d} \end{pmatrix}, \qquad \tau \to \frac{\mathbf{a}\tau + \mathbf{b}}{\mathbf{c}\tau + \mathbf{d}}, \tag{7.1}$$

and we denote its order in the group by \mathbf{r} (thus $\mathbf{g}^{\mathbf{r}} = 1$). In [1] an integer k and a phase $-\pi < \upsilon < \pi$ were assigned to the three $SL(2, \mathbb{Z})$ elements \mathbf{g} as follows:

$$e^{iv} \equiv \mathbf{c}\tau + \mathbf{d}, \qquad k \equiv \frac{2 - \mathbf{a} - \mathbf{d}}{\mathbf{c}}.$$
 (7.2)

It can easily be checked that in our three cases v is real and equal to $2\pi/\mathbf{r}$, and k is an integer. Explicitly,

$$k = 1, \quad \mathbf{r} = 6, \quad \upsilon = \frac{\pi}{3} \quad \text{for} \quad \tau = e^{\pi i/3}, \quad \mathbf{g} = \begin{pmatrix} 1 & -1 \\ 1 & 0 \end{pmatrix};$$

$$k = 2, \quad \mathbf{r} = 4, \quad \upsilon = \frac{\pi}{2} \quad \text{for} \quad \tau = i, \qquad \mathbf{g} = \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix};$$

$$k = 3, \quad \mathbf{r} = 3, \quad \upsilon = \frac{2\pi}{3} \quad \text{for} \quad \tau = e^{\pi i/3}, \quad \mathbf{g} = \begin{pmatrix} 0 & -1 \\ 1 & -1 \end{pmatrix}.$$
(7.3)

In [1] the study of the Tr-S theory with gauge group U(n) compactified on T^2 was started, and the Hilbert space of ground states was determined. We use the notation $\mathcal{H}(k,n)$ to refer to the Hilbert space of ground states of the theory with τ and **g** that are determined according to the list (7.3). In [1] it was also found convenient to restrict attention to the cases with $n < \mathbf{r}$, since these have no Coulomb branch, as we review in Chapter 8. This restriction arises as in cases where $n \geq \mathbf{r}$ there are elements of the Weyl group $S_n \subset U(n)$ that have order **r**, and a gauge transformation by such a Weyl group element can cancel the effect of the R-symmetry twist and produce a zero-mode. For $n < \mathbf{r}$ there are no such zero modes. We believe that in

¹The twists are in $SL(2,\mathbb{Z})$ and not $PSL(2,\mathbb{Z})$ because the central element $-1 \in SL(2,\mathbb{Z})$ acts nontrivially, being equivalent to charge conjugation. For each $\mathbf{g} \in SL(2,\mathbb{Z})$ in the list (7.3) below, one can also use $-\mathbf{g}$ as a twist, but it is always identical to the inverse of an element that already appears in the list, and the resulting theory with $-\mathbf{g}$ is always the parity transform of another theory from the list.

this case the compactification has a mass-gap, and the 2+1D low-energy theory Tr-S is topological.

The purpose of this paper is to continue to explore the Hilbert space of Tr-S on T^2 by introducing supersymmetric static charges corresponding to m pairs of heavy quarks and anti-quarks at fixed locations on T^2 and S^1 . We will study the resulting Hilbert space of ground states, and show that the Witten Index of this problem can be calculated by counting the states of a simple quantum mechanical system with action

$$I = \frac{1}{2\pi} \int \sum_{\mathfrak{a},\alpha=1}^{m} \mathcal{M}_{a\alpha} \mathfrak{p}^{\mathfrak{a}} \dot{\mathfrak{q}}^{\alpha} dt , \qquad (7.4)$$

where $\mathcal{M}_{\mathfrak{a}\alpha}$ is an $m \times m$ matrix with integer entries, and $\mathfrak{p}^1, \mathfrak{q}^1, \ldots, \mathfrak{p}^m, \mathfrak{q}^m$ are periodic coordinates with period 2π . The action (7.4) describes geometric quantization of T^{2m} .

One motivation for introducing static charges into the Tr-S theory — apart from a better understanding of the theory itself — is to clarify the relationship between Tr-S and another (known) topological theory in 2 + 1D, namely the Chern-Simons theory. In fact, when the gauge group is abelian, in which case the $\mathcal{N} = 4$ SYM is a free theory and we have a complete understanding of its S-duality [31], we have an explicit description for Tr-S: it is simply the abelian Chern-Simons theory at level kgiven in (7.2) (see §5 of [1]).

This simple picture does not hold true for nonabelian gauge groups, but the result of [1] suggested that there might still exist a close relationship between the two theories. There, we observed that the Hilbert space of the Tr-S theory compactified on T^2 decomposes into different sectors, and for almost all sectors we were able to show that their symmetry operators and behavior under modular transformations of T^2 agree with those of the Hilbert spaces of the Chern-Simons theory with appropriate gauge groups and levels. Introduction of static charges then provides a further test on the identification of the Hilbert spaces of two theories.

Our strategy for extracting the Witten Index of the system with static charges inserted follows closely that of [1]. Since little is known about the S-duality operator itself for nonabelian gauge groups, we will embed our setting into full type IIB string theory and apply a series of string theory dualities, after which the low-energy description of the system is given by the simple quantum mechanical one in (7.4). For abelian gauge group, the result we obtain in this way precisely agrees with what we expect from introducing Wilson line operator in abelian Chern-Simons theory. This is as it must be, because we already know that the Tr-S theory *is* Chern-Simons theory in this case. For nonabelian gauge groups, we show that our result passes a nontrivial consistency check in itself, but does not agree with Chern-Simons theory predictions in general. We will provide more discussion on this discrepancy in the concluding section.

The paper is organized as follows. In Chapter 8 we explain the problem in detail and describe the S-duality and R-symmetry twists, the amount of supersymmetry that is left unbroken, and the absence of the Coulomb branch for $n < \mathbf{r}$. We then construct the type-IIA dual of the theory without the charges. In Chapter 9 we introduce the supersymmetric static charges and derive their type-IIA dual description. We then explain how the geometric quantization systems of the type (7.4) arise. In Chapter 10 we use the known solution of the problem with U(1) gauge group, which reduces to U(1) Chern-Simons theory at level k [defined in (7.3)], to demonstrate how the type-IIA dual reproduces the known results about the Hilbert space of ground states of U(1) Chern-Simons theory with charges. In Chapter 11 we move to the case of U(n)gauge group. The goal of this section is to calculate the Witten Index of this system as a function of n, k, and m. We describe the technical aspects of the calculation in detail and summarize the final results in Table 11.2. Next, using a Wick-rotation we express the Witten index as a supertrace of a combination of spatial Wilson loops over $\mathcal{H}(k,n)$ (the Hilbert space without external charges). This provides us with a consistency check on the final result, and moreover, allows us to calculate the eigenvalues of the spatial Wilson loop operators on $\mathcal{H}(k,n)$. We then compare the results to Chern-Simons theory as conjectured in [1], and show that they do not match. We conclude in Chapter 12 with a discussion and future directions.

Chapter 8

Review of the S-duality twist and its type-IIA dual

In this section, we carry out a brief review of Tr-S theory, how to realize it in string theory, and how to construct the type-IIA dual of the theory on T^2 . We refer our readers to [1] for more comprehensive details. At the end of this section we include a small discussion on why we believe Tr-S is topological for $n < \mathbf{r}$.

8.1 Definition of Tr-S

By definition, Tr-S is the 2 + 1D low-energy limit of a compactification of $\mathcal{N} = 4$ super Yang-Mills (SYM) theory on S^1 with boundary conditions that include an S-duality twist and an appropriate R-symmetry twist to be discussed below. This compactification was introduced in [26, 1], and similar compactifications have also recently been studied in [44, 45, 46], where the S-duality twist was referred to as a "duality wall."

By itself, S-duality does not commute with supersymmetry [40], and since we want to preserve some supersymmetry we have to supplement the S-twist with an R-symmetry twist. Therefore, in [1] we twisted the boundary conditions on S^1 further by an element γ of the R-symmetry group SU(4), which in a particular basis takes the form:

$$\gamma = \begin{pmatrix} e^{\frac{i}{2}v} & & \\ & e^{\frac{i}{2}v} & \\ & & e^{\frac{i}{2}v} \\ & & & e^{-\frac{3i}{2}v} \end{pmatrix} \in SU(4)_R,$$
(8.1)

where v is given by (7.2). This choice, it turns out, preserves the maximal possible amount of supersymmetry in the presence of S-duality and R-symmetry twists, which is $\mathcal{N} = 6$ in 2 + 1D. It also preserves a $U(3) \subset SU(4)_R$ R-symmetry. This U(3) can be thought of as the unitary group of rotations that act holomorphically on the transverse $\mathbb{C}^3 \simeq \mathbb{R}^6$.

 $3 + 1D \ \mathcal{N} = 4$ SYM has a Coulomb branch on which the gauge group is broken to $U(1)^n \subset U(n)$. After compactification with the S-twist and R-twist together, the Coulomb branch completely disappears for $n < \mathbf{r}$ [1]. This is because a point on the Coulomb branch $(\mathbb{R}^6)^n/S_n$ of $\mathcal{N} = 4 U(n)$ SYM is described by an unordered set of noncoincident n points in \mathbb{R}^6 . S-duality preserves the point in moduli space, but the R-symmetry twist γ acts on it by rotation of \mathbb{R}^6 . Since γ has order $\mathbf{r} \equiv 2\pi/v$ when acting on \mathbb{R}^6 , we see that we need $n \geq \mathbf{r}$ for a point on the Coulomb branch to survive the twist. For $n \geq \mathbf{r}$ the situation is more complicated and some portion of the moduli space of the Coulomb branch survives [1]. In order to avoid these complications we will restrict the discussion that follows to the case $n < \mathbf{r}$.

We can easily realize Tr-S in string theory using D3-branes. Consider the type-IIB background $\mathbb{R}^{9,1}$ with Cartesian coordinates x_0, \ldots, x_9 , and place n D3-branes at $x_4 = x_5 = \cdots = x_9 = 0$. The type-IIB coupling constant is denoted by $\tau = \chi + \frac{i}{g_{\text{IIB}}}$, where g_{IIB} is the string coupling constant, and χ is the RR scalar. The S-duality transformation \mathbf{g} of (7.1) then lifts to an S-duality transformation of the full type-IIB string theory (that we also denote by \mathbf{g}), and the R-symmetry element γ lifts to a geometrical rotation in the six directions transverse to the D3-branes. We compactify the x_3 -direction on a circle of radius R with boundary conditions given by a simultaneous S-duality twist \mathbf{g} and a $\gamma \in \text{Spin}(6)$ geometrical twist in directions x_4, \ldots, x_9 , where γ is given by (8.1). This means that as we traverse the x_3 circle once, we also apply a $\gamma \in \text{Spin}(6)$ rotation in the transverse directions before gluing $x_3 = 0$ to $x_3 = 2\pi R$.

We now compactify directions x_1, x_2 , so that $0 \le x_1 < 2\pi L_1$ and $0 \le x_2 < 2\pi L_2$ are periodic. This puts the 2 + 1D field theory on T^2 with area $4\pi^2 L_1 L_2$ and complex structure iL_2/L_1 . In the limit

$$L_1, L_2, R \gg {\alpha'}^{1/2},$$
 (8.2)

where ${\alpha'}^{1/2}$ is the type-IIB string scale, we can first reduce the description of the D3-branes to $\mathcal{N} = 4 \ U(n)$ SYM at low-energy, and then compactify $\mathcal{N} = 4$ SYM with an S-duality and R-symmetry twist.

8.2 Type-IIA dual

We will now transform the type-IIB background, using string dualities, to one where S-duality is realized geometrically. For this we need to consider the opposite limit $L_1, L_2 \rightarrow 0$ with $R \rightarrow \infty$ (in the order to be specified below). In this limit, the type-IIB description is strongly coupled, but we will perform a U-duality transformation as specified in Table 8.1 to transform the setting to a weakly coupled type-IIA background, enabling us to easily study the ground states of the field theory.

Type	Brane	1	2	3	4	5	6	7	8	9	10	Apply:
IIB	D3	_	_	÷							×	T-duality on 1
IIA	D2		_	÷							×	Lift to M-theory
М	M2		_	÷								Reduction to IIA on 2
IIA	F1		×	÷								This is the final step!

Table 8.1: The sequence of dualities from n D3-branes in type-IIB to n fundamental strings in type-IIA. A direction that the corresponding brane or string wraps with periodic boundary conditions is represented by -, a direction that the object wraps with twisted boundary conditions is represented by \div , and a dimension that doesn't exist in the particular string theory is denoted by \times . All the branes in the table are at the origin of directions $4, \ldots, 9$.

The U-duality transformation proceeds as follows. We first replace type-IIB on a circle of radius L_1 with M-theory on T^2 with complex structure τ and area $\mathcal{A} = (2\pi)^2 \alpha'^2 \tau_2^{-1} L_1^{-2} = (2\pi)^2 M_{\rm p}^{-3} L_1^{-1}$, where $M_{\rm p}$ is the 11-dimensional Planck scale. We now reduce from M-theory to type-IIA on the circle of radius L_2 to get a theory with string coupling constant

$$g_{\text{IIA}} = (M_{\text{p}}L_2)^{3/2} = \tau_2^{1/2} L_1^{1/2} L_2^{3/2} {\alpha'}^{-1}$$

and new string scale

$$\alpha'_{\text{IIA}} = M_{\text{p}}^{-3}L_2^{-1} = {\alpha'}^2 \tau_2^{-1}L_1^{-1}L_2^{-1}.$$

After these dualities, the D3-branes become fundamental type-IIA strings with a total winding number n in the x_3 direction. The S-duality twist **g** is now a diffeomorphism of the type-IIA torus (in the $x_{10}x_1$ directions), which can be realized as a rotation by an angle v listed in (7.3). To make this type-IIA background weakly coupled we assume that the limits are taken in such a way that

$$\mathcal{A} \gg \alpha'_{\text{IIA}}, \qquad g_{\text{IIA}} \ll 1, \qquad R \gg \alpha'_{\text{IIA}}^{1/2}.$$
 (8.3)

This is a different limit than (8.2), but we can use the weakly coupled type-IIA background to study the Hilbert space of supersymmetric ground states, or more precisely the Witten Index of the Tr-S theory on T^2 .

To describe the basic geometry of the dual type-IIA background, it is convenient to divide the 9 directions into three groups and view the spatial manifold as an orbifold of $T^2 \times \mathbb{R} \times \mathbb{C}^3$. We regard the T^2 as the complex plane modded out by a lattice, $\mathbb{C}/(\mathbb{Z} + \tau \mathbb{Z})$, and take

$$z \sim z + 1 \sim z + \tau \tag{8.4}$$

as its coordinate. On \mathbb{R} , we take the coordinate

$$-\infty < x_3 < \infty$$

and on $\mathbb{C}^3 \simeq \mathbb{R}^6$, we take the coordinates to be

$$(\zeta_1, \zeta_2, \zeta_3), \qquad \zeta_1, \zeta_2, \zeta_3 \in \mathbb{C}.$$

The orbifold is then represented by the identification

$$(z, x_3, \zeta_1, \zeta_2, \zeta_3) \sim (e^{iv} z, x_3 + 2\pi R, e^{iv} \zeta_1, e^{iv} \zeta_2, e^{iv} \zeta_3).$$
(8.5)

Also, the shift $x_3 \to x_3 + 2\pi R$ ensures that the orbifold has no fixed points, and thus the geometry is flat and free of singularities. In particular, the $\zeta_1 = \zeta_2 = \zeta_3 = 0$ subspace is a T^2 -fibration over S^1 with structure group \mathbb{Z}_r .

The ground states that are relevant to our problem are those with a total string winding number n along direction x_3 . A state with string winding number n is a p-particle (that is, p-string) state consisting of 1-particle states of winding numbers $n_1 \ge n_2 \ge \cdots \ge n_p > 0$ with $n_1 + n_2 + \cdots + n_p = n$. Thus, the Hilbert space of ground states decomposes as a direct sum:

$$\mathcal{H}(k,n) = \bigoplus_{p=1}^{n} \left(\bigoplus_{\substack{n_1 \ge n_2 \ge \dots \ge n_p > 0\\n_1 + n_2 + \dots + n_p = n}} \mathcal{H}(k; n_1, \dots, n_p) \right).$$
(8.6)

A crucial point is that the problem of finding the ground states can be solved using essentially classical geometry: we simply need to find classical string configurations of minimal length. Consider a superstring with winding number \tilde{n} in direction x_3 . It turns out [8] that (for $\tilde{n} \neq 0$) the ground states are bosonic and in the RR sector. (We will independently verify this in Chapter 10.3.) For \tilde{n} that is not divisible by **r**, there is a basis of ground states that are in one-to-one correspondence with loops of winding number \tilde{n} and minimal length in the geometry (8.5). In the limit $\alpha'_{\text{IIA}} \rightarrow 0$, these states reduce to the classical string configurations.

To describe the classical configurations, we can fix an x_3 coordinate and specify the points where the classical string intersects the transverse coordinates $T^2 \times \mathbb{C}^3$ in the geometry (8.5). At winding number \tilde{n} , the string intersects $T^2 \times \mathbb{C}^3$ at \tilde{n} (not necessarily distinct) points, and in order to be of minimal length the coordinates of these points should be independent of x_3 . The classical configurations are thus characterized by a set of \tilde{n} points in $T^2 \times \mathbb{C}^3$ that is invariant, as a set, under the orbifold operation

$$(z,\zeta_1,\zeta_2,\zeta_3) \sim (e^{iv}z,e^{iv}\zeta_1,e^{iv}\zeta_2,e^{iv}\zeta_3).$$

For \tilde{n} that is not divisible by **r**, there is a finite number of such sets, and they are all localized at the origin of \mathbb{C}^3 , i.e., at $\zeta_1 = \zeta_2 = \zeta_3 = 0$. They are therefore entirely described by the z-coordinates of where the string intersects T^2 : $z, e^{iv}z, e^{2iv}z, \ldots, e^{i(\tilde{n}-1)v}z$, since as we go once around the x_3 direction the coordinate z switches to $e^{iv}z$. After \tilde{n} loops z becomes $e^{i\tilde{n}v}z$, which should be identified with z (up to a shift in $\mathbb{Z} + \mathbb{Z}\tau$) in order to close the string. The classical string configurations are then described by solutions $z = \zeta_{M_a,M_b}$ of

$$e^{i\bar{n}\upsilon}\zeta_{M_a,M_b} = \zeta_{M_a,M_b} + M_a + M_b\tau, \qquad (M_a,M_b\in\mathbb{Z})$$
(8.7)

and we consider two solutions ζ_{M_a,M_b} and $\zeta_{M_a',M_b'}$ as equivalent if they differ by a lattice element, i.e., if $\zeta_{M_a,M_b} - \zeta_{M_a',M_b'} \in \mathbb{Z} + \mathbb{Z}\tau$. In addition, ζ_{M_a,M_b} and $e^{i\upsilon}\zeta_{M_a,M_b}$ give equivalent solutions, since the intersection points of the string with T^2 are unordered.

There is then only a finite number of inequivalent solutions to (8.7), and we have described them in detail in [1]. The full single-particle string spectrum (including excited states) decomposes into a finite sum of distinct sectors, labeled by M_a, M_b , and the solution ζ_{M_a,M_b} , which is a point on T^2 , describes the center of mass of the string in the directions of T^2 . Thus, a single-particle ground state with winding number \tilde{n} can be described by the location of the intersection of the classical string configuration with any particular T^2 fiber at a constant x_3 :

$$\left| \left[z, e^{i\nu} z, \dots, e^{(\tilde{n}-1)i\nu} z \right] \right\rangle, \tag{8.8}$$

where z coordinates are always taken modulo the lattice $\mathbb{Z} + \mathbb{Z}\tau$. The multi-string states can subsequently be described by

$$\left|\left\{[z_1, e^{i\upsilon}z_1, \dots, e^{(n_1-1)i\upsilon}z_1], [z_2, e^{i\upsilon}z_2, \dots, e^{(n_2-1)i\upsilon}z_2], \dots, [z_p, e^{i\upsilon}z_p, \dots, e^{(n_p-1)i\upsilon}z_p]\right\}\right\rangle,$$

where each z_i is a solution $\zeta_{M_{ai},M_{bi}}$ of (8.7) with $\tilde{n} \to n_i$, and $n = \sum_{1}^{p} n_i$ is the total winding number. Also, the number of inequivalent solutions of (8.7) for $\tilde{n} = 1$ is equal to k. It is a function of v alone, as indicated in (7.3).

8.3 \mathbb{Z}_k symmetries

For k > 1 there are two useful \mathbb{Z}_k symmetries that can be described geometrically in the type-IIA background as follows [1]:

1. The metric (8.5) has a discrete isometry that is generated by the operator \mathcal{U} defined to act as a translation in the T^2 fiber:

$$\mathcal{U}: \qquad (z, x_3, \zeta_1, \zeta_2, \zeta_3) \mapsto (z + \frac{1}{k}(1+\tau), x_3, \zeta_1, \zeta_2, \zeta_3). \tag{8.9}$$

2. The first homology group H_1 of the space (8.5) is $\mathbb{Z} \oplus \mathbb{Z}_k$ where \mathbb{Z}_k is generated by the homology class of one of the 1-cycles of the T^2 (in the z direction), and \mathbb{Z} is generated by a cycle that wraps around the x_3 direction at z = 0. The homology class of a fundamental string is conserved, and the projection onto the \mathbb{Z}_k factor describes a conserved quantum number $q \in \mathbb{Z}_k$. We define the operator \mathcal{V} to have the eigenvalue $e^{\frac{2\pi i}{k}q}$ on a state with quantum number q. In other words, \mathcal{U} can be viewed as \mathbb{Z}_k -momentum, and \mathcal{V} can be viewed as \mathbb{Z}_k -winding number. They obey the commutation relations [1]:

$$\mathcal{V}^{k} = \mathcal{U}^{k} = 1, \qquad \mathcal{V}\mathcal{U}\mathcal{V}^{-1}\mathcal{U}^{-1} = e^{\frac{2\pi i n}{k}}.$$
(8.10)

This $\mathbb{Z}_k \times \mathbb{Z}_k$ symmetry also has a natural interpretation in terms of the original gauge theory. Its conserved charges can be expressed in terms of S-duality invariant combinations of electric and magnetic fluxes in the center $U(1) \subset U(n)$; we refer the reader to [1] for details. For the present paper, however, we only need to know the commutation relations of \mathcal{U} and \mathcal{V} with the Wilson loop operators. Those follow directly from the relation of \mathcal{U} and \mathcal{V} to electric and magnetic fluxes, or can be derived directly in the type-IIA dual. The result will be given later on in the paper, in (11.34).

For completeness, we also note that in addition to this $\mathbb{Z}_k \times \mathbb{Z}_k$ symmetry we have an SL(2, \mathbb{Z}) symmetry that acts as the mapping class group of T^2 on which the Tr-S theory is defined. In the type-IIA dual picture, the complex structure parameter of this T^2 becomes the complexified area modulus of the type-IIA T^2 (in $x_{10}x_1$ directions):

$$\rho = \frac{i}{\alpha'_{\text{IIA}}} \operatorname{Area}(T^2) + \frac{1}{2\pi} \int_{T^2} B.$$
 (8.11)

Here B is the NS-NS two-form potential. The $SL(2,\mathbb{Z})$ group acts by T-duality, and is generated by

$$\mathcal{S} \to \begin{pmatrix} 0 & -1 \\ 1 & 0 \end{pmatrix} \in \mathrm{SL}(2, \mathbb{Z}), \qquad \mathcal{S}: \qquad \rho \to -\frac{1}{\rho}$$

and

$$\mathcal{T} \to \begin{pmatrix} 1 & 1 \\ 0 & 1 \end{pmatrix} \in \mathrm{SL}(2, \mathbb{Z}), \qquad \mathcal{T}: \qquad \rho \to \rho + 1.$$

In [1], this correspondence was used to read off the modular transformation properties of the ground states of Tr-S theory.

8.4 Is Tr-S a topological theory?

Before we proceed to study static charges for Tr-S on T^2 , let us explain in more detail why we believe Tr-S is a topological theory. The setting we described above has $\mathcal{N} = 6$ supersymmetry in 2 + 1D and a U(3) R-symmetry, which is the subgroup of SU(4) that commutes with γ in (8.1). If Tr-S is not topological and has propagating degrees of freedom, it must be an interacting $\mathcal{N} = 6$ superconformal field theory. Let us then consider the low-energy limit of Tr-S on S^1 . This compactification has $\mathcal{N} = (6, 6)$ supersymmetry in 1 + 1D. To gain more insight about this 1 + 1D theory, let us look at the list of dualities in Table 8.1, but instead of performing all the dualities all the way to type-IIA at the last line, let us stop one line before the last, at the point where we have M-theory and n M2-branes. At this point direction x_2 is not yet compact, but directions x_1, x_{10} are compact and form a torus with complex structure τ . The directions transverse to the M2-branes are parameterized by the complex coordinates $(z, \zeta_1, \zeta_2, \zeta_3)$. The M2-branes wrap direction x_3 , and the boundary conditions along x_3 are twisted by a geometrical twist, which is a rotation in the directions transverse to the M2-branes. This twist acts as

$$(z,\zeta_1,\zeta_2,\zeta_3) \to (e^{iv}z,e^{iv}\zeta_1,e^{iv}\zeta_2,e^{iv}\zeta_3) \tag{8.12}$$

and corresponds to a rotation by an angle v in 4 transverse planes.

For n = 1, the twisted boundary conditions create a mass gap of $1/\mathbf{r}R$, where $\mathbf{r} = 2\pi/v$ [see the discussion below (7.2)], and the 1 + 1D low-energy theory has no propagating degrees of freedom. This is consistent with the identification of Tr-S at n = 1 with abelian Chern-Simons theory, as will be discussed in Chapter 10 in more detail, and indeed Chern-Simons theory has no propagating degrees of freedom.

What about the nonabelian case, say n = 2? In this case we need to understand the low-energy limit describing two M2-branes compactified on S^1 with transverse directions $T^2 \times \mathbb{C}^3$ parameterized by $(z, \zeta_1, \zeta_2, \zeta_3)$ and with boundary conditions twisted by (8.12) along S^1 . The 1 + 1D low-energy theory corresponds to configurations where $(z, \zeta_1, \zeta_2, \zeta_3)$ are independent of x_3 . Because of the twist, this implies that $\zeta_1 = \zeta_2 = \zeta_3 = 0$ and z is a fixed point of the twist. For given v, the twist has k fixed points on T^2 , as we explained at the end of §8.2. It is easy to check that these fixed points are at

$$z_j = \frac{j}{k}(1+\tau), \qquad j = 0, \dots, k-1.$$
 (8.13)

The 1 + 1D low-energy theory therefore has k(k + 1)/2 sectors. In k(k - 1)/2 of the sectors the two M2-branes sit at different fixed points $z_j \neq z_{j'}$. In this case it is clear that no massless excitations survive the low-energy limit and the low-energy 1 + 1D theory has no propagating degrees of freedom. The remaining k sectors have two M2-branes at the same z_j . Clearly, all these sectors are equivalent and we can concentrate on one of them, say at $z_0 = 0$. Since the M2-branes are pinned to the origin, we can safely replace T^2 with \mathbb{C} and set $z \to \zeta_0$, with $\zeta_0 \in \mathbb{C}$. We have now reduced the problem to understanding the compactification of two M2-branes on S^1 with transverse directions \mathbb{C}^4 and a twist along S^1 given by

$$(\zeta_0, \zeta_1, \zeta_2, \zeta_3) \to (e^{i\upsilon}\zeta_0, e^{i\upsilon}\zeta_1, e^{i\upsilon}\zeta_2, e^{i\upsilon}\zeta_3). \tag{8.14}$$

Up until recently we would have had to proceed indirectly from here, but the recent progress in the low-energy description of M2-branes [47]-[48], culminating in the discovery of the ABJM action [49], allows us, in principle, to explore this problem directly. We need to take the ABJM action and compactify all fields on S^1 with boundary conditions twisted by (8.14). This corresponds to an element of the SO(8) R-symmetry group. However, proceeding to compactify the ABJM theory in this manner involves subtleties that require a separate treatment, which we hope to present elsewhere. Instead, for now we will settle for an indirect approach, modifying the problem a little bit. Instead of taking the discrete value $v = 2\pi/\mathbf{r}$ in (8.14), let us consider the case that $|v| \ll 1$. More precisely, consider the double-scaling limit

$$v \to 0$$
, $R \to 0$, $\beta \equiv \frac{v}{R} \to \text{const.}$

Using a standard technique, we change variables to

$$z'_j \equiv e^{-\frac{ivx_3}{2\pi R}} \zeta_j, \qquad j = 0, 1, 2, 3,$$

and write the metric as

$$ds^{2} = -dx_{0}^{2} + dx_{2}^{2} + dx_{3}^{2} + \sum_{j=0}^{3} |d\zeta_{j}|^{2}$$

= $-dx_{0}^{2} + dx_{2}^{2} + \left(1 + \frac{\beta^{2}}{4\pi^{2}} \sum_{0}^{3} |z_{j}'|^{2}\right) dx_{3}^{2} + \sum_{j=0}^{3} |dz_{j}'|^{2} - \frac{\beta}{\pi} dx_{3} \operatorname{Im} \sum_{j=0}^{3} z_{j}' d\overline{z}_{j}'$

We can now reduce to type-IIA along direction x_3 to obtain a "Melvin background" [7]-[50]. This background has a Ramond-Ramond field strength

$$F^{RR} = dA^{\mathrm{RR}} = -\frac{\beta}{2\pi} \mathrm{Im} \sum_{j=0}^{3} dz'_{j} \wedge d\overline{z}'_{j} + O(\beta^{3}) \,,$$

and a dilaton

$$e^{\Phi} = (M_p R)^{3/2} \left(1 + \frac{\beta^2}{4\pi^2} \sum_{0}^{3} |z'_j|^2\right)^{3/2},$$

where M_p is the 10+1D Planck scale. This background creates mass terms for all low-energy fields that propagate on a long string in direction x_2 . Such a string is pinned to the origin $z'_0 = z'_1 = z'_2 = z'_3$, as is obvious from the M-theory description. In the type-IIA world-sheet description, the fermion mass terms are generated from the coupling of the string modes to the RR field strength, while the bosonic mass terms are generated from the string-frame metric. For small β we can trust the approximate perturbative string analysis, and we see that all propagating modes along the remaining noncompact direction x_2 have mass of order $\beta = v/R$. It is not immediately clear that we can extrapolate this analysis to $v = 2\pi/\mathbf{r}$, but we know that for a single string in this background no complications should arise, and in the limit $R \to 0$ string interactions are small.

Furthermore, if Tr-S is a nontrivial SCFT, and if it has a moduli space, then this moduli space must be compact because we have eliminated all noncompact modes along the Coulomb branch via the twist, and because the type-IIA picture shows no trace of noncompact flat directions. But $\mathcal{N} = 6$ supersymmetry in 2 + 1D is very restrictive and requires the moduli space to be locally flat. It must therefore be of the form T^{8d}/Γ , where Γ is a discrete isometry group, and d is an integer. On the other hand, the unbroken R-symmetry group must act nontrivially on the moduli space (by supersymmetry), but the maximal continuous isometry group of T^{8d}/Γ is abelian and cannot contain SU(3), which is a contradiction.

We will proceed under the assumption that Tr-S is topological, but we note that even if this assumption is incorrect, the results of this paper are still meaningful, but they then correspond to the Witten Index of an interacting SCFT, rather than a TQFT. We now proceed to the calculation of the Witten Index.

Chapter 9

Static charges and their type-IIA dual description

In this paper we study what happens when we insert static charges into the Tr-S theory defined in Chapter 8. We add 2m static external sources to the system at S^1 coordinate $x_3 = 0$. Specifically, we insert m heavy (non dynamical) quarks at the fixed T^2 coordinates $(a_1^{(\mathfrak{a})}, a_2^{(\mathfrak{a})})$ (where $\mathfrak{a} = 1, \ldots, m$), and to cancel the net U(1) charge¹ we insert an equal number m of anti-quarks, which we take to be at fixed T^2 coordinates $(a_1^{(\mathfrak{a}+m)}, a_2^{(\mathfrak{a}+m)})$. (Here $0 \leq a_i^{(\mathfrak{a})} < 2\pi L_i$ (i = 1, 2) are periodic coordinates in type-IIB directions x_1, x_2 .)

The \mathfrak{a}^{th} static charge can be incorporated by introducing a matrix element of a time-like Wilson line

$$\operatorname{tr} P \exp\left(i \int_{-\infty}^{\infty} A_0(t, a_1^{(\mathfrak{a})}, a_2^{(\mathfrak{a})}, 0) dt\right)$$
(9.1)

into the path integral. This prescription, however, breaks all the supersymmetry. To preserve some supersymmetry we follow [51, 52] and add one of the adjoint scalar fields of $\mathcal{N} = 4$ SYM to A_0 in (9.1). For concreteness, we take

tr
$$P \exp\left(i \int_{-\infty}^{\infty} [A_0(t, a_1^{(\mathfrak{a})}, a_2^{(\mathfrak{a})}, 0) + \Phi^9(t, a_1^{(\mathfrak{a})}, a_2^{(\mathfrak{a})}, 0)]dt\right)$$
, (9.2)

where Φ^9 is the scalar field that corresponds to D3-brane fluctuations in the x_9 direction. In §9.1 we will show that inserting charges that interact with Tr-S as the low-energy limit of (9.2) preserves 4 real supercharges.

¹ Actually, it is not necessary for the net charge to be zero, thanks to the S-duality twisted boundary conditions in the x_3 direction. But the system with nonzero net charge is more complicated and will not be studied here.

9.1 Charges as endpoints of type-IIB strings

Our main task now is to identify the type-IIA dual realization of the charges. We start in type-IIB and follow a standard technique to introduce static charges with interactions (9.2) into the type-IIB construction described at the beginning of Chapter 8.

Following [51, 52] we introduce fundamental strings with one endpoint on the D3branes and extending indefinitely in direction x_9 . We label the strings by $\mathbf{c} = 1, \ldots, 2m$ and let the strings labeled by $\mathbf{c} = 1, \ldots, m$ extend along $0 \le x_9 < \infty$ and the strings labeled by $\mathbf{c} = m + 1, \ldots, 2m$ extend along $-\infty < x_9 \le 0$. The low-energy description of this system holds the information about the ground states of Tr-S with static charges, and the (x_1, x_2) coordinates of the endpoints of the strings can be set to $(a_1^{(\mathbf{c})}, a_2^{(\mathbf{c})})$. We are only interested in the low-energy excitations of the system at energies well below the string scale, as well as the compactification scales:

$$E \ll M_{\rm st}, \frac{1}{L_1}, \frac{1}{L_2}, \frac{1}{R}$$

The semi-infinite strings, in this limit, are very heavy and can be treated semiclassically.

The long-wavelength excitations of each string are described by 8 free scalars $X^{\mu}_{\mathfrak{c}}(x_9,t)$ ($\mu = 1,\ldots,8$) and a free Majorana-Weyl fermion $\psi_{\mathfrak{c}}$ satisfying the chirality condition

$$\Gamma^{0123456789}\psi_{\mathfrak{c}}=\psi_{\mathfrak{c}}$$

and the free massless Dirac equation along the string:

$$(\Gamma^0 \partial_t + \Gamma^9 \partial_9) \psi_{\mathfrak{c}} = 0.$$

Their low-energy effective action is of the form

$$I = I^{(\text{int})} + \sum_{\mathfrak{c}=1}^{2m} I_{\mathfrak{c}}^{(F1)}, \qquad (9.3)$$

where $I_{\mathfrak{c}}$ is the bulk 1 + 1D action of the free fields $X^{\mu}_{\mathfrak{c}}, \psi_{\mathfrak{c}}$, and $I^{(\text{int})}$ is the 0+1D action that couples the fields $X^{\mu}_{\mathfrak{c}}, \psi_{\mathfrak{c}}$ at $x_9 = 0$ to the low-energy degrees of freedom of Tr-S theory. In addition to the boundary values of $X^{\mu}_{\mathfrak{c}}, \psi_{\mathfrak{c}}$, the interaction term $I^{(\text{int})}$ depends on additional local 0+1D degrees of freedom, whose form we seek to find. (See Figure 9.1.)

We will now provide a preview of what $I^{(\text{int})}$ looks like, and we will explain the derivation at length in the following subsections. The term $I^{(\text{int})}$ is formulated in terms of additional variables that are localized at the interaction point $x_9 = 0$. These variables include a discrete variable that specifies the "sector," with a finite number of sectors altogether. Each sector is then described by m periodic variables \mathfrak{p}^a (a =



Figure 9.1: External quark and anti-quark sources are realized as endpoints of fundamental strings. At low-energy, the strings are described by free 1 + 1D fields $X^{\mu}_{\mathfrak{c}}(x_9, t), \psi_{\mathfrak{c}}(x_9, t)$ and the low-energy modes of the compact interacting system of D3-branes and charges are described by periodic variables $\mathfrak{p}^a(t), \mathfrak{q}^{\alpha}(t)$.

 $1, \ldots, m$) and additional m' periodic variables \mathfrak{q}^{α} ($\alpha = 1, \ldots, m'$), both of which take values in the range $[0, 2\pi)$. The number m' of \mathfrak{q}^{α} 's depends on the sector, but generally $m' \geq m$. The action $I^{(\text{int})}$ is then a sum of two terms, which we write schematically as:

$$I^{(\text{int})} = I_0 + I_1, \qquad I_0 \equiv I_0(\{\mathfrak{p}^a, \mathfrak{q}^\alpha\}), \qquad I_1 \equiv I_1(\{\mathfrak{p}^a, \mathfrak{q}^\alpha\}, \{X^1_{\mathfrak{c}}(0), X^2_{\mathfrak{c}}(0)\}).$$

The first term I_0 describes the local system at $x_9 = 0$, while I_1 is the interaction term that couples the system to the boundary values of the scalar fields on the 2mfundamental strings. (The fermions will be discussed later, but are suppressed at the moment.)

For the sectors for which m' = m the configuration space of $\{\mathfrak{p}^{\mathfrak{a}}, \mathfrak{q}^{\alpha}\}$ is T^{2m} , and the system described by I_0 is equivalent to geometric quantization on T^{2m} with the following action:

$$I_0^{(\text{g.q.})} = \frac{1}{2\pi} \int \sum_{a,\alpha=1}^m \mathcal{M}_{\mathfrak{a}\alpha} \mathfrak{p}^{\mathfrak{a}} \dot{\mathfrak{q}}^{\alpha} dt , \qquad (9.4)$$

where $\mathcal{M}_{\mathfrak{a}\alpha}$ is a nonsingular matrix of integers that we will describe below. As mentioned in the Introduction, the sectors of most interest to us will be of this form. Any additional kinetic terms that are quadratic in $\dot{\mathfrak{q}}^{\alpha}, \dot{\mathfrak{p}}^{\mathfrak{a}}$ are irrelevant at low-energy. The other sectors, with m' > m, also have a piece of the form $I_0^{(g.q.)}$ in their action, but it is necessary to include additional kinetic terms. The coupling term I_1 is linear in $X^1_{\mathfrak{c}}, X^2_{\mathfrak{c}}$ and the derivatives $\dot{\mathfrak{q}}^{\alpha}, \dot{\mathfrak{p}}^{\mathfrak{a}}$. It is of the form:

$$I_1 = \frac{1}{2\pi} \int \sum_{\mathfrak{c}\alpha} \mathcal{N}_{\mathfrak{c}\alpha} X^1_{\mathfrak{c}}(0) \dot{\mathfrak{g}}^{\alpha} dt + \frac{1}{2\pi} \int \sum_{\mathfrak{c}\mathfrak{a}} \mathcal{K}_{\mathfrak{c}\mathfrak{a}} X^2_j(0) \dot{\mathfrak{p}}^{\mathfrak{a}} dt , \qquad (9.5)$$

where $\mathcal{N}_{c\alpha}, \mathcal{K}_{c\alpha}$ are matrices of integers to be specified later. The remaining fields X_c^{μ} with $\mu = 4, \ldots, 8$ have Dirichlet boundary conditions $X_c^{\mu}(0) = 0$, while X_c^3 has Neumann boundary conditions. These fields are however irrelevant for our discussion. In §9.3 we will explain how the interactions (9.3)-(9.5) are derived from the type-IIA dual. But first, let us discuss how much supersymmetry is left.

9.2 Supersymmetry

To see how much supersymmetry is preserved we consider once again the realization of (9.2) in type-IIB. We have fundamental strings that stretch along direction x_9 and end on the *n* D3-branes. Let Γ^A (A = 0, ..., 9) be 9+1D Dirac gamma matrices, which we take to be real. Let $\epsilon = \epsilon_1 + i\epsilon_2$ be a complex 9+1D Weyl spinor, where ϵ_1, ϵ_2 are Majorana-Weyl, and $\epsilon^* = \epsilon_1 - i\epsilon_2$ its complex conjugate.

The supersymmetry preserved by the n D3-branes is parameterized by those combinations of the supercharges with coefficients ϵ that satisfy:

$$\Gamma^{0123}\epsilon = -i\epsilon \,. \tag{9.6}$$

The S-R-twist preserves

$$\epsilon = e^{-\frac{iv}{2}} e^{\frac{v}{2}(\Gamma^{45} + \Gamma^{67} + \Gamma^{89})} \epsilon \,, \tag{9.7}$$

where the first factor is from the S-twist, and the second from the R-twist, and the interaction (9.2) preserves the same combinations of supersymmetry generators that a fundamental string in directions 0, 9 preserves, which is given by

$$\Gamma^{09}\epsilon = \epsilon^* \,. \tag{9.8}$$

Combining (9.8) and (9.7) we find

$$\epsilon^* = e^{\frac{i\upsilon}{2}} e^{\frac{\upsilon}{2}(\Gamma^{45} + \Gamma^{67} - \Gamma^{89})} \epsilon^* ,$$

while taking complex conjugate of (9.7) yields (keeping in mind that the gamma matrices are all real)

$$\epsilon^* = e^{-\frac{iv}{2}} e^{\frac{v}{2}(\Gamma^{45} + \Gamma^{67} + \Gamma^{89})} \epsilon^* \,.$$

Together, these two equations imply

$$\Gamma^{89}\epsilon = i\epsilon\,.$$

Brane	1	3	4	5	6	7	8	9	10
F1		÷							
D2								H	—

Table 9.1: Open D2-branes are the U-duals of the type-IIB strings. Here – denotes a direction that the brane wraps, \div denotes a direction that the brane/string wraps with a twist, and \vdash denote a direction in which the brane extends but with endpoints (see §9.9).

Then, from (9.7), we obtain

$$\Gamma^{45}\epsilon = -\Gamma^{67}\epsilon\,,\tag{9.9}$$

and together with (9.6) this leaves four linearly independent complex supersymmetry parameters. But (9.8) then puts a reality condition on these parameters, and leaves four real supercharges unbroken.

Out of the U(3) R-symmetry that is preserved by the R-twist (8.1), the static charges only preserve $SU(2) \times U(1) \subset U(3)$. This is the double-cover of the unitary group $U(2) \simeq [SU(2) \times U(1)]/\mathbb{Z}_2$ that acts as unitary rotations of the variables $x_4 + ix_5$, $x_6 + ix_7$, and preserves $x_8 + ix_9$. The surviving supercharges transform as a doublet of SU(2) and are neutral under U(1) (which is generated by $\Gamma^{45} + \Gamma^{67}$).

9.3 Constructing the type-IIA dual of charges

Now we transform the system of D3-branes and fundamental strings to type-IIA by applying the U-duality transformation described in Table 8.1. The 2m fundamental strings turn into D2-branes, and the n D3-branes turn into fundamental strings, as listed in Table 9.1. In the type-IIB picture the strings end on the n D3-branes, but in the type-IIA picture the D2-branes are too big to end on the n strings. The system must therefore rearrange itself, and we have to pair up each D2-brane that corresponds to a quark (extending in the positive x_9 direction) with a D2-brane that corresponds to an anti-quark (extending in the negative x_9 direction), and glue them into a single smooth D2-brane. We thus get m D2-branes whose worldvolume is equivalent to an infinite cylinder.

Some of the type-IIA closed strings that we had in Chapter 8 must now be allowed to break into open strings and end on the D2-branes. Every D2-brane must have at least one such open string attached to it, because otherwise the corresponding type-IIB string would not be bound to any of the n D3-branes. For ease of discussion it will be convenient to slightly separate the D2-branes in the x_3 direction. The resulting configuration is depicted in Figure 9.2.



Figure 9.2: The type-IIA configuration following the U-duality transformation of Table 8.1. The m pairs of type-IIB open strings become m continuous D2-branes. The n D3-branes become n fundamental strings, which in the presence of the D2-branes can break up into open strings. At least one pair of open strings must be attached to each D2-brane. In this example m = 4 and n = 2.

9.4 Local $\mathfrak{p}^{\mathfrak{a}}, \mathfrak{q}^{\alpha}$ variables and their action I_0

To understand how the $\mathfrak{p}^{\mathfrak{a}}, \mathfrak{q}^{\alpha}$ that appear in (9.4) arise, it is instructive to start with a simple example: one D2-brane (m = 1) and one string (n = 1) for the case k = 2 with $v = \frac{\pi}{2}$. We have one fundamental string wrapping direction x_3 and bound to the D2-brane, which wraps direction x_{10} and extends in direction x_9 . The system is therefore dual to U(1) Tr-S theory with an external quark and anti-quark pair.

To bind to a D2-brane, the closed string must break to become an open string that starts and ends on the D2-brane located at $x_3 = 0$. Let the string start at T^2 coordinate $z = x_{10} + ix_1 \equiv \mathbf{q} + i\mathbf{u}$ on the D2-brane, where \mathbf{u} and \mathbf{q} are functions of time. To make \mathbf{u} and \mathbf{q} compact variables with period 2π , we will rescale the x_{10} and x_1 coordinates so that from now on they take values in $[0, 2\pi)$. To be of minimal length the string must remain at $x_9 = 0$ and at constant z, as x_3 varies from 0 to $2\pi R$. The point with coordinates $x_3 = 2\pi R$ and $z = \mathbf{q} + i\mathbf{u}$ is equivalent in the geometry of (8.5) to the point with coordinates $x_3 = 0$ and $z = e^{iv}(\mathbf{q} + i\mathbf{u}) = -\mathbf{u} + i\mathbf{q}$. This point must also be on the D2-brane, which wraps direction x_{10} but is at a fixed x_1 coordinate. We thus find that $\mathbf{u} = \mathbf{q}$. In other words, the bound fundamental string starts at $z' = (1 + i)\mathbf{q}$ on the D2-brane, and ends at $z'' = (-1 + i)\mathbf{q}$ on the same D2-brane (see Figure 9.3).

The starting point and endpoint of the string are oppositely charged under the U(1) gauge field that resides on the D2-brane. Let A_{10} be the component of this



Figure 9.3: (a) A fundamental string (F1) bound to the D2-brane. The D2-brane wraps the compact direction x_{10} (the vertical direction) and extends indefinitely in direction x_9 (not shown in the picture). The fundamental string is at $x_9 = 0$ and extends in direction x_3 (perpendicular to the plane of the drawing). Because of the S-duality twist, which in the type-IIA picture translates to a rotation, the fundamental string's endpoint z'' can be different from its starting point z'. (b) Configurations of the D2-brane with the two endpoints z', z'' of the string marked as oppositely charged points. As the D2-brane changes its x_1 -position, the positions of the charges change accordingly.

gauge field in the x_{10} direction. We can fix the gauge so that A_{10} is independent of x_{10} , and set $\mathbf{p} \equiv 2\pi A_{10}$. The starting point and endpoint of the string are separated along the x_{10} direction by $\Delta x_{10} = 2\mathbf{q}$, so the action of the system includes the term:

$$I_0 \equiv 2 \int A_{10} d\mathfrak{q} = \frac{1}{2\pi} \int 2\mathfrak{p} d\mathfrak{q} \,. \tag{9.10}$$

We claim that I_0 is the only relevant term at low-energy. For example, the kinetic energies of the fundamental string and of the D2-brane are irrelevant at low-energy, because they are proportional to $(\partial_0 \mathbf{q})^2$. Indeed, setting the mass dimensions of \mathbf{p}, \mathbf{q} to zero, we see that the kinetic term has dimension 2 and is irrelevant in a 0+1D theory.

The action I_0 describes the geometric quantization of a torus with the symplectic form $2d\mathfrak{p} \wedge d\mathfrak{q}$. The Hilbert space has two states, and this is indeed the number of states we expect, since the U(1) Tr-S theory is known to be equivalent to Chern-Simons theory at level k. (See Chapter 10 for more details.)

We can now turn to the general case. The 0+1D variables \mathfrak{p}^a , \mathfrak{q}^{α} that appear in (9.4) arise out of the type-IIA picture as follows. The low-energy description of each of the *m* D2-branes includes a gauge field $A^{(\mathfrak{a})}$ ($\mathfrak{a} = 1, \ldots, m$). The periodic variable

 $\mathfrak{p}^{\mathfrak{a}}$ is identified with the holonomy of $A^{(\mathfrak{a})}$ around the x_{10} circle at $x_9 = 0$:

$$\mathfrak{p}^{\mathfrak{a}}(t) = \left. \int_{0}^{2\pi} A_{10}^{(\mathfrak{a})} dx_{10} \right|_{x_{9}=0} \,. \tag{9.11}$$

In the presence of the D2-branes, fundamental strings can break into open strings. A string doesn't have to break at every D2-brane, but for ease of notation let us first assume that each of the *n* fundamental strings does break at every D2-brane. We thus have $n \times (m + 1)$ open string segments. Each segment must be located at a constant x_{10} in order for its length to be minimal. We denote the x_{10} coordinates of the open string segments by variables $\mathbf{q}^{\mathbf{i}\mathbf{a}'}$ (with $\mathbf{i} = 1, \ldots, n$ and $\mathbf{a}' = 0, \ldots, m$). The string segments are ordered via the index \mathbf{a}' in the direction of increasing x_3 . (See Figure 9.4 for an illustration.) In addition to the x_{10} coordinates we also need to know the x_1 coordinates of the strings. We denote them by $\mathbf{u}^{\mathbf{i}\mathbf{a}'}$:

$$(x_1, x_{10}) \rightarrow (\mathfrak{u}^{\mathfrak{i}\mathfrak{a}'}, \mathfrak{q}^{\mathfrak{i}\mathfrak{a}'}).$$

However, as noted above, an open string doesn't have to break at every D2-brane. For a given state, we define the *binding matrix* \mathfrak{B} to be the following $n \times m$ matrix of 0 and 1's that encodes which strings break and which do not:

$$\mathfrak{B}_{\mathfrak{i}\mathfrak{a}} = \begin{cases} 1 & \text{if the } \mathfrak{i}^{th} \text{ string is bound to the } \mathfrak{a}^{th} \text{ D2-brane,} \\ 0 & \text{otherwise,} \end{cases}$$
(9.12)

for $i = 1, \ldots, n$ and $a = 1, \ldots, m$. We therefore have

$$\mathfrak{u}^{\mathfrak{i}\mathfrak{a}} = \mathfrak{u}^{\mathfrak{i}(\mathfrak{a}-1)}, \quad \mathfrak{q}^{\mathfrak{i}\mathfrak{a}} = \mathfrak{q}^{\mathfrak{i}(\mathfrak{a}-1)}, \qquad \text{if } \mathfrak{B}_{\mathfrak{i}\mathfrak{a}} = 0,$$

which indicates that the i^{th} string is continuous at the \mathfrak{a}^{th} brane. On the other hand, at every break point (i, \mathfrak{a}) on the \mathfrak{a}^{th} D2-brane the i^{th} string is charged under the gauge field on the \mathfrak{a}^{th} D2-brane $A^{(\mathfrak{a})}$, and the $A_{10}^{(\mathfrak{a})} dx_{10}$ interaction of the gauge field with the charged particle produces a term proportional to $\mathfrak{p}^{\mathfrak{a}} d\mathfrak{q}^{i\mathfrak{a}}$ in the effective action. The sign of this term is positive for one end of the string and negative for the other, as the charges of the two ends are opposite. Thus an open segment of the i^{th} string that starts on the \mathfrak{a}^{th} D2-brane and ends on the \mathfrak{b}^{th} D2-brane contributes $\mathfrak{p}^{\mathfrak{a}} d\mathfrak{q}^{i\mathfrak{a}} - \mathfrak{p}^{\mathfrak{b}} d\mathfrak{q}^{i(\mathfrak{b}-1)}$ to the action.

On top of this, direction x_3 is compact with $x_3 \in [0, 2\pi R]$ and the S-duality and R-symmetry operators act at $x_3 = 0$. The S-R-twisted boundary conditions induce linear relations between the x_1 and x_{10} coordinates of the strings they connect, where we are free to add a permutation $\sigma \in S_n$ among the *n* strings before applying the S-R-twist. A particular sector of the Hilbert space is thus described by the $\mathfrak{B}_{i\mathfrak{a}}$ matrix, as well as the permutation $\sigma \in S_n$. We collect this information in a matrix and denote



Figure 9.4: A variable $\mathfrak{p}^{\mathfrak{a}}$ ($\mathfrak{a} = 1, \ldots, m$) is associated with each D2-brane, ordered in the direction of increasing x_3 . Let every string break at every brane into a total of m + 1 open segments. The constant x_{10} coordinate of each of these segments is denoted by $\mathfrak{q}^{\mathfrak{i}\mathfrak{a}'}$ with $\mathfrak{i} = 1, \ldots, n$ and $\mathfrak{a}' = 0, \ldots, m$ in the direction of increasing x_3 . However, the strings actually break only at positions marked with a • and the main contribution to the action comes from these break points. The strings are coincident, but we separated them in the picture for clarity.

a given sector as

$$\begin{bmatrix} \mathfrak{B}_{11} & \mathfrak{B}_{12} & \cdots & \mathfrak{B}_{1m} & \sigma(1) \\ \mathfrak{B}_{21} & \mathfrak{B}_{22} & \cdots & \mathfrak{B}_{2m} & \sigma(2) \\ \vdots & \vdots & \ddots & \vdots & \vdots \\ \mathfrak{B}_{n1} & \mathfrak{B}_{n2} & \cdots & \mathfrak{B}_{nm} & \sigma(n) \end{bmatrix}$$

Obviously, two sectors $[\mathfrak{B}_{\mathfrak{i}\mathfrak{a}}|\sigma]$ and $[\mathfrak{B}_{\sigma'(\mathfrak{i})\mathfrak{a}}|\sigma'\circ\sigma\circ\sigma'^{-1}]$ are equivalent (with $\sigma'\in S_n$). The boundary conditions can now be written as

$$e^{i\upsilon}(\mathfrak{q}^{im} + \tau\mathfrak{u}^{im}) = \mathfrak{q}^{\sigma(i)0} + \tau\mathfrak{u}^{\sigma(i)0} \pmod{2\pi(\mathbb{Z} + \tau\mathbb{Z})}.$$
(9.13)

Using (7.2) and $\tau = \frac{\mathbf{a}\tau + \mathbf{b}}{\mathbf{c}\tau + \mathbf{d}}$ we can rewrite (9.13) as an equation with integer coefficients:

$$\mathfrak{q}^{\sigma(\mathfrak{i})0} = \mathbf{d}\mathfrak{q}^{\mathfrak{i}m} + \mathbf{b}\mathfrak{u}^{\mathfrak{i}m}, \qquad \mathfrak{u}^{\sigma(\mathfrak{i})0} = \mathbf{c}\mathfrak{q}^{\mathfrak{i}m} + \mathbf{a}\mathfrak{u}^{\mathfrak{i}m}.$$
(9.14)

For example, for $v = \frac{\pi}{2}$ (k = 2), this becomes

$$\mathfrak{u}^{im} = -\mathfrak{q}^{\sigma(i)0}, \qquad \mathfrak{q}^{im} = \mathfrak{u}^{\sigma(i)0} \pmod{2\pi\mathbb{Z}}.$$

Finally, let $\mathfrak{u}^{\mathfrak{a}}$ ($\mathfrak{a} = 1, \ldots, m$) be the x_1 coordinate of the \mathfrak{a}^{th} D2-brane. Then we have the equations

$$\mathfrak{u}^{\mathfrak{a}} = \mathfrak{u}^{\mathfrak{i}\mathfrak{a}} = \mathfrak{u}^{\mathfrak{i}(\mathfrak{a}-1)}, \quad \text{whenever } \mathfrak{B}_{\mathfrak{i}\mathfrak{a}} = 1, \quad (9.15)$$

since the i^{th} string connects with the \mathfrak{a}^{th} D2-brane. The equations (9.13)-(9.15) reduce the total number of independent $\mathfrak{q}^{i\mathfrak{a}'}$ variables. A linearly independent basis can be chosen, and these furnish the \mathfrak{q}^{α} variables in (9.4). In Chapter 10-Chapter 11 we will present explicit detailed examples.

Congested and decongested matrices

We say that a binding matrix $\mathfrak{B}_{i\mathfrak{a}}$ is *congested* if there is at least one \mathfrak{a} for which there are two distinct $\mathfrak{i} \neq \mathfrak{j}$ such that $\mathfrak{B}_{\mathfrak{i\mathfrak{a}}} = \mathfrak{B}_{\mathfrak{j\mathfrak{a}}} = 1$. This means that there is at least one D2-brane from which at least two strings emanate. If every D2-brane has exactly one string emanating from it, we say that the binding matrix is *decongested*. In this case, for every $\mathfrak{a} = 1, \ldots, m$ there is exactly one \mathfrak{i} for which $\mathfrak{B}_{\mathfrak{i\mathfrak{a}}} = 1$. The difference between congested and decongested binding matrices will become relevant when we discuss fermionic zero modes in §11.2.

9.5 The interaction term I_1

We will now derive the interaction of the 0+1D variables $\mathfrak{p}^{\mathfrak{a}}, \mathfrak{q}^{\alpha}, \mathfrak{u}^{\mathfrak{a}}$ with the 1+1D fields. The low-energy 1+1D fields that are relevant for the present discussion can

be described either in type-IIA or in type-IIB. In type-IIB, these fields are the two scalars $X^1(x_9, t), X^2(x_9, t)$ (with the index **c** of the string suppressed). In type-IIA, the two relevant low-energy fields on the D2-brane are the gauge field component A_{10} and the x_1 coordinate of the D2-brane, which we denote by Φ^1 . Note that direction 1 in type-IIA is related to direction 1 in type-IIB via T-duality. Following the U-duality of Table 8.1, it is then easy to see that X^1, X^2 are the duals (as 1 + 1D free compact scalar fields) of Φ^1, A_{10} , respectively:

$$\partial_9 X^1 = \partial_t \Phi^1, \quad \partial_t X^1 = -\partial_9 \Phi^1, \quad \partial_9 X^2 = \partial_t A_{10}, \quad \partial_t X^2 = \partial_9 A_{10}. \tag{9.16}$$

From this simple observation, it is easy to derive the requisite interaction between X^1, X^2 and $\mathfrak{p}^{\mathfrak{a}}, \mathfrak{q}^{\alpha}, \mathfrak{u}^{\mathfrak{a}}$. In the type-IIA picture $\{\mathfrak{p}^{\mathfrak{a}}, \mathfrak{u}^{\mathfrak{a}}\}$ determine the boundary conditions at $x_9 = 0$ of Φ^1, A_{10} . If \mathfrak{a} is the index of the brane, then by definition, we have the Dirichlet boundary conditions:

$$2\pi A_{10}(x_9 = 0, t) = \mathbf{p}^{\mathfrak{a}}(t) ,$$

$$\Phi^1(x_9 = 0, t) = \mathbf{u}^{\mathfrak{a}}(t) ,$$

where we suppressed brane indices on the left-hand sides. The duality (9.16) then converts the Dirichlet boundary conditions of type-IIA to Neumann boundary conditions of type-IIB. The latter can be incorporated into the action with the addition of the term

$$\int (\mathfrak{u}^{\mathfrak{a}} dX^1 + \mathfrak{p}^{\mathfrak{a}} dX^2) \,. \tag{9.17}$$

For most sectors the $\mathfrak{u}^{\mathfrak{a}}$'s can be written as linear combinations of the \mathfrak{q}^{α} 's by using the various constraints discussed at the end of §9.4. In these cases the sum of the terms (9.17) for all the D2-branes takes the form of I_1 in (9.5).

In the present paper we will not have much use for the interaction term I_1 , since we keep the charge coordinates X^1, X^2 constant. We have nevertheless presented it here for completeness. In a future work we hope to explore the dependence of the Hilbert space on the position of the charges, and the term I_1 will then play a central role. We conclude the discussion of the interaction term by presenting a more geometrical interpretation of I_1 .

D2-**F**1 intersections

The interaction term $\mathfrak{p}^{\mathfrak{a}} dX^2$ in (9.17) has a simple interpretation in terms of the geometry of the lift of the D2-branes and F1-strings to M-theory. To see this, let us focus on a single F1-string and a single D2-brane. We start by recalling some facts about this system, following the techniques developed in [53]. There, a configuration of D4-branes and NS5-branes was analyzed by lifting it to M-theory. Here, we need to analyze a similar configuration of fundamental strings and D2-branes, and we are also going to lift it to M-theory.



Figure 9.5: (a) A junction of two open fundamental strings, one starting and one ending on a D2-brane. (b) The D2-brane wraps direction $10 \equiv \natural$ and the configuration can be deformed so that it lifts to a smooth holomorphic curve in M-theory.

The lift essentially brings us back to the second-from-last row of Table 8.1, and the M-theory direction in the present discussion is therefore denoted by x_2 . We assume that x_2, x_{10} are periodic with period 2π . Let us also assume for simplicity that the length of the x_3 direction is very large, so that $-\infty < x_3 < \infty$. (That is, we will suspend the effect of S-R-twist for the moment.) Following [53], we define complex variables

$$v = e^{x_9 + ix_{10}}, \qquad u = e^{x_3 + ix_2},$$

The configuration of an F1-string intersecting a D2-brane then lifts to a single M2brane which extends along the locus of the complex equation

$$v = \frac{u - e^{i\alpha}}{u - e^{-i\alpha}},\tag{9.18}$$

where α is a real constant. Equation (9.18) is designed so that as $x_9 \to \infty$ (where v has a pole) $x_2 = -\alpha$, and as $x_9 \to -\infty$ (where v has a zero) $x_2 = \alpha$ (see Figure 9.5b). Equation (9.18) also tells us that as $x_3 \to \infty$ we have v = 1, while as $x_3 \to -\infty$ we have $v = e^{2i\alpha}$. If we now let q^1 , q^2 be the x_{10} coordinates of the string between $x_3 = \infty$ and $x_3 = -\infty$, then we have

$$\Delta x_{10} \equiv \mathfrak{q}^2 - \mathfrak{q}^1 = 2\alpha = -X^2(x_9 = +\infty) + X^2(x_9 = -\infty) \mod 2\pi.$$
(9.19)

Going to the low-energy limit, we can interpret the boundary condition $X^2(x_9 = +\infty)$ as the boundary value at $x_9 = 0$ of the X^2 field on the $x_9 > 0$ portion of the string (which we denoted by $X^2_{\mathfrak{a}}$), and similarly the boundary condition $X^2(x_9 = -\infty)$ is the low-energy boundary value at $x_9 = 0$ of the X^2 field on the $x_9 < 0$ portion of the string (which we denoted by $X^2_{\mathfrak{a}+m}$). The geometrical equation (9.19) is therefore consistent with the equations of motion derived from varying \mathfrak{p} in the action

$$\int \mathfrak{p}(\dot{\mathfrak{q}}^2 - \dot{\mathfrak{q}}^1 + \dot{X}_{\mathfrak{a}}^2 - \dot{X}_{\mathfrak{a}+m}^2) \,.$$

After taking into account the S-R-twist, q^1 and q^2 are not independent anymore, and the above expression then becomes the contribution of p to the action (9.3).

9.6 Bosonic zero modes

In §9.4 we explained how to determine the bosonic part of the action in terms of the variables $\mathfrak{p}^{\mathfrak{a}}, \mathfrak{u}^{\mathfrak{i}\mathfrak{a}'}$, and $\mathfrak{q}^{\mathfrak{i}\mathfrak{a}'}$. Generally, this action can be further simplified because equations (9.13)-(9.15) reduce the total number of independent $\mathfrak{q}^{\mathfrak{i}\mathfrak{a}'}$ variables. So in general, as we shall see in concrete examples in Chapter 10 and Chapter 11, some of the $\mathfrak{u}^{\mathfrak{i}\mathfrak{a}'}$ variables can be expressed as linear combinations of the $\mathfrak{q}^{\mathfrak{i}\mathfrak{a}'}$'s, while the remaining $\mathfrak{u}^{\mathfrak{i}\mathfrak{a}'}$'s are reduced to discrete values. After eliminating the $\mathfrak{u}^{\mathfrak{i}\mathfrak{a}'}$'s we get an action of the form (9.4). If the corresponding constants $\mathcal{M}_{\mathfrak{a}\alpha}$ that appear in this action form a nonsingular square matrix, quantization of (9.4) gives rise to a finite dimensional Hilbert space.

However, for some sectors we end up with more than m independent $q^{ia'}$'s and then the procedure of §9.4 yields an expression similar to (9.4) but where $\mathcal{M}_{a\alpha}$ is not a square matrix. The simplest sector for which this happens is for n = m = 2 with

$$[\mathfrak{B}|\sigma] = \begin{bmatrix} 1 & 1 & | \\ 1 & 1 & | \\ 2 \end{bmatrix} .$$
(9.20)

In general for n = 2 there are more than m independent $\mathfrak{q}^{i\mathfrak{a}'}$'s when \mathfrak{B} has at least two columns of the form $(11)^{\top}$ [where $(\cdots)^{\top}$ denotes the transposed matrix]. As will be explained in more detail in one of the examples of §11.1, for k = 2 the action for the sector (9.20) is

$$I_0 = \frac{1}{2\pi} \int \left\{ \mathfrak{p}^1 \left[d \left(\mathfrak{q}^{11} + \mathfrak{q}^{21} \right) + 2d\mathfrak{q}^{12} \right] + \mathfrak{p}^2 \left[-d \left(\mathfrak{q}^{11} + \mathfrak{q}^{21} \right) + 2d\mathfrak{q}^{12} \right] \right\}.$$
(9.21)

In this sector there are two strings connecting the two D2-branes, and the action only depends on \mathfrak{q}^{11} and \mathfrak{q}^{21} through the center of mass $Q_{com} \equiv \mathfrak{q}^{11} + \mathfrak{q}^{21}$, and is independent of the relative coordinate $Q_{rel} \equiv \mathfrak{q}^{11} - \mathfrak{q}^{21}$. Therefore, in order to proceed we need to add a kinetic term, proportional to \dot{Q}_{rel}^2 . However, if we are only interested in ground states we may simply rewrite (9.21) in terms of Q_{com} ,

$$I_0 = \frac{1}{2\pi} \int \left\{ \mathfrak{p}^1 (dQ_{com} + 2d\mathfrak{q}^{12}) + \mathfrak{p}^2 (-dQ_{com} + 2d\mathfrak{q}^{12}) \right\} , \qquad (9.22)$$

which is of the form (9.4) with a nonsingular square matrix $\mathcal{M} = \begin{pmatrix} 1 & 2 \\ -1 & 2 \end{pmatrix}$. From (9.22) we can determine the number of ground states, which happens to be 4.

We will show later on that sectors with bosonic zero modes invariably also possess fermionic zero modes and therefore do not contribute to the Witten Index. In fact, the sectors with bosonic zero modes form a proper subset of the set of congested sectors, and all congested sectors have fermionic zero modes. In this paper we are only concerned with the Witten Index, and we therefore do not need to consider sectors with bosonic zero modes anymore.

9.7 Fermionic zero modes

So far in this section, we have mainly focused on the bosonic degrees of freedom of the system. While we will not present the explicit form of the fermionic part of the action, it will turn out to be important to understand the fermionic zero modes of the system in each sector described by the binding matrix $\mathfrak{B}_{i\mathfrak{a}}$ and permutation $\sigma \in S_n$. Specifically, they will be crucial to our argument that only decongested sectors contribute to the Witten Index.² Therefore, in this subsection, we will discuss various chirality and boundary conditions that these fermionic zero modes have to satisfy.

Our conventions for fermions are as follows. In describing the fermionic modes of type-IIA theory we find it more convenient to consider its M-theory lift. Therefore, we denote a fermion by a real 32-component 10+1D spinor on which the Dirac matrices $\Gamma^0, \ldots, \Gamma^9, \Gamma^{\natural}$ act. (We use the notation $\natural \equiv 10$ to avoid confusion between Γ^{10} and $\Gamma^1\Gamma^0$.) They satisfy the identity

$$\Gamma^{0123456789\natural} = 1. \tag{9.23}$$

The low-energy fermionic modes of the system can then be described in terms of 1 + 1D fermionic fields that are supported on the type-IIA open strings and on the dimensional reduction (on x_{10} direction) of the D2-branes. (See Figure 9.6 for illustration.) The fermionic field along the i^{th} open F1-string between the $\mathfrak{a'}^{th}$ and $(\mathfrak{a'} + 1)^{st}$ D2-branes is denoted by $\psi_{i\mathfrak{a'}}$ (with $i = 1, \ldots, n$ and $\mathfrak{a'} = 1, \ldots, m-1$). For every $i = 1, \ldots, n$, there is another piece of string starting on the m^{th} D2-brane, going through the S-R-twist, and ending on the 1^{st} D2-brane. To capture the fields on this string using the same notation, we extend the range of $\mathfrak{a'}$ to $0, \ldots, m$ and postulate that the fields ψ_{i0} and ψ_{im} are identified up to the S-R-twist

$$\psi_{\sigma(\mathbf{i})0} = e^{\frac{\upsilon}{2}(\Gamma^{1\natural} + \Gamma^{45} + \Gamma^{67} + \Gamma^{89})} \psi_{\mathbf{i}m} , \qquad (9.24)$$

where we have also allowed the possibility of the action of the permutation $\sigma \in S_n$, as discussed in §9.4. We set

$$\mathcal{P} \equiv e^{-\frac{\upsilon}{2}(\Gamma^{1\natural} + \Gamma^{45} + \Gamma^{67} + \Gamma^{89})}.$$
(9.25)

²This will be our only use for the fermionic degrees of freedom, so a reader who doesn't wish to go into the detailed proof of this statement can skip the present section and $\S10.3, \$11.2$, as well as some portions of $\S9.9$.



Figure 9.6: The fermionic zero modes are constructed from solutions of the linear equations for the boundary conditions of the gluino fields $\lambda_{\mathfrak{a}\mathfrak{i}'}$ on the D2-brane sections (viewed as 1-dimensional segments below the x_{10} compactification scale) and the fermionic modes $\psi_{\mathfrak{i}\mathfrak{a}}$ on string sections. The S-R-twist is denoted by a \times .

Generally, the $\psi_{i\mathfrak{a}'}$ fields are functions of (x_3, t) (with the appropriate finite range for x_3), but at low-energy only the zero modes are important, so we can assume that $\psi_{i\mathfrak{a}'}$ is independent of x_3 . The $\psi_{i\mathfrak{a}'}$ fields also satisfy the obvious boundary conditions that if the i^{th} string is not bound to the \mathfrak{a}^{th} D2-brane then it continuously connects with $\psi_{i(\mathfrak{a}-1)}$:

$$\psi_{\mathfrak{ia}} = \psi_{\mathfrak{i(a-1)}} \qquad \text{if } \mathfrak{B}_{\mathfrak{ia}} = 0$$

All the ψ_{ia} fields satisfy the chirality condition

$$0 = (1 + \Gamma^{023})\psi_{i\mathfrak{a}} \tag{9.26}$$

corresponding to the low-energy fields on an M2-brane with x_2 being the M-theory direction.

Next, we define the fields along the D2-branes. Dimensionally reducing along the compact direction of x_{10} , the D2-branes become 1 + 1D objects, and we denote by

 $\lambda_{\mathfrak{a}\mathfrak{i}'}$ (with $\mathfrak{a} = 1, \ldots, m$ and $\mathfrak{i}' = 0, \ldots, n$) the low-energy fermionic fields supported on the segment of \mathfrak{a}^{th} D2-brane between the $\mathfrak{i'}^{th}$ and $(\mathfrak{i'}+1)^{st}$ F1-string, where we use the convention that $\mathfrak{i'} = 0$ corresponds to the segment that continues from the first string to $x_9 > 0$, and $\mathfrak{i'} = n$ continues from the n^{th} string to $x_9 < 0$ (see Figure 9.6). These fields are generally functions of (x_9, t) , but at low-energy they can again be assumed to be constant. Formally, the range of x_9 for $1 \leq \mathfrak{i'} < n$ is zero, but because the fields are constant this does not matter. A more elaborate treatment starting with 2D fields that are harmonic functions of $x_9 + ix_{10}$, with poles at the intersections with the F1-string, will lead to a similar result.

Similarly as for $\psi_{ia'}$, we have for $\lambda_{ai'}$ the continuity conditions

$$\lambda_{\mathfrak{a}\mathfrak{i}} = \lambda_{\mathfrak{a}(\mathfrak{i}-1)}$$
 if $\mathfrak{B}_{\mathfrak{i}\mathfrak{a}} = 0$

and we also have the chirality conditions:

$$0 = (1 + \Gamma^{09\natural})\lambda_{\mathfrak{a}\mathfrak{i}'}.$$
(9.27)

At the D2-F1 junctions where the \mathfrak{a}^{th} D2-brane and \mathfrak{i}^{th} F1-string intersect (so that $\mathfrak{B}_{\mathfrak{i}\mathfrak{a}} = 1$), the zero modes have to satisfy the following boundary conditions:

$$(1 - \Gamma^{239\natural})\psi_{i\mathfrak{a}} = (1 - \Gamma^{239\natural})\psi_{i(\mathfrak{a}-1)} = (1 - \Gamma^{239\natural})\lambda_{\mathfrak{a}i} = (1 - \Gamma^{239\natural})\lambda_{\mathfrak{a}(i-1)}, \qquad (9.28)$$

and

$$0 = (1 + \Gamma^{239\natural})(\psi_{\mathfrak{i}(\mathfrak{a}-1)} - \psi_{\mathfrak{i}\mathfrak{a}} + \Gamma^{39}\lambda_{\mathfrak{a}\mathfrak{i}} - \Gamma^{39}\lambda_{\mathfrak{a}(\mathfrak{i}-1)}).$$

$$(9.29)$$

These equations are derived by first going to the M-theory picture as in Figure 9.5, where the D2-F1 junction is described by a single M2-brane, and then deforming the M2-brane worldvolume. Details of the derivation of (9.28)–(9.29) are provided in Appendix 13.1.

9.8 The eigenvalues of \mathcal{P}

The operator \mathcal{P} , defined in (9.25), realizes the S-R-twist on the fermionic modes of the fundamental strings in the type-IIA picture. In this subsection we calculate its eigenvalues. This will be important in §10.3 and §11.2 where we prove the absence of zero modes for certain sectors.

Consider a spinor ψ that satisfies $\mathcal{P}\psi = \varepsilon\psi$ for some eigenvalue ε . Since \mathcal{P} commutes with Γ^{023} we may assume that ψ has a specific Γ^{023} chirality. We first assume that

$$\Gamma^{023}\psi = \psi.$$

Then, $\Gamma^{0123456789\natural} = 1$ implies

$$0 = (1 - \Gamma^{023})\psi = (1 - \Gamma^{1456789\natural})\psi,$$

and hence

$$\Gamma^{451\natural}\psi = \Gamma^{451\natural}\Gamma^{1456789\natural}\psi = \Gamma^{6789}\psi \,.$$

We can therefore rewrite $\mathcal{P}\psi$ as

$$\mathcal{P}\psi = e^{\frac{v}{2}\Gamma^{45}(1-\Gamma^{451\natural}) + \frac{v}{2}\Gamma^{67}(1-\Gamma^{6789})}\psi = e^{\frac{v}{2}\Gamma^{45}(1-\Gamma^{6789}) + \frac{v}{2}\Gamma^{67}(1-\Gamma^{6789})}\psi.$$

Since Γ^{6789} has eigenvalues ± 1 , and Γ^{ij} has eigenvalues $\pm i$ for all spatial indices i, j $(i \neq j)$, we deduce that \mathcal{P} has eigenvalues 1 and $e^{\pm 2iv}$ on the subspace with Γ^{023} -chirality 1.

We can similarly analyze the eigenvalues of \mathcal{P} on the subspace of ψ 's with the opposite Γ^{023} -chirality, i.e., $\Gamma^{023}\psi = -\psi$. On that subspace we find

$$\mathcal{P}\psi = e^{\frac{\nu}{2}\Gamma^{45}(1-\Gamma^{451\natural}) + \frac{\nu}{2}\Gamma^{67}(1-\Gamma^{6789})}\psi = e^{\frac{\nu}{2}\Gamma^{45}(1+\Gamma^{6789}) + \frac{\nu}{2}\Gamma^{67}(1-\Gamma^{6789})}\psi, \qquad (9.30)$$

and hence deduce that \mathcal{P} has eigenvalues $e^{\pm iv}$. Note that $e^{\pm iv} \neq 1$ and so there is no nontrivial solution to $\psi = \mathcal{P}\psi$ on the subspace with Γ^{023} -chirality -1. This fact will come in handy later on.

9.9 Constructing a Witten Index

The states of the system discussed in §9.1, while they contain all the information about the Hilbert space of Tr-S theory with external charges, also contain superfluous excitations in the form of long wavelength modes of $X_{\mathfrak{c}}^{\mu}$, $\psi_{\mathfrak{c}}$ along the semi-infinite strings. We can eliminate these excitations by imposing appropriate boundary conditions on the modes $X_{\mathfrak{c}}^{\mu}$, $\psi_{\mathfrak{c}}$ at some finite distance from the origin, say at $x_9 = \pm \Delta$ (where the + sign is for $\mathfrak{c} \leq m$ and the - sign is for $\mathfrak{c} > m$, for some positive constant Δ). The following boundary conditions preserve the four real supersymmetries left unbroken by (9.6)-(9.8).

In the type-IIB picture, we pick Neumann boundary conditions for fluctuations in directions $4, \ldots, 8$:

$$\partial_9 X^4_{\mathfrak{c}}(\pm \Delta) = \partial_9 X^5_{\mathfrak{c}}(\pm \Delta) = \partial_9 X^6_{\mathfrak{c}}(\pm \Delta) = \partial_9 X^7_{\mathfrak{c}}(\pm \Delta) = \partial_9 X^8_{\mathfrak{c}}(\pm \Delta) = 0, \quad (9.31)$$

Dirichlet boundary conditions in directions 1, 2, 3:

$$X^{1}_{\mathfrak{c}}(\pm\Delta) = a^{(\mathfrak{c})}_{1}, \quad X^{2}_{\mathfrak{c}}(\pm\Delta) = a^{(\mathfrak{c})}_{2}, \quad X^{3}_{\mathfrak{c}}(\pm\Delta) = 0,$$
 (9.32)

and supersymmetric boundary conditions for the fermions:

$$\Gamma^{045678}\psi_{\mathfrak{c}}(\pm\Delta) = \psi_{\mathfrak{c}}(\pm\Delta).$$
(9.33)

These boundary conditions are formally what we would get if we let the string end on a D5-brane that extends in directions $4, \ldots, 8$ and is fixed in directions 1, 2, 3, 9. However, we must note that because directions 1, 2, 3 are compact, such a D5-brane will back-react strongly on the metric and will require additional orientifolds or other objects to make a complete string-theory solution. (This is similar to the situation with D8-branes as developed in [54], or with D7-branes in [55].) Here, we will simply regard (9.31)—(9.33) as formal boundary conditions that we impose to get rid of unwanted zero modes.

To write down the boundary conditions equivalent to (9.33) in the type IIA picture, we first decompose the fermionic zero mode $\psi_{\mathfrak{c}}$ on the type-IIB fundamental string into left-moving and right-moving components as $\psi_{\mathfrak{c}} = \psi_{\mathfrak{c}+} + \psi_{\mathfrak{c}-}$, where

$$\psi_{\mathfrak{c}\pm} = \pm \Gamma^{09} \psi_{\mathfrak{c}\pm} = \pm \Gamma^{12345678} \psi_{\mathfrak{c}\pm} \,. \tag{9.34}$$

The boundary condition (9.33) can now be written as a relation between the leftmoving and right-moving modes at the end of the string $(x_9 = \pm \Delta)$:

$$\psi_{\mathfrak{c}+} = \Gamma^{045678} \psi_{\mathfrak{c}-} \,. \tag{9.35}$$

We now follow the dualities of Table 8.1, each time transforming the fermionic field to a dual field on a dual brane. The type-IIA boundary conditions then become:

$$0 = (1 - \Gamma^{\natural})(1 + \Gamma^{0145678})\lambda_{\mathfrak{a}\mathfrak{i}'}$$
(9.36)

for $\mathbf{i}' = 0, n$. We can now define the Witten index $I = \text{Tr}(-1)^F$ of the quantum mechanical system of D3-branes and open strings in the type IIB picture, and we can calculate it using the system of D2-branes and F1-strings in the dual type-IIA picture. With boundary conditions (9.31)—(9.33), the semi-infinite open strings can be regarded as external charges with no internal dynamics, so the Witten index I will simply count the number of ground states of the theory with external charges inserted.

On the other hand, in the type IIA picture, we have various configurations of D2branes and F1-strings, which can be classified into sectors described by the binding matrix \mathfrak{B} and permutation σ . With the extra conditions (9.36) we will show (see §10.3 and §11.2) that only decongested sectors do not have fermionic zero modes. Such sectors will make a nonzero contribution to the Witten Index.

We also remark that it is possible to create open strings in a supersymmetric configuration by using D3-branes instead of D5-branes. This will avoid the problem of incompleteness of the background, but will instead create additional fermionic zero modes to make the Witten Index identically zero. We briefly discuss this construction in Appendix 13.3.

9.10 Summary of the rules

A sector of Tr-S theory with U(n) gauge group in the presence of m external quark and anti-quark pairs is described by a permutation $\sigma \in S_n$ and an $n \times m$ binding
matrix $\mathfrak{B}_{i\mathfrak{a}}$ of 0's and 1's. We build the action in terms of periodic variables $\mathfrak{u}^{i\mathfrak{a}'}, \mathfrak{q}^{i\mathfrak{a}'}, \mathfrak{u}^{\mathfrak{a}}$, and $\mathfrak{p}^{\mathfrak{a}}$, with $\mathfrak{i} = 1, \ldots, n$, $\mathfrak{a} = 1, \ldots, m$, and $\mathfrak{a}' = 0, \ldots, m$. All variables have 2π periodicity. The variables are further restricted by linear relations with integer coefficients:

$$\mathfrak{u}^{\mathfrak{i}\mathfrak{a}} = \mathfrak{u}^{\mathfrak{i}(\mathfrak{a}-1)}, \quad \mathfrak{q}^{\mathfrak{i}\mathfrak{a}} = \mathfrak{q}^{\mathfrak{i}(\mathfrak{a}-1)}, \qquad \text{whenever } \mathfrak{B}_{\mathfrak{i}\mathfrak{a}} = 0,$$
(9.37)

$$\mathfrak{u}^{\mathfrak{a}} = \mathfrak{u}^{\mathfrak{i}\mathfrak{a}} = \mathfrak{u}^{\mathfrak{i}(\mathfrak{a}-1)}, \quad \text{whenever } \mathfrak{B}_{\mathfrak{i}\mathfrak{a}} = 1,$$
 (9.38)

and

$$\mathfrak{q}^{\sigma(\mathfrak{i})0} = \mathbf{d}\mathfrak{q}^{\mathfrak{i}m} + \mathbf{b}\mathfrak{u}^{\mathfrak{i}m}, \qquad \mathfrak{u}^{\sigma(\mathfrak{i})0} = \mathbf{c}\mathfrak{q}^{\mathfrak{i}m} + \mathbf{a}\mathfrak{u}^{\mathfrak{i}m}.$$
(9.39)

The action is

$$I_0 = \frac{1}{2\pi} \int dt \sum_{i=1}^n \sum_{a=1}^m \mathfrak{B}_{ia} \mathfrak{p}^{\mathfrak{a}} (\dot{\mathfrak{q}}^{ia} - \dot{\mathfrak{q}}^{i(a-1)}) \,. \tag{9.40}$$

Furthermore, the coordinates $(X^1_{\mathfrak{a}}, X^2_{\mathfrak{a}})$ of the *m* quarks, as well as the, the coordinates $(X^1_{\mathfrak{a}+m}, X^2_{\mathfrak{a}+m})$ of the *m* anti-quarks are encoded in the action via an extra term

$$I_{1} = \frac{1}{2\pi} \int dt \sum_{\mathfrak{a}=1}^{m} [(\dot{X}_{\mathfrak{a}}^{1} - \dot{X}_{\mathfrak{a}+m}^{1})\mathfrak{u}^{\mathfrak{a}} + (\dot{X}_{\mathfrak{a}}^{2} - \dot{X}_{\mathfrak{a}+m}^{2})\mathfrak{p}^{\mathfrak{a}}].$$
(9.41)

Finally, we only keep those sectors for which $\sum_{i} \mathfrak{B}_{i\mathfrak{a}} = 1$ for all $\mathfrak{a} = 1, \ldots, m$ (we called these *decongested* sectors), because sectors for which $\sum_{i} \mathfrak{B}_{i\mathfrak{a}} > 1$ for some \mathfrak{a} have zero modes and therefore do not contribute to the Witten Index, while sectors for which $\sum_{i} \mathfrak{B}_{i\mathfrak{a}} = 0$ for some \mathfrak{a} have a D2-brane disconnected from the rest of the system.

Chapter 10

U(1) gauge group

We will now apply the rules of §9.10 to study our setup with U(1) gauge group. We already know that abelian Tr-S is equivalent to U(1) Chern-Simons theory at level k [1]. This model will therefore provide us with a good example of how the rules of §9.10 work. In §10.1-§10.3 we apply the rules of §9.10 to U(1) Tr-S with m charge pairs, and in §10.4 we show how they reproduce the predictions from Chern-Simons theory.

In solving the problem, a central role is played by the action (7.4). The general actions that we will consider are equivalent to quantum mechanical systems that are obtained by geometric quantization of T^{2m} . They are of the form:

$$I = \frac{1}{2\pi} \int \mathcal{M}_{\mathfrak{a}\alpha} \mathfrak{p}^{\mathfrak{a}} d\mathfrak{q}^{\alpha}, \qquad \mathfrak{a}, \alpha = 1, \dots, m.$$
(10.1)

Here $\mathfrak{p}^{\mathfrak{a}}, \mathfrak{q}^{\alpha}$ are periodic coordinates parameterizing T^{2m} , in the range $[0, 2\pi)$, and $\mathcal{M}_{\mathfrak{a}\alpha}$ are the components of a nonsingular $m \times m$ matrix of integers. We denote its determinant by

$$\Delta \equiv \det_{m \times m}(\mathcal{M}_{\mathfrak{a}\alpha}) \neq 0.$$

The dimension of the Hilbert space is then $|\Delta|$. We denote the inverse matrix by $(\mathcal{M}^{-1})^{\alpha\mathfrak{a}}$, i.e., $\mathcal{M}_{\mathfrak{a}\alpha}(\mathcal{M}^{-1})^{\alpha\mathfrak{b}} = \delta^{\mathfrak{b}}_{\mathfrak{a}}$. Suppose that (k_1, \ldots, k_m) and (l_1, \ldots, l_m) are vectors of integers. Then the operators $\exp(i\sum_{\mathfrak{a}} k_{\mathfrak{a}}\mathfrak{p}^{\mathfrak{a}})$ and $\exp(i\sum_{\alpha} l_{\alpha}\mathfrak{q}^{\alpha})$ are well-defined, and we have the commutation relation

$$e^{i\sum_{\mathfrak{a}}k_{\mathfrak{a}}\mathfrak{p}^{\mathfrak{a}}}e^{i\sum_{\alpha}l_{\alpha}\mathfrak{q}^{\alpha}}e^{-i\sum_{\mathfrak{a}}k_{\mathfrak{a}}\mathfrak{p}^{\mathfrak{a}}}e^{-i\sum_{\alpha}l_{\alpha}\mathfrak{q}^{\alpha}} = e^{2\pi i\sum_{\alpha,\mathfrak{a}}(\mathcal{M}^{-1})^{\alpha\mathfrak{a}}k_{\mathfrak{a}}l_{\alpha}}.$$
 (10.2)

10.1 A quark and anti-quark pair (m = 1)

Using the rules summarized in §9.10, we can now write down explicitly the lowenergy effective action for the type-IIA configuration that corresponds to inserting a quark and anti-quark pair in Tr-S theory. As we have explained in the previous section, in such a case we have one fundamental string wrapping the x_3 circle and attached to one D2-brane, forming a bound state with it. The closed string that we would have in the absence of external charges breaks to become an open string starting and ending on the D2-brane. In the notations explained in §9.10, for fixed charges we have the action

$$I_0 = \frac{1}{2\pi} \int \mathfrak{p}^1 \left(d\mathfrak{q}^{11} - d\mathfrak{q}^{10} \right), \qquad (10.3)$$

and if we wish to allow the charges to move around we need to add the term

$$I_1 = -\frac{1}{2\pi} \int \left[\left(X_1^1 - X_2^1 \right) d\mathfrak{u}^1 + \left(X_1^2 - X_2^2 \right) d\mathfrak{p}^1 \right].$$
(10.4)

To proceed, we recall that the S-R-twisted boundary conditions induce linear relations among the variables, following (9.39). For k = 2 this yields $\mathfrak{u}^1 = \mathfrak{q}^{11} = -\mathfrak{q}^{10} \equiv \mathfrak{q}^1$, and thus (10.3)-(10.4) can be simplified to:

$$I \equiv I_0 + I_1 \tag{10.5}$$

$$I_0 = \frac{1}{2\pi} \int 2\mathfrak{p}^1 d\mathfrak{q}^1, \qquad (10.6)$$

$$I_1 = -\frac{1}{2\pi} \int \left\{ \left[a_1^{(1)}(t) - a_1^{(2)}(t) \right] d\mathfrak{q}^1 + \left[a_2^{(1)}(t) - a_2^{(2)}(t) \right] d\mathfrak{p}^1 \right\}, \qquad (10.7)$$

where, as defined in §7, $(a_1^{(j)}, a_2^{(j)}) = (X_j^1, X_j^2)$ refer to the T^2 coordinates of the quark's and anti-quark's world-lines.

After taking into account similarly the constraints in (9.14) and (9.15) for the cases k = 1, 3, the action for static charges (10.3) can be written collectively for all three values of k as¹

$$I_0 = \frac{k}{2\pi} \int \mathfrak{p} d\mathfrak{q} \,. \tag{10.8}$$

This action is of the form (10.1) and describes geometric quantization of T^2 . It gives rise to a k-dimensional Hilbert space.

We also note that the action (10.5) is of the same form as the action for the ground states of the well-known Landau problem describing the low-energy spectrum of a two-dimensional charged particle moving on a torus with k units of magnetic flux. In this context, the velocity of the quark relative to the anti-quark, which is given by (1) = (2) = (2)

$$(\dot{a}_1^{(1)}(t) - \dot{a}_1^{(2)}(t), \, \dot{a}_2^{(1)}(t) - \dot{a}_2^{(2)}(t))$$

is interpreted as the electric field in the Landau problem. This is consistent with the interpretation of the quark and anti-quark as charges in U(1) Chern-Simons theory (see §10.4 below). To simplify matters, in the following, we consider static charges and let the quarks and anti-quarks be located at the origin.

¹For
$$k = 1$$
, $q^{11} = 0$ and $u^1 = -q^{10}$, whereas for $k = 3$, $-q^{10} = 2q^{11} = 2u^1$.

10.2 Multiple quark and anti-quark pairs (m > 1)

Generalization of the results of §10.1 to m > 1 cases is straightforward. There is only one sector: the permutation $\sigma \in S_1$ is the identity, and the $1 \times m$ binding matrix \mathfrak{B} is

$$\mathfrak{B} = \begin{pmatrix} 1 & 1 & \cdots & 1 \end{pmatrix}$$

The relations (9.38) imply that all $\mathfrak{u}^{i\mathfrak{a}}$ variables are equal:

$$\mathfrak{u}^{10} = \mathfrak{u}^{11} = \mathfrak{u}^{12} = \cdots = \mathfrak{u}^{1m} = \mathfrak{u}^1 = \cdots = \mathfrak{u}^m \equiv \mathfrak{u}.$$

Just like what we have done in $\S10.1$, we can write down the low-energy effective action as

$$I = \frac{1}{2\pi} \int \sum_{\mathfrak{a}=1}^{m} \mathfrak{p}^{\mathfrak{a}} \left(d\mathfrak{q}^{\mathfrak{l}\mathfrak{a}} - \mathfrak{q}^{\mathfrak{l}(\mathfrak{a}-1)} \right) - \frac{1}{2\pi} \int \sum_{\mathfrak{a}=1}^{m} \left[\left(a_{1}^{(\mathfrak{a})}(t) - a_{1}^{(\mathfrak{a}+m)}(t) \right) d\mathfrak{u} - \left(a_{2}^{(\mathfrak{a})}(t) - a_{2}^{(\mathfrak{a}+m)}(t) \right) d\mathfrak{p}^{\mathfrak{a}} \right].$$

$$(10.9)$$

Furthermore, (9.39) gives linear constraints among $\mathfrak{u}, \mathfrak{q}^{10}, \mathfrak{q}^{1m}$ which depend on k:

$$\mathfrak{q}^{10} = \mathbf{d}\mathfrak{q}^{1m} + \mathbf{b}\mathfrak{u}, \qquad \mathfrak{u} = \mathbf{c}\mathfrak{q}^{1m} + \mathbf{a}\mathfrak{u}.$$
 (10.10)

This can be solved using the explicit expressions for $\mathbf{a}, \mathbf{b}, \mathbf{c}, \mathbf{d}$ in terms of k, given in (7.3), and we get

$$q^{10} = -\mathfrak{u}, \quad q^{1m} = 0, \qquad \text{for } k = 1,$$
 (10.11)

and

$$q^{10} = (1-k)u, \quad q^{1m} = u, \quad \text{for } k = 2, 3.$$
 (10.12)

After taking into account these linear constraints, the action for fixed external charges becomes

$$I_0 = \frac{1}{2\pi} \int \left\{ \mathfrak{p}^1(d\mathfrak{q}^1 + d\mathfrak{q}^m) + \sum_{\mathfrak{a}=2}^{m-1} \mathfrak{p}^\mathfrak{a}(d\mathfrak{q}^\mathfrak{a} - d\mathfrak{q}^{\mathfrak{a}-1}) - \mathfrak{p}^m d\mathfrak{q}^{m-1} \right\}, \qquad \text{for } k = 1, \ (10.13)$$

where we set $q^{\mathfrak{a}} = q^{1\mathfrak{a}}$ for $\mathfrak{a} = 1, \dots, m-1$ and $q^m = \mathfrak{u}$, and

$$I_{0} = \frac{1}{2\pi} \int \left\{ \mathfrak{p}^{1}[d\mathfrak{q}^{1} + (k-1)d\mathfrak{q}^{m}] + \sum_{\mathfrak{a}=2}^{m} \mathfrak{p}^{\mathfrak{a}}(d\mathfrak{q}^{\mathfrak{a}} - d\mathfrak{q}^{\mathfrak{a}-1}) \right\}, \quad \text{for } k = 2, 3, \quad (10.14)$$

where we set $q^{\mathfrak{a}} \equiv q^{1\mathfrak{a}}$. The action (10.14) is of the form (10.1) with

$$\mathcal{M} = \begin{pmatrix} 1 & 0 & 0 & 0 & \cdots & 0 & k-1 \\ -1 & 1 & 0 & 0 & \cdots & 0 & 0 \\ 0 & -1 & 1 & 0 & \cdots & 0 & 0 \\ \vdots & \ddots & \ddots & \vdots & \vdots & & \\ 0 & 0 & 0 & 0 & \cdots & -1 & 1 \end{pmatrix},$$
(10.15)

and it is also straightforward to write down the corresponding matrix \mathcal{M} for k = 1 case. Following the discussion below (10.1), we can immediately compute the dimension of each Hilbert space as $|\Delta| = k$.

10.3 Absence of fermionic zero modes

In this subsection, we show that there is no fermionic zero mode in the sector discussed in §10.2. This is necessary for the consistency of our result, because otherwise the contribution of the sector to the Witten index would be zero, and since this is the only sector for n = 1 cases, this would mean that the Witten index of U(1) Tr-S theory would be zero too, regardless of the value of k. On the other hand, we know that U(1) Tr-S theory is simply U(1) Chern-Simons theory, which contains k bosonic ground states only.

As explained in §9.7, the low-energy fermionic modes of our system can be understood via 1 + 1D fermionic fields $\psi_{i\mathfrak{a}'}$ and $\lambda_{\mathfrak{a}i'}$ supported on the open strings and D2-branes. Each $\psi_{i\mathfrak{a}'}$ and $\lambda_{\mathfrak{a}i'}$ satisfies the chirality conditions (9.26) and (9.27), respectively. In addition, the $\lambda_{\mathfrak{a}i'}$'s at the two far ends of the D2-branes (that is, those with $\mathfrak{i}' = 0, n$) satisfy the constraint (9.36) which derives from dualizing the boundary conditions for open strings ending on D5-branes in the type-IIB setting. Each intersection of D2-F1 satisfies the two junction conditions (9.28) and (9.29). Finally, the S-R-twist on the F1-string gives rise to the boundary condition (9.24). We now proceed to prove that in the n = 1 abelian case, these various boundary conditions dictate that we have no fermionic zero modes. (See Figure 10.1 for notation.)

As a warm-up, we start with the m = 0 case. In this case there is only one variable ψ_{10} which satisfies the boundary conditions (9.24) and the chirality condition (9.26):

$$\psi_{10} = \mathcal{P}\psi_{10}, \qquad (\Gamma^{023} + 1)\psi_{10} = 0.$$
 (10.16)

But in §9.8 we saw that all the eigenvectors of \mathcal{P} with eigenvalue 1 have the opposite Γ^{023} chirality from that of ψ_{10} . It follows that(10.16) does not have any non-trivial solutions and the m = 0 states have no zero modes.

Now, let us study the m > 0 case. We first consider the continuity of $\lambda_{\mathfrak{a}\mathfrak{i}'}$ at each D2-F1 junction. Define

$$\zeta_{\mathfrak{a}} \equiv \lambda_{\mathfrak{a}1} - \lambda_{\mathfrak{a}0}$$



Figure 10.1: Illustration of fermionic zero modes for n = 1 and m = 2 case. The fundamental string breaks at every D2-brane intersection.

for $\mathfrak{a} = 1, \ldots, m$. Since in the abelian case (n = 1) both $\lambda_{\mathfrak{a}0}$ and $\lambda_{\mathfrak{a}1}$ satisfy the chirality condition (9.27) and the boundary condition (9.36), $\zeta_{\mathfrak{a}}$ must also satisfy the equations

$$0 = (1 + \Gamma^{09\natural})\zeta_{\mathfrak{a}}, \qquad (10.17)$$

$$0 = (1 - \Gamma^{\natural})(1 + \Gamma^{0145678})\zeta_{\mathfrak{a}}, \qquad (10.18)$$

$$0 = (1 - \Gamma^{239\natural})\zeta_{\mathfrak{a}}, \qquad (10.19)$$

where the last equation comes from (9.28). Equations (10.17) and (10.19) together imply

$$\zeta_{\mathfrak{a}} = \Gamma^{023} \zeta_{\mathfrak{a}} \,. \tag{10.20}$$

Since in our convention $\Gamma^{0123456789\natural} = 1$ (see Appendix 13.1), this in turn implies

$$\Gamma^{0145678}\zeta_{\mathfrak{a}} = \zeta_{\mathfrak{a}}\,,\tag{10.21}$$

and (10.18) becomes

$$(1-\Gamma^{\natural})\zeta_{\mathfrak{a}}=0.$$

Then (10.19) now reads

$$\Gamma^{239}\zeta_{\mathfrak{a}} = \zeta_{\mathfrak{a}} \,, \tag{10.22}$$

which has no non-trivial solution, because Γ^{239} has no real eigenvalues (it squares to -1). Therefore, we obtain $\zeta_{\mathfrak{a}} = 0$, or $\lambda_{\mathfrak{a}0} = \lambda_{\mathfrak{a}1}$ for all $\mathfrak{a} = 1, \ldots, m$.

Next, we consider the $\psi_{1\mathfrak{a}'}$ fields. Since $\zeta_{\mathfrak{a}} = 0$, equations (9.28)—(9.29) now become

$$(1 \pm \Gamma^{239\natural})(\psi_{1,\mathfrak{a}-1} - \psi_{1\mathfrak{a}}) = 0,$$

or simply

$$\psi_{10} = \psi_{11} = \dots = \psi_{1m} \,. \tag{10.23}$$

On the other hand, we have from the S-R-twist condition $\psi_{1m} = \mathcal{P}\psi_{10}$, where the operator \mathcal{P} is defined in (9.25), and together with the above equalities,

$$\psi_{10} = \mathcal{P}\psi_{10} \,. \tag{10.24}$$

At this point, we can again use the fact that all the eigenvectors of \mathcal{P} with eigenvalue 1 have the opposite Γ^{023} chirality from that of ψ_{10} . Therefore, (10.24) does not have a non-trivial solution, and hence

$$\psi_{10} = \psi_{11} = \dots = \psi_{1m} = 0.$$
(10.25)

Finally, let us define

$$\xi_{\mathfrak{a}} = \lambda_{\mathfrak{a}0} + \lambda_{\mathfrak{a}1}$$

With $\zeta_{\mathfrak{a}} = \psi_{1\mathfrak{a}'} = 0$, we find that $\xi_{\mathfrak{a}}$ satisfies the same set of equations (10.17)-(10.19) as $\zeta_{\mathfrak{a}}$:

$$0 = (1 + \Gamma^{09\natural})\xi_{\mathfrak{a}}, \qquad (10.26)$$

$$0 = (1 - \Gamma^{\natural})(1 + \Gamma^{0145678})\xi_{\mathfrak{a}}, \qquad (10.27)$$

$$0 = (1 - \Gamma^{239\natural})\xi_{\mathfrak{a}}. \tag{10.28}$$

The first two equations follow from the chirality and boundary conditions for $\lambda_{\mathfrak{a}i'}$, while the third follows from (9.28) and the previous result $\psi_{1\mathfrak{a}'} = 0$ for all \mathfrak{a}' . Therefore, we find $\xi_{\mathfrak{a}} = 0$ as before. Together with $\zeta_{\mathfrak{a}} = 0$, this implies $\lambda_{\mathfrak{a}i'} = 0$ for all $\mathfrak{a} = 1, \ldots, m$ and $\mathfrak{i}' = 0, 1$. To summarize, we conclude that there is no fermionic zero mode for the abelian n = 1 case with an arbitrary number m of D2-branes.

10.4 Comparison with Chern-Simons theory results

At this point, it is pertinent to discuss consistency with abelian Chern-Simons theory, which we had explained in [1] to be the low-energy limit of abelian Tr-S theory. For U(1) Chern-Simons theory the dimension of the Hilbert space with m external charge pairs is always equal to the level k, independently of m. This is indeed what we found from the type-IIA dual picture in §10.2. There, the dimension dim $\mathcal{H}(k, n =$ 1, m) can be calculated as the determinant of the matrix \mathcal{M} that appears in (10.15) and its k = 1 counterpart, and we indeed find the result dim $\mathcal{H}(k, n = 1, m) = k$ independently of m. The underlying reason for this coincidence is the fact that there are no fermionic zero modes, as we have proved in §10.3. If follows that the dimension we calculated is in fact the Witten Index of the system.

To go beyond the mere equality of dimensions of the Hilbert spaces, we can consider for k > 1 the action of the \mathbb{Z}_k symmetry operators \mathcal{U}, \mathcal{V} discussed in §8.3. At the classical level, the discrete translation \mathcal{U} acts as

$$\mathcal{U}:\qquad \mathfrak{u}^{\mathfrak{i}\mathfrak{a}'} \to \mathfrak{u}^{\mathfrak{i}\mathfrak{a}'} + \frac{2\pi}{k}, \quad \mathfrak{q}^{\mathfrak{i}\mathfrak{a}'} \to \mathfrak{q}^{\mathfrak{i}\mathfrak{a}'} + \frac{2\pi}{k}, \quad \mathfrak{u}^{\mathfrak{a}} \to \mathfrak{u}^{\mathfrak{a}} + \frac{2\pi}{k}, \quad \mathfrak{p}^{\mathfrak{a}} \to \mathfrak{p}^{\mathfrak{a}}, \quad (10.29)$$

while \mathcal{V} , which is related to the homology class of the fundamental string, can be interpreted as electric flux on the D2-branes and acts as

$$\mathcal{V}:\qquad \mathfrak{u}^{\mathfrak{i}\mathfrak{a}'} \to \mathfrak{u}^{\mathfrak{i}\mathfrak{a}'}, \quad \mathfrak{q}^{\mathfrak{i}\mathfrak{a}'} \to \mathfrak{q}^{\mathfrak{i}\mathfrak{a}'}, \quad \mathfrak{u}^{\mathfrak{a}} \to \mathfrak{u}^{\mathfrak{a}}, \quad \mathfrak{p}^{\mathfrak{a}} \to \mathfrak{p}^{\mathfrak{a}} + \frac{2\pi}{k}.$$
(10.30)

After geometric quantization, the actions (10.29)-(10.30) translate to actions on quantum operators of the system. The action is by conjugation; for example the rightmost expression of (10.30) is to be read as $\mathcal{V}^{-1}\mathfrak{p}^{\mathfrak{a}}\mathcal{V} = \mathfrak{p}^{\mathfrak{a}} + \frac{2\pi}{k}$.

For n = 1 we find in terms of the variables of (10.14)

$$\mathcal{V}^{-1}\mathfrak{p}^{\mathfrak{a}}\mathcal{V} = \mathfrak{p}^{\mathfrak{a}} + \frac{2\pi}{k}, \qquad \mathcal{V}^{-1}\mathfrak{q}^{\mathfrak{a}}\mathcal{V} = \mathfrak{q}^{\mathfrak{a}}, \qquad \mathcal{U}^{-1}\mathfrak{p}^{\mathfrak{a}}\mathcal{U} = \mathfrak{p}^{\mathfrak{a}}, \qquad \mathcal{U}^{-1}\mathfrak{q}^{\mathfrak{a}}\mathcal{U} = \mathfrak{q}^{\mathfrak{a}} + \frac{2\pi}{k},$$
(10.31)

for $\mathfrak{a} = 1, \ldots, m$. Using the commutation relation (10.2), together with the inverse of (10.15):

$$\mathcal{M}^{-1} = \begin{pmatrix} \frac{1}{k} & \frac{1}{k} - 1 & \frac{1}{k} - 1 & \frac{1}{k} - 1 & \cdots & \frac{1}{k} - 1 & \frac{1}{k} - 1 \\ \frac{1}{k} & \frac{1}{k} & \frac{1}{k} - 1 & \frac{1}{k} - 1 & \cdots & \frac{1}{k} - 1 & \frac{1}{k} - 1 \\ \frac{1}{k} & \frac{1}{k} & \frac{1}{k} & \frac{1}{k} - 1 & \cdots & \frac{1}{k} - 1 & \frac{1}{k} - 1 \\ \vdots & \ddots & \ddots & \vdots & \vdots & \vdots \\ \frac{1}{k} & \frac{1}{k} & \frac{1}{k} & \frac{1}{k} & \frac{1}{k} & \cdots & \frac{1}{k} & \frac{1}{k} \end{pmatrix},$$
(10.32)

we find that, up to a possible constant phase, we can identify

$$\mathcal{U} = e^{-i\mathfrak{p}^1}, \qquad \mathcal{V} = e^{i\mathfrak{q}^m}. \tag{10.33}$$

Now, let us consider the effect of the interaction I_1 of (9.5). For charge positions $a_1^{(\mathfrak{c})}, a_2^{(\mathfrak{c})}$ that are independent of time, I_1 contributes a total derivative term in (10.9):

$$-\frac{1}{2\pi}\sum_{\mathfrak{a}=1}^{m}\left(a_{1}^{(\mathfrak{a})}-a_{1}^{(\mathfrak{a}+m)}\right)\int d\mathfrak{q}^{m}-\frac{1}{2\pi}\int\sum_{\mathfrak{a}=1}^{m}\left(a_{2}^{(\mathfrak{a})}-a_{2}^{(\mathfrak{a}+m)}\right)d\mathfrak{p}^{\mathfrak{a}},$$

where we used (10.12). We now claim that the effect of the interaction term I_1 is to modify (10.33) to

$$\mathcal{U} = e^{-i\mathfrak{p}^{1} + \frac{i}{k}\sum_{a=1}^{m} \left(a_{1}^{(a)} - a_{1}^{(m+a)}\right)}, \qquad \mathcal{V} = e^{i\mathfrak{q}^{m} + \frac{i}{k}\sum_{a=1}^{m} \left(a_{2}^{(a)} - a_{2}^{(m+a)}\right)}.$$
 (10.34)

This is not so obvious for static charge positions, and to see it we actually need to let the position, say $a_1^{(a)}$, vary as a function of time. It can then be checked that an initial state $|i\rangle$ at time $t = -\infty$ evolves into

$$|f\rangle = e^{\frac{i}{2\pi}[a_1^{(\mathfrak{a})}(\infty) - a_1^{(\mathfrak{a})}(-\infty)]\mathfrak{q}^m}|i\rangle$$

at $t = \infty$. But in order for $\mathcal{U}|i\rangle$ to evolve into $\mathcal{U}|f\rangle$, the operator \mathcal{U} must add a phase of $\frac{1}{k}a_1^{(\mathfrak{a})}(-\infty)$ to $|i\rangle$ and a similar phase of $\frac{1}{k}a_1^{(\mathfrak{a})}(\infty)$ to $|f\rangle$, since $\mathcal{U}^{-1}\mathfrak{q}^m\mathcal{U} = \mathfrak{q}^m + \frac{2\pi}{k}$. A similar method can be used to derive the extra phase $\frac{1}{k}\sum_{\mathfrak{a}=1}^m \left(a_2^{(\mathfrak{a})} - a_2^{(m+\mathfrak{a})}\right)$ in the action of \mathcal{V} .

Now we can compare the above discussion with Chern-Simons theory. It was argued in [1] that in terms of U(1) Tr-S theory (namely Chern-Simons theory) defined on T^2 in the x_1x_2 directions, \mathcal{U} and \mathcal{V} can be understood as gauge transformations with discontinuous gauge parameters:

$$\Lambda_{\mathcal{U}} = e^{\frac{i}{k}x_1}, \qquad \Lambda_{\mathcal{V}} = e^{\frac{i}{k}x_2}, \qquad (10.35)$$

where the coordinates x_1, x_2 take values in $[0, 2\pi)$. (This was argued by relating \mathcal{U}, \mathcal{V} to momentum and winding number in type-IIA and then mapping these quantum numbers to electric fluxes on the D3-brane in type-IIB.) Equation (10.35) in conjunction with (10.34) allows us to directly map states of Chern-Simons theory to states of the system we got from geometric quantization of T^{2m} by, for example, mapping eigenstates of \mathcal{U} in (10.35) to eigenstates of \mathcal{U} in (10.34). Moreover, the extra $a_i^{(\mathfrak{a})}$ -dependent phase that we got in (10.34) has a natural interpretation in Chern-Simons theory. This is precisely the phase that we would expect to pick up when acting with a gauge transformation (10.35) on a system that contains m positive charges at positions $(a_1^{(\mathfrak{a})}, a_2^{(\mathfrak{a})})$ and m negative charges at $(a_1^{(\mathfrak{a}+m)}, a_2^{(\mathfrak{a}+m)})$, for $\mathfrak{a} = 1, \ldots, m$. This concludes our map from the Hilbert space of the geometric quantization system to the Hilbert space of U(1) Chern-Simons theory.

Chapter 11 U(n) gauge group

We now turn to the nonabelian case with a U(n) gauge group. Our goal is to calculate the Witten Index of Tr-S theory on T^2 as a function of k, n and the number of charge pairs m. We will begin in §11.1 with a few examples of sectors for n = 2, including a brief description of all its sectors with low m. We then show in §11.2 that only decongested sectors contribute to the Witten Index. This greatly simplifies the computation, since decongested sectors are equivalent to a product of decoupled U(1)Hilbert spaces. We describe the results in §11.3. A reader who wishes to skip the details is advised to jump directly to §11.3. In §11.4 we test our results by rewriting the Witten Index as a trace of products of Wilson loop operators in Tr-S theory without charges (m = 0). This provides us with a consistency check, and also allows us to calculate the eigenvalues of Wilson loop operators acting on the m = 0 Hilbert space. Appendix 13.2 includes some additional details of the combinatorics involved, and for curiosity, we included in Appendix 13.2.2 a combinatorical derivation of the total number of sectors. Interestingly, it is described by a Fibonacci sequence.

11.1 Examples of U(2) sectors and states

As explained in §9.3, each sector corresponds to a different choice of the binding matrix \mathfrak{B} and a permutation $\sigma \in S_n$ that accompanies the action of the S-R-twist on the *n* strings. Since for n = 2 the permutation group is $S_n \simeq \mathbb{Z}_2$, in the following we shall simply express the permutation as $\sigma \in \{1, -1\}$, and write it as a subscript of \mathfrak{B} , so a sector will be denoted as $\mathfrak{B}_{\pm 1}$. Physically, $\sigma = 1$ implies that each string's endpoint is connected to its own starting point so that we have two strings, each with winding number 1 along the x_3 circle, whereas $\sigma = -1$ means that the string's endpoint is connected to the other string's starting point so that we end up with one string with winding number 2. If the permutation results in an equivalent configuration, the subscript is omitted. This happens when at least two strings start on the same D2-brane (i.e., the binding matrix is congested), and relabeling the string indices results in equivalent sectors with different σ 's (see for example the third configuration in §11.1). It is also useful to recall that our conventions are such that all strings begin at $x_3 = 0$, and the S-R-twist is located at $x_3 = 2\pi R$.

In this subsection, we will count the states of each sector following the rules of $\S9.10$, and show that in each case the low-energy action can be reduced to the form (7.4).

m = 1

We have one D2-brane and a string winding number of n = 2. This yields three sectors described below [where $(\cdots)^{\top}$ denotes the transposed matrix]. The diagram for each sector is a miniature of Figure 9.4, with the S-R-twists colored to reflect σ ,¹ while the black circles depict junctions where open strings attach to D2-branes.



One open string of winding number 1 bound to the D2-brane and one closed string of winding number 1. We saw in §10.1 that the open string plus the D2brane system yields k states, while the closed string, being dual to the abelian Chern-Simons theory without charges, also gives rise to k states. In total, we get $k \times k = k^2$ states.



 $\begin{array}{c|c} 2 \times & \times & 2 \\ 1 \times & \times & 1 \end{array}$

2. $\mathfrak{B} = (1\,0)_{-1}^{\top}$

One open string of winding number 2. Let the string start at $z = \mathbf{q} + i\mathbf{u}$ and end at $e^{2i\nu}(\mathbf{q} + i\mathbf{u})$ (the phase is 2ν because the string passes through the S-R-twist twice before ending on the D2-brane). For k = 2, for which $\nu = \frac{\pi}{2}$, this means $\mathbf{u} = -\mathbf{u} \mod 2\pi$, or $\mathbf{u} = 0$ or π . For each choice of \mathbf{u} , the string starts at $x_{10} = \mathbf{q}$ on the D2-brane and end at $x_{10} = -\mathbf{q}$, so the effective action is

$$I_{k=2} = \frac{2}{2\pi} \int \mathfrak{p} d\mathfrak{q}$$
, with $\mathfrak{u} = 0$, or π

This gives us 2+2=4 states. Note that in this case the parameter \mathfrak{u} is discrete and decoupled from the geometrically quantized T^2 , which in turn is described only by $(\mathfrak{p}, \mathfrak{q})$.

¹For configurations equivalent under σ , we chose $\sigma = 1$ for our analysis.

For both k = 1 and k = 3, the phase e^{2iv} is effectively what we had for k = 3 case in §10.1. Therefore, from similar analysis, the effective action is

$$I_{k=1,3} = \frac{3}{2\pi} \int \mathfrak{p} d\mathfrak{q} \,,$$

which gives rise to 3 states each.

 $\begin{array}{c|c} 2 \times & \times & 2 \\ 1 \times & & \times & 1 \end{array}$

3. $\mathfrak{B} = (1\,1)^{\top}$

Here we have two open strings of winding number 1 both of which bound to the D2-brane. For the D2-brane worldvolume gauge field this means that there are twice as many charged particles as in the sector $(10)_1^{\top}$. Therefore, the bosonic part of the action is also twice that of the $(10)_1^{\top}$ sector, i.e.,

$$\frac{1}{2\pi}\int 2k\mathfrak{p}d\mathfrak{q}.$$

This is a *congested* sector that also has 4 real fermionic zero modes, as will be shown in §11.2. The bosonic part of the action has 4 states, but the actual Hilbert space is more complicated because of the fermionic zero modes and because of possible interactions between the bosonic and fermionic modes.

m = 2

We have seven sectors in the m = 2 case, as briefly described below. For each sector, the effective action is given by

$$I = \frac{1}{2\pi} \int \mathfrak{B}_{\mathfrak{i}\mathfrak{a}} \mathfrak{p}^{\mathfrak{a}} (d\mathfrak{q}^{\mathfrak{i}\mathfrak{a}} - d\mathfrak{q}^{\mathfrak{i}(\mathfrak{a}-1)}) \,,$$

but the q^{ia} variables are constrained by the relations (9.37)-(9.39). We derive the dimension of the Hilbert space of each sector in detail for k = 2, but simply state the results for k = 1, 3.

1. $\mathfrak{B} = \left(\begin{array}{cc} 1 & 1\\ 0 & 0 \end{array}\right)_1$



One closed string of winding number 1 breaks on each of the D2-branes to form two open strings, one of which passes through the S-R-twist once. In addition, there is also a closed string. The open string states were analyzed in §10.2 to give rise to k states, while the closed string gives rise to k states as well (as mentioned in the first sector of §11.1). Thus, there are a total of k^2 states in this sector.



2.
$$\mathfrak{B} = \left(\begin{array}{cc} 1 & 1\\ 0 & 0 \end{array}\right)_{-1}$$

One string connects the two D2-branes [and hence $\mathfrak{u}^{11} = \mathfrak{u}^{12} = \mathfrak{u}^1 = \mathfrak{u}^2 \equiv \mathfrak{u}$ from (9.38)], while the other starts at the second D2-brane, winds around the x_3 circle and passes through the S-R-twist twice before ending on the first D2brane. For k = 2, the string that starts on the second D2-brane at $\mathfrak{q}^{12} + i\mathfrak{u}$ ends on the first brane at $-\mathfrak{q}^{12} - i\mathfrak{u}$, giving us the constraint $2\mathfrak{u} = 0$ (modulo 2π), which implies $\mathfrak{u} = 0$, or π . The effective action becomes

$$I_{k=2} = \frac{1}{2\pi} \int \left[\mathfrak{p}^1 \left(d\mathfrak{q}^{11} + d\mathfrak{q}^{12} \right) + \mathfrak{p}^2 \left(d\mathfrak{q}^{12} - d\mathfrak{q}^{11} \right) \right],$$

which gives us 2 states for each of the two possible values of \mathfrak{u} . In total, we get 4 states. For k = 1, 3, the computation is similar to that of the second sector of §11.1, and we get 3 states in both cases.



 $\begin{array}{c|c} 2 \times & & & \\ 1 \times & & & \\ \end{array} \xrightarrow{} & & \\ & & & \\ \end{array} \begin{array}{c} \times & 2 \\ & & \\ & & \\ & & \\ \end{array} \begin{array}{c} \times & 2 \\ & & \\ & & \\ & & \\ \end{array} \begin{array}{c} \times & 2 \\ & & \\ & & \\ & & \\ \end{array} \begin{array}{c} \times & 2 \\ & & \\ & & \\ & & \\ & & \\ \end{array}$

One open string starts and ends on each D2-brane, and each string passes through the S-R-twist once. It is easy to see that the two strings are completely decoupled from each other, each being bound to a separate D2-brane. We therefore get two decoupled n = 1 sectors, each with m = 1. The action is a sum of two terms:

$$I_{k=2} = \frac{2}{2\pi} \int \left[\mathfrak{p}^1 d\mathfrak{q}^{11} + \mathfrak{p}^2 d\mathfrak{q}^{12} \right],$$

which gives us 4 states. A similar computation gives us 1 and 9 states for k = 1, 3, respectively.



4. $\mathfrak{B} = \begin{pmatrix} 1 & 0 \\ 0 & 1 \end{pmatrix}_{-1}$

starts on the first D2-brane, winds around the x_3 circle once, passing through the S-R-twist, before ending on the second D2-brane. For k = 2, it starts at $\mathfrak{q}^{11} + i\mathfrak{u}^{11}$ and ends at $\mathfrak{q}^{21} + i\mathfrak{u}^{21} = -\mathfrak{u}^{11} + i\mathfrak{q}^{11}$. The other open string starts on the second D2-brane at $\mathfrak{q}^{22} + i\mathfrak{u}^{22}$, passes the S-R-twist once before ending on the first D2-brane at $\mathfrak{q}^{10} + i\mathfrak{u}^{10} = -\mathfrak{u}^{22} + i\mathfrak{q}^{22}$. Since $-\mathfrak{u}^{11} + i\mathfrak{q}^{11}$ and $\mathfrak{q}^{22} + i\mathfrak{u}^{22}$ are on the same D2-brane, it follows that $\mathfrak{q}^{11} = \mathfrak{u}^{22} = -\mathfrak{q}^{10}$, and since $\mathfrak{q}^{11} + i\mathfrak{u}^{11}$ and $-\mathfrak{u}^{22} + i\mathfrak{q}^{22}$ are on the same D2-brane it follows that $\mathfrak{u}^{11} = \mathfrak{q}^{22} = -\mathfrak{q}^{21}$. We therefore get the effective action

$$\frac{2}{2\pi}\int \left[\mathfrak{p}^1 d\mathfrak{q}^{11} + \mathfrak{p}^2 d\mathfrak{q}^{22}\right],$$

which has the same form as that in Sector 3, and thus there are 4 states. A similar computation gives 3 states for both k = 1, 3 respectively.



One open string stretches between the two D2-branes and is located at $q^{11} + iu^1$ for k = 2. Another string starts on the second D2-brane, passes through the S-R-twist and winds around the x_3 circle once before ending on the first D2brane. It starts at $q^{12} + iu^2$ and ends at $-u^2 + iq^{12}$. A third string starts on the first D2-brane at $q^{21} + iu^1$, passes the S-R-twist once before ending on the same D2-brane at $-u^1 + iq^{21}$. Taking into account the constraints (9.37)-(9.39), we get $q^{12} = q^{21} = -q^{20} = -q^{10} = u^1 = u^2$, and the bosonic part of the effective action simplifies to

$$I_{k=2} = \frac{1}{2\pi} \int \left[\mathfrak{p}^1 \left(d\mathfrak{q}^{11} + 3d\mathfrak{q}^{12} \right) + \mathfrak{p}^2 \left(d\mathfrak{q}^{12} - d\mathfrak{q}^{11} \right) \right]$$

This is a congested sector which, as will be shown in §11.2, has 4 real fermionic zero modes.





for bosonic zero modes in §9.6. Each closed string of winding number 1 breaks on each of the D2-branes to form two open strings, one of which passes through the S-R-twist once. One string starts on the second D2-brane at $q^{i2} + iu^1$, passes the twist once before ending on the first D2-brane at $-u^1 + iq^{i2}$. The remaining string stretches between the two D2-branes and is located at $q^{i1} + iu$, where $\mathfrak{u} = \mathfrak{u}^1 = \mathfrak{u}^2$. We have two sets of such strings, for $\mathfrak{i} = 1, 2$. Taking into account the constraints, we get $\mathfrak{q}^{\mathfrak{i}0} = -\mathfrak{q}^{\mathfrak{i}2} = -\mathfrak{u}$ for $\mathfrak{i} = 1, 2$, and the bosonic part of the effective action simplifies to the expression given in (9.21). This is again a congested sector which, as will be shown in §11.2, has 12 real fermionic zero modes.

We thus see that the binding matrix \mathfrak{B} and the permutation σ can be used to help us visualize the string and brane configurations and determine the effective action rather easily.

m = 3

As a last explicit example, let us enumerate the sectors for m = 3. It turns out that there are 18 sectors described by the following set of binding matrices:

In (11.1), there are 4 decongested \mathfrak{B} 's of which each permutation σ gives rise to a distinct sector, and thus these binding matrices generate 8 sectors in total. Apart from the following pair

$$\left(\begin{array}{rrr} 1 & 1 & 1 \\ 0 & 1 & 0 \end{array}\right)_{\{1,-1\}} = \left(\begin{array}{rrr} 1 & 1 & 0 \\ 0 & 1 & 1 \end{array}\right)_{\{-1,1\}},$$

which are equivalent after relabeling of the strings (note that σ has to be changed as well), the rest of the \mathfrak{B} 's remain invariant under relabeling. There are thus a total of 18 different sectors for m = 3.

11.2 Counting fermionic zero modes

In order to properly compute the Witten index of our system in the type-IIA picture, we will now count the number of fermionic zero modes in each sector characterized by the binding matrix \mathfrak{B} and permutation σ . If a sector does not support a

fermionic zero mode, then its contribution to the Witten index is just the number of ground states of its Hilbert space; if on the other hand a sector does support fermionic zero modes, then after quantization, its Hilbert space will contain an equal number of bosonic and fermionic ground states, thereby making the net contribution to the Witten index zero. It is therefore crucial in the computation of the Witten index to determine which sectors support fermionic zero modes and which sectors do not.

In §11.2, we will address this question for the three sectors that arise in the case n = 2 with m = 1, as discussed in §11.1. This simplest example will serve to illustrate the salient points of the discussion. We will then tackle the cases with general m in §11.2. For ease of discussion, we will explicitly treat k = 2 and n = 2 cases only; generalization to other values of k and n however is straightforward, and leads to the same conclusion.

m = 1

Of the three sectors described in §11.1, the first two sectors do not support fermionic zero modes. This essentially follows from our abelian result in §10.3. Sector 1 consists of the abelian n = 1, m = 1 sector plus a closed string, neither of which supports a fermionic zero mode. The fermionic zero modes of Sector 2 must satisfy the same set of equations as those of the abelian n = 1, m = 1 sector, except for those coming from the S-R-twist. In other words, in the notation of Figure 11.1, the $\psi_{1\mathfrak{a}'}$ for $\mathfrak{a}' = 0, 1$ and $\lambda_{1\mathfrak{i}'}$ for $\mathfrak{i}' = 0, 1$ (note that $\lambda_{11} = \lambda_{12}$ and $\psi_{20} = \psi_{21}$ in this sector) will satisfy all the equations of §10.3 with \mathcal{P} replaced by \mathcal{P}^2 , because the open string starting on the D2-brane passes through the S-R-twist twice before ending on the same D2-brane. Therefore, the boundary condition from the S-R-twist now reads

$$\psi_{11} = \mathcal{P}\psi_{20} \equiv \mathcal{P}\psi_{21} = \mathcal{P}^2\psi_{10} \,,$$

due to the permutation $\sigma = -1$. But the argument otherwise does not change, because the only property of \mathcal{P} that we used there was the fact that it does not have an eigenvalue +1, and neither does \mathcal{P}^2 . We conclude that Sector 2 does not support fermionic zero modes.

It remains to consider Sector 3. The full set of equations that we need to solve is as follows (see Figure 11.1). First, we have the chirality conditions:

$$0 = (1 + \Gamma^{023})\psi_{10} = (1 + \Gamma^{023})\psi_{11} = (1 + \Gamma^{023})\psi_{20} = (1 + \Gamma^{023})\psi_{21}, \quad (11.2)$$

$$0 = (1 + \Gamma^{09\natural})\lambda_{10} = (1 + \Gamma^{09\natural})\lambda_{11} = (1 + \Gamma^{09\natural})\lambda_{12}.$$
(11.3)

Then, we have the boundary conditions at the end of the D2-branes (9.36):

$$0 = (1 - \Gamma^{\natural})(1 + \Gamma^{0145678})\lambda_{10} = (1 - \Gamma^{\natural})(1 + \Gamma^{0145678})\lambda_{12}.$$
(11.4)

Next, we have two junctions with boundary conditions (9.28)-(9.29), which read:

$$(1 - \Gamma^{239\natural})\psi_{10} = (1 - \Gamma^{239\natural})\psi_{11} = (1 - \Gamma^{239\natural})\lambda_{10} = (1 - \Gamma^{239\natural})\lambda_{11}, \quad (11.5)$$

$$0 = (1 + \Gamma^{239\natural})(\psi_{10} - \psi_{11} + \Gamma^{39}\lambda_{11} - \Gamma^{39}\lambda_{10}).$$
(11.6)



Figure 11.1: Fermionic zero modes on the D2-brane and F1-strings for (a) Sectors 1 and 2, and (b) Sector 3 of the m = 1 cases listed in §11.1. In (a), we have trivial identifications $\psi_{20} = \psi_{21}$ and $\lambda_{10} = \lambda_{11}$.

for the first junction, and

$$(1 - \Gamma^{239\natural})\psi_{20} = (1 - \Gamma^{239\natural})\psi_{21} = (1 - \Gamma^{239\natural})\lambda_{11} = (1 - \Gamma^{239\natural})\lambda_{12}, \quad (11.7)$$

$$0 = (1 + \Gamma^{239\natural})(\psi_{20} - \psi_{21} + \Gamma^{39}\lambda_{12} - \Gamma^{39}\lambda_{11}).$$
 (11.8)

for the second junction. And finally we have the S-R-twist condition (9.24)-(9.25):

$$\psi_{11} = \mathcal{P}\psi_{10}, \qquad \psi_{21} = \mathcal{P}\psi_{20}.$$
 (11.9)

To solve these equations, we first eliminate the fermionic mode λ_{11} which lives on the middle section of the D2-brane. To do this, we note that the junction conditions (11.5)-(11.6) together imply

$$\lambda_{11} = \frac{1}{2} (1 + \Gamma^{239\natural}) \lambda_{11} + \frac{1}{2} (1 - \Gamma^{239\natural} \lambda_{11} = \lambda_{10} + \frac{1}{2} (1 + \Gamma^{239\natural}) \Gamma^{39} (\psi_{10} - \psi_{11}) . \quad (11.10)$$

It is not hard to check that if we set λ_{11} to the RHS of (11.10), and if we assume that $\lambda_{10}, \psi_{10}, \psi_{11}$ satisfy the chirality conditions that are required of them in (11.2)-(11.3), then λ_{11} will automatically satisfy the chirality condition that is required of it in (11.3). It follows that we can safely eliminate λ_{11} from the equations using (11.10). But if we choose to eliminate λ_{11} from the second junction (11.7)-(11.8), we get, instead of (11.10),

$$\lambda_{11} = \lambda_{12} + \frac{1}{2} (1 + \Gamma^{239\natural}) \Gamma^{39}(\psi_{21} - \psi_{20}).$$
(11.11)

Comparing (11.10) and (11.11), we obtain

$$\lambda_{10} + \frac{1}{2}(1 + \Gamma^{239\natural})\Gamma^{39}(\psi_{10} - \psi_{11}) = \lambda_{12} + \frac{1}{2}(1 + \Gamma^{239\natural})\Gamma^{39}(\psi_{21} - \psi_{20}).$$
(11.12)

Now, we need to solve (11.4)-(11.9) together with (11.12) for

$$\lambda_{10}, \lambda_{12}, \psi_{10}, \psi_{11}, \psi_{20}, \psi_{21}$$

that are subject to the chirality conditions specified in (11.2)-(11.3).

To proceed, we note that we can combine (11.5) with (11.7) to get

$$(1 - \Gamma^{239\natural})\psi_{10} = (1 - \Gamma^{239\natural})\psi_{11} = (1 - \Gamma^{239\natural})\psi_{20} = (1 - \Gamma^{239\natural})\psi_{21}$$

= $(1 - \Gamma^{239\natural})\lambda_{10} = (1 - \Gamma^{239\natural})\lambda_{12}$. (11.13)

Let

$$\zeta_1 \equiv \lambda_{10} - \lambda_{12}$$

It satisfies the same set of equations (10.17)—(10.19) of the abelian case: (10.17) because of the chirality condition (11.3) on λ_{ab} , (10.18) because both λ_{10} and λ_{12} have the boundary conditions (11.4) that is dual to type-IIB strings ending on (formal) NS5-branes, and (10.19) because of (11.13). Hence, the same argument we used before, next to (10.20)-(10.22), implies $\zeta_1 = 0$ again, and so

$$\lambda_{10} = \lambda_{12} \,. \tag{11.14}$$

Next, substitute (11.14) into (11.12) to obtain

$$0 = (1 + \Gamma^{239\natural})(\psi_{10} - \psi_{11} + \psi_{20} - \psi_{21}).$$
(11.15)

On the other hand, from (11.13) we have

$$0 = (1 - \Gamma^{239\natural})(\psi_{10} - \psi_{11} + \psi_{20} - \psi_{21}).$$
(11.16)

The two equations (11.15) and (11.16) together imply

$$\psi_{10} - \psi_{11} + \psi_{20} - \psi_{21} = 0 \,,$$

or

$$\psi_{10} + \psi_{20} = \psi_{11} + \psi_{21} \,. \tag{11.17}$$

But from the boundary condition (11.9) that describes the S-R-twist we can write (11.17) as:

$$\psi_{11} + \psi_{21} = \mathcal{P}(\psi_{10} + \psi_{20}),$$

and since \mathcal{P} does not have an eigenvalue +1, we get

$$\psi_{10} + \psi_{20} = \psi_{11} + \psi_{21} = 0.$$
(11.18)

Now, from (11.13) we have

$$(1 - \Gamma^{239\natural})(\psi_{10} - \psi_{20}) = 0,$$

and together with (11.18), we deduce that

$$(1 - \Gamma^{239\natural})\psi_{10} = (1 - \Gamma^{239\natural})\psi_{20} = 0, \qquad (11.19)$$

and hence all the other expressions appearing in (11.13) also vanish.

If we now define

$$\xi_1 \equiv \lambda_{10} + \lambda_{12} \,,$$

the result of the last paragraph implies that ξ_1 satisfies the same equations that ζ_1 satisfies in (10.17)-(10.19), and hence $\xi_1 = 0$. Therefore, we conclude that

$$\lambda_{10} = \lambda_{12} = 0. \tag{11.20}$$

At this point, there is essentially only one unknown variable, say ψ_{11} . The other variables ψ_{ij} and λ_{11} are determined in terms of it by (11.11), (11.18), and (11.9). The equations it should satisfy are

$$0 = (1 - \Gamma^{239\natural})\mathcal{P}^{-1}\psi_{11} = (1 - \Gamma^{239\natural})\psi_{11} = (1 + \Gamma^{023})\mathcal{P}^{-1}\psi_{11} = (1 + \Gamma^{023})\psi_{11}.$$
 (11.21)

where we substituted $\psi_{10} = \mathcal{P}^{-1}\psi_{11}$ from (11.9). Next, we recall that for k = 2 the operator \mathcal{P} realizes a rotation by $\frac{\pi}{2}$ in four transverse two-planes, and so

$$\mathcal{P}\Gamma^{239\natural}\mathcal{P}^{-1} = -\Gamma^{1238}, \qquad \mathcal{P}\Gamma^{023}\mathcal{P}^{-1} = \Gamma^{023}.$$

Using these relations, we can write (11.21) as

$$\psi_{11} = \Gamma^{239\natural} \psi_{11} = -\Gamma^{023} \psi_{11} = -\Gamma^{1238} \psi_{11}.$$
(11.22)

We can now work in a basis for which Γ^{23} , Γ^{18} , $\Gamma^{9\natural}$ and Γ^{0} are simultaneously diagonal. It is then easy to see that (11.22) has 4 linearly independent solutions. This corresponds to 4 zero modes of our system. These four real zero modes transform as singlets under the SU(2) factor of the $SU(2) \times U(1)$ symmetry group that was mentioned at the end of §9.2 and they have ± 1 charges under the U(1) factor, which is generated by $\frac{i}{2}(\Gamma^{45} + \Gamma^{67})$. These statements are easy to derive from (11.22), which together with (9.23) implies that $\Gamma^{4567}\psi_{11} = -\psi_{11}$. (The other fermionic fields of the problem are determined in terms of ψ_{11} and are easily seen to satisfy the same chirality condition.) In this subsection we have restricted for simplicity to the k = 2case, but the same result of 4 zero modes is also true for the other cases k = 1, 3.

m > 1

Having considered the fermionic zero modes for the n = 2, m = 1 case, we now move on to consider the cases with general m. In this subsection, we prove the following criterion for the existence of fermionic zero modes: the fermionic zero modes exist precisely in those sectors for which $\mathfrak{B}_{1\mathfrak{a}} = \mathfrak{B}_{2\mathfrak{a}} = 1$ for some $\mathfrak{a} = 1, \ldots, m$. In



Figure 11.2: An example of sectors without fermionic zero modes. Only one string is attached to each D2-brane.

other words, they exist if and only if there is at least one D2-brane to which both open strings attach. These are what we called *congested* sectors.

It is easy to see that there is no fermionic zero mode if for each of the *m* D2-branes there is only one string attached to it (see Figure 11.2 for an example). For those sectors for which the permutation σ that accompanies the S-R-twist is the identity, we can divide the D2-branes into two groups: those connecting to the F1-string that is labeled by $\mathbf{i} = 1$, and those connecting to the F1-string labeled by $\mathbf{i} = 2$. Each group together with the respective F1-string then forms an abelian system discussed in §10.3, and hence fermionic zero modes are absent.

For sectors with $\sigma = -1$, we can divide the D2-brane indices $\mathfrak{a} = 1, \ldots, m$ into two groups so that those with $\mathfrak{a} = \mathfrak{a}_1, \ldots, \mathfrak{a}_k$ attach to the F1-string that is labeled by $\mathfrak{i} = 1$, and those with $\mathfrak{a} = \mathfrak{a}_{k+1}, \ldots, \mathfrak{a}_m$ to the F1-string labeled by $\mathfrak{i} = 2$. Then the system can again be regarded as an abelian case, the D2-branes now being arranged in the new order $\mathfrak{a}_1, \ldots, \mathfrak{a}_m$, except for the S-R-twist condition, which should now read

$$\psi_{1\mathfrak{a}_k} = \mathcal{P}\psi_{2\mathfrak{a}_{k+1}}, \quad \psi_{2\mathfrak{a}_m} = \mathcal{P}\psi_{1\mathfrak{a}_1}$$

The effect of this new boundary condition is that instead of (10.23), we get

$$\psi_{1\mathfrak{a}_1} = \psi_{1\mathfrak{a}_2} = \cdots = \psi_{1\mathfrak{a}_k} = \mathcal{P}\psi_{2\mathfrak{a}_{k+1}} = \cdots = \mathcal{P}\psi_{2\mathfrak{a}_m} = \mathcal{P}^2\psi_{1\mathfrak{a}_1}.$$

But since \mathcal{P}^2 does not have an eigenvalue +1, the conclusion (10.25) remains the same, and hence there is no zero mode.

So let us now establish the fact that if there is at least one D2-brane to which both strings attach, there are fermionic zero modes (see Figure 11.3 for an example). Let us first note that if only one string is attached to the \mathfrak{a}^{th} brane, then we have

$$\psi_{1,\mathfrak{a}-1} = \psi_{1\mathfrak{a}}, \quad \psi_{2,\mathfrak{a}-1} = \psi_{2\mathfrak{a}}.$$
 (11.23)



Figure 11.3: An example of sectors *with* fermionic zero modes. Both strings attach to the first, second, and fourth D2-branes.

This follows from the abelian result of §10.3: for example, if it is the i = 1 F1-string that attaches to the \mathfrak{a}^{th} D2-brane, then we have a trivial identification $\psi_{2,\mathfrak{a}-1} = \psi_{2\mathfrak{a}}$ (and $\lambda_{\mathfrak{a}1} = \lambda_{\mathfrak{a}2}$), while the equality $\psi_{1,\mathfrak{a}-1} = \psi_{1\mathfrak{a}}$ follows from the same reasoning that leads to (10.23) in §10.3. If, on the other hand, both strings attach to the \mathfrak{a}^{th} D2-brane, then we have a weaker identity

$$\psi_{1,\mathfrak{a}-1} + \psi_{2,\mathfrak{a}-1} = \psi_{1\mathfrak{a}} + \psi_{2\mathfrak{a}} \,. \tag{11.24}$$

This follows from the same reasoning that leads to (11.17) in §11.2.

From (11.23) and (11.24), we now get

$$\psi_{10} + \psi_{20} = \psi_{11} + \psi_{21} = \dots = \psi_{1m} + \psi_{2m},$$
 (11.25)

and from the S-R-twist,

$$\psi_{1m} + \psi_{2m} = \mathcal{P}(\psi_{10} + \psi_{20}), \qquad (11.26)$$

regardless of the choice of permutation σ . These two relations together imply that

$$\psi_{10} + \psi_{20} = \psi_{11} + \psi_{21} = \dots = \psi_{1m} + \psi_{2m} = 0.$$
 (11.27)

Let us now divide the D2-brane indices $\mathfrak{a} = 1, \ldots, m$ so that the subset $\{\mathfrak{a}_1, \ldots, \mathfrak{a}_l\}$ refer to those D2-branes to which both strings attach. Then we can eliminate the fermionic modes $\lambda_{\mathfrak{a}_1 1}, \ldots, \lambda_{\mathfrak{a}_l 1}$ that reside on the middle D2-brane segments, similarly to (11.10). Then the reasoning leading to (11.20) gives us

$$\lambda_{\mathfrak{a}_10} = \lambda_{\mathfrak{a}_12} = \lambda_{\mathfrak{a}_20} = \lambda_{\mathfrak{a}_22} = \cdots = \lambda_{\mathfrak{a}_l0} = \lambda_{\mathfrak{a}_l2} = 0.$$

For $\mathfrak{a} \notin {\mathfrak{a}_1, \ldots, \mathfrak{a}_l}$, only one string attaches to the \mathfrak{a}^{th} D2-brane, so we have trivial identification $\lambda_{\mathfrak{a}1} = \lambda_{\mathfrak{a}0}$ or $\lambda_{\mathfrak{a}1} = \lambda_{\mathfrak{a}2}$. But since $\lambda_{0\mathfrak{a}} = \lambda_{2\mathfrak{a}} = 0$ from the same argument

as in the abelian result, we have $\lambda_{1\mathfrak{a}} = 0$ in either case. Also for $\mathfrak{a} \notin {\mathfrak{a}_1, \ldots, \mathfrak{a}_l}$, we have $\psi_{\mathfrak{i}\mathfrak{a}} = \psi_{\mathfrak{i},\mathfrak{a}-1}$ for $\mathfrak{i} = 1, 2$ from (11.23). It follows therefore that the only independent variables that we have at this point are $\psi_{1\mathfrak{a}_1}, \ldots, \psi_{1\mathfrak{a}_l}$, since $\psi_{2\mathfrak{a}_1}, \ldots, \psi_{2\mathfrak{a}_l}$ can be recovered from (11.27), and $\psi_{\sigma(\mathfrak{i})0} = \mathcal{P}^{-1}\psi_{\mathfrak{i}m}$. The remaining equations are

$$0 = (1 - \Gamma^{239\natural})\psi_{1\mathfrak{a}} = (1 + \Gamma^{023})\psi_{1\mathfrak{a}}, \qquad \mathfrak{a} \in \{\mathfrak{a}_1, \dots, \mathfrak{a}_l\}.$$
(11.28)

Similarly to the discussion following (11.21), there are 4 zero modes for ψ_{1a_l} and 8 zero modes for each of $\psi_{1a_1}, \ldots, \psi_{1a_{l-1}}$. We get 8l - 4 zero modes in total.

11.3 The Witten Index — results

According to the discussion of §11.2 (and its extension to n > 2), sectors with a congested binding matrix have a nonzero number of zero modes. They therefore do not contribute to the Witten Index. Only sectors with no zero modes contribute to the Witten Index, and these are precisely the decongested sectors.

By our definition at the end of §9.4, a decongested sector is a sector whose binding matrix has exactly one '1' in each column. It can be alternatively described as follows. Start with p closed strings of winding numbers n_1, n_2, \ldots, n_p , such that $n = n_1 + \cdots + n_p$, and attach each of the m D2-branes to one string. The point of attachment along the string is also important, and since the x_3 coordinate of the D2brane is fixed, there are n_j choices for the j^{th} string, since the string passes through the x_3 coordinate n_j times. The partition $n = n_1 + \cdots + n_p$ determines the conjugacy class $[\sigma]$ of the permutation $\sigma \in S_n$ (where σ is represented as a product of cycles and n_1, \ldots, n_p are the lengths of the cycles), and the points of attachment determine the binding matrix. Thus, there are initially n^m choices for the attachment points, but choices that are equivalent up to relabeling of the strings should be counted only once.

Let m_j be the number of D2-branes that end up being attached to the j^{th} string. Then $m = m_1 + \cdots + m_p$, and it is not difficult to see that the Hilbert space of the corresponding sector is equivalent to a tensor product $\otimes_{j=1}^{p} \mathcal{H}_j(n_j, m_j)$ of decoupled Hilbert spaces. The dimension of $\mathcal{H}_j(n_j, m_j)$ can be determined explicitly from the *effective twist phase* $e^{in_j v}$. Since we have the constraint $n_j \leq n < \mathbf{r}$, the effective phase is never trivial. In fact, in the cases relevant to us, the effective twist phase takes one of the following seven values:

$$e^{in_jv} \in \{\pm i, e^{\pm \frac{\pi i}{3}}, e^{\pm \frac{2\pi i}{3}}, -1\}.$$

The first six values are the same twist we got in §10 for U(1) theory at levels $k = \pm 2, \pm 1, \pm 3$. (It is necessary to keep track of the sign if relative parity is important.) The dimensions in these cases are given by dim $\mathcal{H}_j(n_j, m_j) = |k|$, regardless of m_j . The last phase $e^{in_j v} = -1$ is a new case that hasn't been discussed in §10, but appeared for example in the second sectors of §11.1 and §11.1 (for k = 2). As we saw there, it is not hard to check that in this case dim $\mathcal{H}_j(n_j, m_j) = 4$, independently of m_j, n_j .

We can now summarize the results for the dimensions in Table 11.1. The contribution to the Witten Index is calculated as

$$I_m(n_1,\ldots,n_j) = \sum_{\{m_j\}:\sum m_j=m} \left(\prod_{j=1}^p \dim \mathcal{H}_j(n_j,m_j)\right).$$

As an example for how the entries in Table 11.1 were derived, take the case n = 3and k = 2 with partition 3 = 1 + 1 + 1. The number of decongested binding matrices is 3^m . (This is the total number of ways to attach m D2-branes to the strings.) But this over-counts binding matrices that are related by a permutation of the string labels 1, 2, 3. There are three binding matrices in which all m D2-branes are attached to the same string (so that we have one row of all 1's and two other rows of all 0's). They are of course equivalent to one another after relabeling the string indices. Excluding these 3 configurations, to which we shall return later, we are left with $3^m - 3$ configurations, and accounting for the relabeling redundancy 3! = 6, we get $\frac{1}{6}(3^m - 3)$ inequivalent configurations. For each of this type of configurations, let m_1, m_2, m_3 denote the number of D2-branes that are attached to the $1^{st}, 2^{nd}$, and 3^{rd} string, respectively (so that $m = m_1 + m_2 + m_3$). For each configuration, the Hilbert space is a product of three Hilbert spaces of the U(1) theory with $2m_1, 2m_2$, or $2m_3$ charges, respectively. These Hilbert spaces were analyzed in §10 and have k = 2 states each. So we get a total of 8 states for each configuration of this form. On the other hand, the remaining 3 configurations, for which all m D2-branes are attached to the same string, are equivalent to each other up to relabeling. The string with its m attached D2-branes has a Hilbert space that is equivalent to that of the U(1) theory with 2mquarks, and hence possesses 2 states, while the remaining 2 unattached strings form a Hilbert space that corresponds to the $\sigma = 1$ sector of the U(2) Tr-S theory with no charges, and has 3 states [1]. The total number of states for this configuration is therefore $2 \times 3 = 6$ and altogether we get the total number of states for k = 2 and n = 3 = 1 + 1 + 1:

$$8 \times \frac{1}{6}(3^m - 3) + 6 = \frac{4}{3}3^m + 2.$$

As another example, take k = 1 and n = 4 = 2 + 2. The binding matrix has four rows, labeled by string index i = 1, ..., 4, and for definiteness we take the permutation $\sigma = (12)(34)$. For this discussion it is convenient to pretend that this sector has two closed strings, one formed by connecting i = 1 and i = 2 strings, the other by connecting i = 3 and i = 4. Each closed string has winding number 2, to which some D2-branes are possibly attached. We start by considering all 4^m possible binding matrices, and note that there are 2×2^m binding matrices for which all D2-branes are attached to the same (pretend closed) string. The other $(4^m - 2 \times 2^m)$ binding matrices have a nonzero number of D2-branes attached to each of the two (pretend closed) strings. For this latter type of configurations, the over-counting factor is 8 because we can exchange the two strings (namely exchanging $\{1,2\}$ and $\{3,4\}$, contributing an over-counting factor of 2) and within each string we can exchange the two string indices (i = 1 and i = 2 for the first string, and similarly i = 3 andi = 4 for the second). The total number of inequivalent configurations for which neither of the strings is unattached is therefore $\frac{1}{8}(4^m-2^{m+1})$. Let $m_1 > 0, m_2 > 0$ be the number of D2-branes attached to each string, with $m = m_1 + m_2$. Each string has an effective twist phase of $e^{2iv} = e^{\frac{2\pi i}{3}}$, and therefore its Hilbert space corresponds to the Hilbert space of a U(1) theory with k' = 3 (which is the value of k for which the phase is $e^{iv'} = e^{\frac{2\pi i}{3}}$, and with $2m_1$ or $2m_2$ charges. Each configuration therefore has $3 \times 3 = 9$ states. The remaining configurations have one unattached (pretend closed) string, and all m D2-branes are attached to the other (pretend closed) string. There are 2×2^m such binding matrices, with an over-counting factor of 4 = 8/2 (where 2) is the symmetry factor which corresponds to switching the two string indices of the unattached string), so we get 2^{m-1} configurations. Each configuration has a Hilbert space that corresponds to U(1) theory with k' = 3 and 2m charges inserted times the Hilbert space of U(2) theory with k = 1 in the sector $\sigma = (12)$, and no external charges. The latter Hilbert space is two-dimensional [1]. The Hilbert space of the configuration thus has a total dimension of $3 \times 2 = 6$. Altogether we find the total number of states of the n = 4 = 2 + 2 sector of the k = 1 theory to be:

$$9 \times \frac{1}{8}(4^m - 2^{m+1}) + 6 \times 2^{m-1} = \frac{9}{8}4^m + \frac{3}{4}2^m.$$

In addition to the sector-by-sector analysis described above, we can also write down closed formulas for certain types of sectors. In general, it is useful to have a formula for the number f(n,m) of non-equivalent decongested binding matrices that have at least one nonzero entry in each of the *n* rows. This corresponds to the number of configurations for which each of the *n* strings is attached to at least one D2-brane. In Appendix 13.2.1 we show that:

$$f(n,m) = \sum_{j=1}^{n} \frac{(-1)^{n-j}}{j!(n-j)!} j^m.$$
(11.29)

Using this result we can write a general expression for the Witten Index in the sectors with partition $n = 1 + 1 + \dots + 1$. Such a sector can have $1 \le l \le n$ unattached strings. The dimension of the Hilbert space of the unattached strings is the same as the number of states that a system of l identical bosons each occupying one of k states has, which is $\binom{k+l-1}{k-1}$, while the dimension of the Hilbert space of all (n-l) attached strings is k^{n-l} . So, we get a total of

$$\sum_{l=0}^{n-1} \binom{k+l-1}{k-1} k^{n-l} f(n-l,m) = \sum_{j=1}^{n} \left(\sum_{l=0}^{n-j} \frac{(-1)^{n-l-j} k^{n-l} (k+l-1)!}{(k-1)! l! j! (n-l-j)!} \right) j^m$$

states. Using this and similar techniques we get the results listed in Table 11.1. We can now add the contribution of the various sectors to the Witten Index for each k and n, listed in Table 11.1, and obtain the results of Table 11.2.

11.4 Wilson loop operators and their eigenvalues

From the expressions for the dimensions of Hilbert spaces with static charges we can get information about basic properties of Wilson loop operators along the two (spatial) cycles of T^2 . Let us define the two basic 1-cycles of T^2 , one along the x_1 direction, and the other along the x_2 direction. We denote the low-energy limits of the two supersymmetric Wilson loop operators that correspond to these cycles, and in the fundamental representation \Box of U(n), by \mathcal{W}_1 and \mathcal{W}_2 :

$$\operatorname{tr}_{\Box} \left(Pe^{i \int_{0}^{2\pi L_{1}} (A_{1}(t,x_{1},x_{2},x_{3}) + \Phi^{9}(t,x_{1},x_{2},x_{3})) dx_{1}} \right) \xrightarrow[\text{low-energy limit}]{} \mathcal{W}_{1}, \quad (11.30)$$
$$\operatorname{tr}_{\Box} \left(Pe^{i \int_{0}^{2\pi L_{2}} (A_{2}(t,x_{1},x_{2},x_{3}) + \Phi^{9}(t,x_{1},x_{2},x_{3})) dx_{2}} \right) \xrightarrow[\text{low-energy limit}]{} \mathcal{W}_{2}. \quad (11.31)$$

Here Φ^9 is the adjoint scalar from the $\mathcal{N} = 4$ multiplet that corresponds to fluctuations of the D3-branes in direction x_9 as in (9.2), and $\mathcal{W}_1, \mathcal{W}_2$ are operators on the Hilbert space $\mathcal{H}(k, n, m = 0)$ (namely, Hilbert space without external charges). Assuming that the low-energy theory is topological we expect $\mathcal{W}_1, \mathcal{W}_2$ to be independent of t, x_1, x_2 altogether.

A simple Wick rotation now allows us to derive the eigenvalues of $W_i^{\dagger}W_i$ from the dimensions of the Hilbert spaces $\mathcal{H}(k, n, m)$ as a function of m. Obviously, if the theory is topological the eigenvalues are the same for i = 1, 2. Let us compactify time on a circle with (supersymmetric) periodic boundary conditions so that $0 \leq x_0 < 2\pi T$. Tr-S theory is now formulated on T^3 in directions x_0, x_1, x_2 . Now insert the m quark and anti-quark pairs. At this point, if we let x_2 , for example, play the role of Euclidean time then every quark corresponds to a Wilson loop operator for a loop around direction x_0 . In the microscopic 3 + 1D theory, let \mathcal{W}' be such a supersymmetric Wilson loop around direction x_0 and at a fixed x_3 . In the Hilbert space of Tr-S theory on T^2 (in directions x_0, x_1), let \mathcal{W} be the operator that is the low-energy limit of \mathcal{W}' . It is, of course, independent of x_3 . Now we can write

$$\dim \mathcal{H}(k, n, m) = \operatorname{tr}[(\mathcal{W}^{\dagger}\mathcal{W})^{m}].$$
(11.32)

Thus, if we calculate dim $\mathcal{H}(k, n, m)$ for all m, we will be able to read off the eigenvalues of $\mathcal{W}^{\dagger}\mathcal{W}$. Since $\mathcal{W}^{\dagger}\mathcal{W}$ is a matrix of dimension dim $\mathcal{H}(k, n, 0)$, it follows that dim $\mathcal{H}(k, n, m)$ has to be a sum of at most dim $\mathcal{H}(k, n, 0)$ *m*-powers. We therefore expect

$$\dim \mathcal{H}(k,n,m) = \sum_{j=1}^{\dim \mathcal{H}(k,n,0)} \lambda_j(k,n)^m , \qquad (11.33)$$

where the λ_j 's are eigenvalues of $\mathcal{W}^{\dagger}\mathcal{W}$ and thus are independent of m. If indeed we can write dim $\mathcal{H}(k, n, m)$ in the form (11.33), that will provide us with a nice test of our construction, and in particular the conjecture that Tr-S is topological. Moreover, we will be able to find the eigenvalues of $\mathcal{W}^{\dagger}\mathcal{W}$.

Also, note that for k = 1, 2 we have $\mathcal{W} = \mathcal{W}^{\dagger}$ for the following reason. For k = 1, 2we find that the order $\mathbf{r} = 6, 4$ of the S-duality twist is even. Thus, the cyclic group $\{1, \mathbf{g}, \mathbf{g}^2, \dots, \mathbf{g}^{\mathbf{r}-1}\}$ that is generated by the S-duality twist $\mathbf{g} \in \mathrm{SL}(2, \mathbb{Z})$ contains $\mathbf{g}^{\mathbf{r}/2}$ which is equal to the central element $-I \in \mathrm{SL}(2, \mathbb{Z})$. This is physically equivalent to the charge-conjugation operator \mathcal{C} [see (7.3)]. Therefore, when we continuously change the x_3 position of the Wilson loop \mathcal{W}' until it completes $\frac{\mathbf{r}}{2}$ cycles along the x_3 circle, it becomes the charge conjugate $(\mathcal{W}')^{\dagger}$. Since we assumed that the low-energy limit of \mathcal{W}' is independent of x_3 , we find that \mathcal{W} is hermitian for those values of k. Thus, for k = 1, 2 the eigenvalues of \mathcal{W} are simply the square-roots of the eigenvalues of $\mathcal{W}^{\dagger}\mathcal{W}$, and are therefore known up to an overall sign.

11.5 Consistency checks

Now let us check the consistency of our results. Comparing the first three columns of Table 11.2 with the corresponding columns of Table 11.1 we observe an interesting phenomenon — whereas individual sectors in Table 11.1 do not generally conform to the required form (11.33), their total contribution in Table 11.2 does! We believe this result is a nontrivial test of our construction and derivation, and we will discuss its meaning further in $\S11.6$. Moreover, from the behavior of the Witten Index as a function of m in Table 11.2 we can read off the eigenvalues of the Wilson loop operator combination $\mathcal{W}^{\dagger}\mathcal{W}$ on the Hilbert space of Tr-S without charges (m=0). The results are listed in the last column of Table 11.2. In deriving the eigenvalues of $\mathcal{W}^{\dagger}\mathcal{W}$ we matched the expressions for the Witten Index with (11.33). In (11.33) the total number of eigenvalues, taking multiplicities into account, has to be equal to the dimension of the Hilbert space without charges. These dimensions are listed in the 4th column of Table 11.2 as dim $\mathcal{H}(k, n, 0)$, and in cases where the number of powers appearing in the expression for the Witten Index in the 3^{rd} column falls short of dim $\mathcal{H}(k, n, 0)$ we have to add zero eigenvalues. (That the number of powers is always smaller than dim $\mathcal{H}(k, n, 0)$ constitutes another consistency check.) There is, however, an independent check on these results and the number of zero eigenvalues as follows.

For k > 1, there are symmetry operators \mathcal{U}, \mathcal{V} that act on the Hilbert spaces $\mathcal{H}(k, n, 0)$. They were introduced in [1] and reviewed in §8.3. These operators have a geometrical interpretation in the type-IIA description, but in the original gauge theory description they are understood as large gauge transformations in the $U(1) \subset U(n)$ center. This latter interpretation allows us to immediately write their commutation relations with \mathcal{W} . For concreteness, let's assume that $\mathcal{W} \equiv \mathcal{W}_1$ is a Wilson loop

around the x_1 direction of the (type-IIB) T^2 . Then,

$$\mathcal{V}^{-1}\mathcal{W}\mathcal{V} = \mathcal{W}, \qquad \mathcal{U}^{-1}\mathcal{W}\mathcal{U} = e^{\frac{2\pi i}{k}}\mathcal{W}.$$
 (11.34)

For k = 2 we argued above that $\mathcal{W} = \mathcal{W}^{\dagger}$, and \mathcal{W} is therefore diagonalizable. The second equation of (11.34) then shows that the nonzero eigenvalues of \mathcal{W} must come in pairs $(\lambda, -\lambda)$, and so the multiplicities of the nonzero eigenvalues $|\lambda|^2$ of $\mathcal{W}^{\dagger}\mathcal{W}$ are all even. For k = 3 we don't have a similar argument to show that \mathcal{W} is diagonalizable, but assuming that it is, the nonzero eigenvalues of \mathcal{W} must come in triplets $(\lambda, e^{\frac{2\pi i}{3}}\lambda, e^{-\frac{2\pi i}{3}}\lambda)$ and therefore the multiplicities of the nonzero eigenvalues $|\lambda|^2$ of $\mathcal{W}^{\dagger}\mathcal{W}$ must all be divisible by 3. This is indeed the case, as we can see from Table 11.2.

For k = n = 2 we can say more. In this case \mathcal{U} and \mathcal{V} commute and we can write $\mathcal{H}(k = 2, n = 2, 0)$ as a direct sum $\bigoplus \mathcal{H}_{(u,v)}(2,2,0)$ of simultaneous eigenstates of $(\mathcal{U}, \mathcal{V})$, with eigenvalues $(u = \pm 1, v = \pm 1)$. It is easy to check that

$$\dim \mathcal{H}_{(+1,+1)} = \dim \mathcal{H}_{(+1,-1)} = \dim \mathcal{H}_{(-1,+1)} = 2, \qquad \dim \mathcal{H}_{(-1,-1)} = 0$$

Now take a state $|\psi\rangle \in \mathcal{H}_{(+1,-1)}$ and consider the $(\mathcal{U}, \mathcal{V})$ eigenvalues of $\mathcal{W}|\psi\rangle$. By (11.34) they must be (-1, -1), and since $\mathcal{H}_{(-1,-1)}$ is trivial it follows that $\mathcal{W}|\psi\rangle = 0$. Therefore, \mathcal{W} is identically zero on the two-dimensional subspace $\mathcal{H}_{(+1,-1)}$. It follows that \mathcal{W} has at least two eigenvalues that are identically zero, and so does $\mathcal{W}^{\dagger}\mathcal{W}$. From Table 11.2 we see that this is indeed the case, and that the multiplicity of the zero eigenvalue of $\mathcal{W}^{\dagger}\mathcal{W}$ is exactly 2.

11.6 Comparison with Chern-Simons theory

So far we have found the Witten Indices of Tr-S theory on T^2 in individual sectors, listed in Table 11.1, and their sum over all sectors, listed in Table 11.2. We have also seen that the results pass some nontrivial consistency checks in §11.5. These results are however supposed to provide some clues about what Tr-S theory is. Is it a known theory, or is it an entirely new theory? How should we interpret the results from Table 11.1?

As a first step, we have to know whether different "sectors" correspond to different theories, or whether they are part of the same theory. Following the results in [1] regarding the Hilbert spaces $\mathcal{H}(k, n, m = 0)$ and their decomposition as representations of the mapping class group $\mathrm{SL}(2,\mathbb{Z})$ of (the type-IIB) T^2 , it was proposed there that a sector $[\sigma]$ corresponds to a superselection sector of Tr-S theory on $\mathbb{R}^{2,1}$ — perhaps a discrete remnant of an expectation value of a Wilson loop along the compact x_3 direction. Furthermore, it was observed in [1] that strictly as representations of $\mathrm{SL}(2,\mathbb{Z})$ and operators \mathcal{U}, \mathcal{V} , the Hilbert spaces of most of the sectors are equivalent to the Hilbert spaces of (pure) Chern-Simons theories at various levels and with various gauge groups that are in general subgroups of U(n). We have reproduced the general results of [1] in Table 11.3. For n = 2, for example, the breakdown into individual sectors is reproduced in Table 11.4. (Note that, as explained in [1], the Chern-Simons theory level of the U(1) part of the gauge group is given by k' = nk in all cases.) Naturally, it was then conjectured that Tr-S in each of these sectors is equivalent to the Chern-Simons theory at the corresponding level and with the corresponding gauge group. But given the results of Table 11.1, we can now take a critical look at some of these conjectures.

To extract useful information out of Table 11.1 we need to know how to match a sector of Tr-S with m > 0 charge pairs to a sector of Tr-S with no charges (m = 0). A sector with m = 0 is described entirely by the conjugacy class $[\sigma]$ of the permutation $\sigma \in S_n$, or alternatively, by the partition $n = n_1 + n_2 + \cdots + n_p$. A sector with m > 0, on the other hand, is described by $[\sigma]$ together with a binding matrix \mathfrak{B}_{ia} , up to relabeling of string indices i, and for general sectors, combinations $(\mathfrak{B}, [\sigma])$ and $(\mathfrak{B}', [\sigma'])$ with different conjugacy classes $([\sigma] \neq [\sigma'])$ may be equivalent. In general, therefore, we cannot unambiguously assign a sector of m = 0 theory to a given sector with m > 0. This is also clear because the m > 0 sectors have open strings while the m = 0 sectors only have closed strings.

However, if we restrict ourselves to *decongested* sectors we can overcome this problem. Since a decongested sector has exactly one pair of open strings ending on each D2-brane, we can formally align and recombine without ambiguity these two ends to form a configuration of closed strings, thereby creating a unique m = 0 sector out of a decongested m > 0 sector. In fact, the "pretend closed" terminology of §11.3 and the partitions $n = n_1 + \cdots + n_p$ appearing in Table 11.1 took advantage of this fact.

But now we face a serious obstacle. It was argued in §11.5 that any sector whose entry in Table 11.1 does not conform to (11.33) — one for which the coefficient of any m^{th} power in its contribution to the Witten Index is not an integer — cannot possibly be a stand-alone theory. For consistency we have to, at the very least, combine sectors so that their total contribution to the Witten Index will be of the form (11.33). Thus, for example, both of the k = n = 2 sectors might be individual theories corresponding to different "superselection" sectors. But the k = 2 and n = 3 sectors corresponding to the partitions 3 = 1 + 1 + 1 and 3 = 3 cannot be separate theories. Similarly, the k = 1 and k = 3 sectors with partitions 2 = 1 + 1 and 2 = 2 cannot be separate theories either. This, we have to admit, is evidence against at least some of the conjectures that are implicit in Table 11.3.

So, still focusing on the U(2) case, let us assume that we need to combine both 2 = 1 + 1 and 2 = 2 sectors for k = 1, 3, and let us remain agnostic about whether we need to combine or not the two sectors for the k = 2 case. Let us proceed and ask whether in this way Tr-S theory can still be a pure Chern-Simons theory in these cases. What can we learn from Table 11.2? We are going to make the assumption that if indeed Tr-S is identified with pure Chern-Simons theory then the Wilson loop operator \mathcal{W} is identified with a Wilson loop in Chern-Simons theory (wound around

one of the nontrivial 1-cycles of T^2). We will now compare the information from Table 11.2 about the eigenvalues of Wilson loops with what we know about U(2) Chern-Simons theory.

Wilson loop operators in U(2) Chern-Simons theory

Let us begin by reviewing the known U(2) Chern-Simons results. The Hilbert space of SU(2) Chern-Simons theory on a torus T^2 at level k'' is (k''+1)-dimensional, and as explained in [30, 56], there exists a canonical basis in which basis states are labeled by SU(2) spin $j = 0, \frac{1}{2}, \ldots, \frac{k''}{2}$, once we choose a basis of 1-cycles a and b for the first homology group $H_1(T^2; \mathbb{Z})$ of the torus. When we think of T^2 as the boundary of a solid torus, the a-cycle is the one that becomes contractible inside the solid torus, while the b-cycle remains non-trivial. The state labeled by spin j is then defined in terms of the wave-function whose value is given by the path integral of Chern-Simons theory on the solid torus with a Wilson loop in the spin j representation inserted along the b-cycle. We will denote such basis states of the Hilbert space by $|\mathfrak{m}\rangle$, with $\mathfrak{m} \equiv 2j = 0, \ldots, k''$.

The action of a Wilson loop operator $W^{(n_a,n_b)}$ in any representation of SU(2) that winds around the torus n_a times along the *a*-cycle and n_b times along the *b*-cycle was given in [57]. For our present purpose, we need the result for the Wilson loop in the fundamental representation with $n_a = 1$, $n_b = 0$:

$$W \equiv W^{(1,0)} = \sum_{\mathfrak{m}} 2\cos\frac{\pi(\mathfrak{m}+1)}{k''+2} |\mathfrak{m}\rangle\langle\mathfrak{m}|. \qquad (11.35)$$

On the other hand, the Hilbert space of U(1) Chern-Simons theory at level k' is k'-dimensional, and the Wilson loops act as

$$W^{(1,0)} = \sum_{p=0}^{k'-1} e^{\frac{2\pi i}{k'}p} |p\rangle\langle p|, \qquad W^{(0,1)} = \sum_{p=0}^{k'-1} |p+1\rangle\langle p|, \qquad (11.36)$$

where $|p\rangle$ for $p = 0, \ldots, k' - 1$ are the basis states.

We can now combine the results for the U(1) and SU(2) theories to construct the Hilbert space for the U(2) theory. The Hilbert space, denoted by $U(2)_{k',k''}$, can be obtained by first taking the tensor product of the Hilbert space of U(1) theory at level k' = 2k and that of SU(2) at level k'', and then restricting to the subspace where a certain "large" gauge transformation acts trivially. This is because the group U(2)is not simply the product of U(1) and SU(2), but rather $U(2) = [U(1) \times SU(2)]/\mathbb{Z}_2$, where \mathbb{Z}_2 is the center of SU(2).

Specifically, let us first consider the following "illegal" gauge transformations of the U(1) gauge theory on T^2 :

$$\Lambda_1'(x_1, x_2) = e^{ix_1/2}, \quad \Lambda_2'(x_1, x_2) = e^{ix_2/2}.$$
(11.37)

Here, x_1, x_2 are periodic coordinates on the torus with $0 \le x_i \le 2\pi$, i = 1, 2. Since Λ'_i (i = 1, 2) changes its value from +1 to $-1 \in \mathbb{Z}_2$ as x_i changes from 0 to 2π , it is not a genuine gauge transformation, and hence acts nontrivially on the physical Hilbert space. If we let Ω'_1, Ω'_2 be the corresponding operators on the Hilbert space of U(1) Chern-Simons theory at level k' = 2k, then their action on the basis states $|p\rangle$ defined in (11.36) is given by

$$\Omega_1'|p\rangle = |p+k\rangle, \quad \Omega_2'|p\rangle = (-1)^p|p\rangle.$$
(11.38)

We can similarly define the "illegal" gauge transformations for the SU(2) theories:

$$\Lambda_i''(x_1, x_2) = \operatorname{diag}(e^{ix_i/2}, e^{-ix_i/2}), \quad i = 1, 2.$$
(11.39)

They also change their values from the identity to $-1 \in \mathbb{Z}_2$ as x_i change from 0 to 2π . The corresponding operators Ω''_i (i = 1, 2) act on the Hilbert space of SU(2) Chern-Simons theory at level k'' > 0 by

$$\Omega_1''|\mathfrak{m}\rangle = |k'' - \mathfrak{m}\rangle, \quad \Omega_2''|\mathfrak{m}\rangle = (-1)^{\mathfrak{m}}|\mathfrak{m}\rangle.$$
(11.40)

In both U(1) and SU(2) theories, the action of Ω'_2 and Ω''_2 is easy to understand from the definition of the basis states $|p\rangle$ and $|\mathfrak{m}\rangle$, and then the action of Ω'_1 and Ω''_1 can be inferred from the modular transformation properties of the basis states.

We can now consider the U(2) gauge theory on T^2 and perform the transformations Λ'_i and Λ''_i simultaneously. The point is that while they are "illegal" gauge transformations when applied separately, they together become a genuine U(2) gauge transformation, as can be seen explicitly from the above expressions. Therefore, the Hilbert space $U(2)_{k',k''}$ is the subspace of the tensor product of the Hilbert spaces of $U(1)_{k'}$ and $SU(2)_{k''}$ theories on which the operators $\Omega'_i \otimes \Omega''_i$ act trivially. We can then read off the action of the Wilson loop operators on this subspace from those of the $U(1)_{k'}$ and $SU(2)_{k''}$ theories. The results for the cases listed in Table 11.4 are as follows. (In the following, we consider only the action of $W = W^{(1,0)}$, the Wilson loop going around the *a*-cycle once, but $W^{(0,1)}$ is related to W by modular transformation.)

• k = 1: For $U(2)_{2,1}$, the invariant subspace is one-dimensional, spanned by

$$|0\rangle_{U(1)} \otimes |0\rangle_{SU(2)} + |1\rangle_{U(1)} \otimes |1\rangle_{SU(2)}$$
.

The Wilson loop operator is just the identity: $\mathcal{W} = 1$.

For $U(2)_{2,3}$, the invariant subspace is two-dimensional, spanned by

$$\begin{aligned} |0\rangle_{U(1)} \otimes |0\rangle_{SU(2)} + |1\rangle_{U(1)} \otimes |3\rangle_{SU(2)}, \\ |0\rangle_{U(1)} \otimes |2\rangle_{SU(2)} + |1\rangle_{U(1)} \otimes |1\rangle_{SU(2)}. \end{aligned}$$

The Wilson loop operator is given in this basis by

$$\mathcal{W} = \operatorname{diag}(\phi, \phi - 1),$$

where $\phi = \frac{1}{2}(1 + \sqrt{5})$ is the "golden ratio." On the other hand, according to Table 11.2 the Tr-S results are:

$$\mathcal{W} = \operatorname{diag}(\pm\sqrt{2}, \pm\sqrt{2}, 0),$$

and they clearly don't agree with Chern-Simons results for $U(n)_{2,k''}$ for any k''. (We have only explicitly written down the cases k'' = 1,3 above, since they appear in Table 11.4, but it can be easily checked that other values don't give the right answer either.)

• k = 2: For $U(2)_{4,2}$, the invariant subspace is three-dimensional, spanned by

$$\begin{aligned} |0\rangle_{U(1)} \otimes |0\rangle_{SU(2)} + |2\rangle_{U(1)} \otimes |2\rangle_{SU(2)}, \\ |1\rangle_{U(1)} \otimes |1\rangle_{SU(2)} + |3\rangle_{U(1)} \otimes |1\rangle_{SU(2)}, \\ |0\rangle_{U(1)} \otimes |2\rangle_{SU(2)} + |2\rangle_{U(1)} \otimes |0\rangle_{SU(2)}. \end{aligned}$$

The Wilson loop operator in this basis is given by

$$\mathcal{W} = \operatorname{diag}(\sqrt{2}, 0, -\sqrt{2}),$$

which is also the result for $U(2)_{4,-2}$. Thus, the conjectures from Table 11.4 of $U(2)_{4,2}$ and $U(2)_{4,-2}$ for the sectors 2 = 1 + 1 and 2 = 2, respectively, are in precise agreement with the eigenvalues of \mathcal{W} that we calculated and summarized in Table 11.2.

• k = 3: For $U(2)_{6,1}$ the invariant subspace is three-dimensional, spanned by

$$\begin{aligned} |0\rangle_{U(1)} \otimes |0\rangle_{SU(2)} + |3\rangle_{U(1)} \otimes |1\rangle_{SU(2)}, \\ |1\rangle_{U(1)} \otimes |1\rangle_{SU(2)} + |4\rangle_{U(1)} \otimes |0\rangle_{SU(2)}, \\ |2\rangle_{U(1)} \otimes |0\rangle_{SU(2)} + |5\rangle_{U(1)} \otimes |1\rangle_{SU(2)}. \end{aligned}$$

The Wilson loop operator in this basis is given by

$$\mathcal{W} = \operatorname{diag}(1, -\omega, \omega^2),$$

where $\omega = e^{\pi i/3}$.

For $U(2)_{6,3}$ the invariant subspace is six-dimensional, spanned by

$$\begin{split} |0\rangle_{U(1)} \otimes |0\rangle_{SU(2)} + |3\rangle_{U(1)} \otimes |3\rangle_{SU(2)} ,\\ |0\rangle_{U(1)} \otimes |2\rangle_{SU(2)} + |3\rangle_{U(1)} \otimes |1\rangle_{SU(2)} ,\\ |1\rangle_{U(1)} \otimes |1\rangle_{SU(2)} + |4\rangle_{U(1)} \otimes |2\rangle_{SU(2)} ,\\ |1\rangle_{U(1)} \otimes |3\rangle_{SU(2)} + |4\rangle_{U(1)} \otimes |0\rangle_{SU(2)} ,\\ |2\rangle_{U(1)} \otimes |0\rangle_{SU(2)} + |5\rangle_{U(1)} \otimes |3\rangle_{SU(2)} ,\\ |2\rangle_{U(1)} \otimes |2\rangle_{SU(2)} + |5\rangle_{U(1)} \otimes |1\rangle_{SU(2)} .\end{split}$$

The Wilson loop operator in this basis is given by

$$\mathcal{W} = \operatorname{diag}(\phi, 1 - \phi, -\omega(1 - \phi), -\omega\phi, \omega^2\phi, \omega^2(1 - \phi)),$$

where ϕ is the golden ratio as before. The Tr-S eigenvalues that we expect have to have an absolute value of $\sqrt{2}$ or 0, and so we don't find an agreement in this case either.

In the above, we explicitly compared the eigenvalues of Wilson loop operators only for the gauge group U(2), but we can do similar computations for other gauge groups as well using the formula of [57]. We find that in general the results of Table 11.2 do not agree with the eigenvalues of Wilson loop operators in Chern-Simons theories with gauge group listed in Table 11.3, except for the k = 2, n = 2 case discussed above.

k	$n = n_1 + \dots + n_p$	Contribution to Witten Index
3	1 = 1	3
3	2 = 1 + 1	$\frac{9}{2}2^{m}$
3	2 = 2	$\frac{3}{2}2^{m}$
2	1 = 1	2
2	2 = 1 + 1	$2 \cdot 2^m$
2	2 = 2	$2 \cdot 2^m$
2	3 = 1 + 1 + 1	$\frac{4}{3} \cdot 3^m + 2$
2	3 = 2 + 1	$4 \cdot 3^m + 2$
2	3 = 3	$-\frac{2}{3}3^{m}$
1	1 = 1	1
1	2 = 1 + 1	$\frac{1}{2}2^{m}$
1	2 = 2	$\frac{3}{2}2^m$
1	3 = 1 + 1 + 1	$\frac{1}{6}3^m + \frac{1}{2}$
1	3 = 2 + 1	$\frac{3}{2}3^m + \frac{1}{2}$
1	3 = 3	$\frac{4}{3}3^{m}$
1	4 = 1 + 1 + 1 + 1	$\frac{1}{24}4^m + \frac{1}{4}2^m + \frac{1}{3}$
_1	4 = 2 + 1 + 1	$\frac{3}{4}4^m + 2^m$
_1	4 = 2 + 2	$\frac{9}{8}4^m + \frac{3}{4}2^m$
1	4 = 3 + 1	$\frac{4}{3}4^m + \frac{2}{3}$
_1	4 = 4	$\frac{3}{4}4^m$
1	5 = 1 + 1 + 1 + 1 + 1	$\frac{\frac{1}{120}5^m + \frac{1}{12}3^m + \frac{1}{6}2^m + \frac{3}{8}}{12}$
1	5 = 2 + 1 + 1 + 1	$\frac{\frac{1}{4}5^m + \frac{5}{6}3^m + \frac{1}{2}2^m + \frac{1}{4}}{2}$
1	5 = 2 + 2 + 1	$\frac{\frac{9}{8}5^m + \frac{3}{4}3^m + \frac{9}{8}}{2}$
1	5 = 3 + 1 + 1	$\frac{4}{3}5^m + \frac{4}{3}3^m + \frac{1}{3}2^m$
1	5 = 3 + 2	$2 \cdot 5^m + \frac{2}{3}3^m + 2^m$
1	5 = 4 + 1	$\frac{35m}{4}$
_1	5 = 5	$\frac{1}{5}5^{m}$

Table 11.1: The contribution to the Witten Index of the sector with given k and n, and a permutation σ in the conjugacy class that corresponds to the partition $n = n_1 + \cdots + n_p$.

k	n	Witten Index	$\dim \mathcal{H}(k, n, m = 0)$	$\mathcal{W}^{\dagger}\mathcal{W}$ eigenvalues
3	1	3	3	$1_{(3)}$
3	2	$6 \cdot 2^m$	9	$2_{(6)}, 0_{(3)}$
2	1	2	2	$1_{(2)}$
2	2	$4 \cdot 2^m$	6	$2_{(4)}, 0_{(2)}$
2	3	$6 \cdot 3^m + 4$	12	$3_{(6)}, 1_{(4)}, 0_{(2)}$
1	1	1	1	$1_{(1)}$
1	2	$2 \cdot 2^m$	3	$2_{(2)}, 0_{(1)}$
1	3	$3 \cdot 3^m + 1$	5	$3_{(3)}, 1_{(1)}, 0_{(1)}$
1	4	$4 \cdot 4^m + 2 \cdot 2^m + 1$	10	$4_{(4)}, 2_{(2)}, 1_{(1)}, 0_{(3)}$
1	5	$5 \cdot 5^m + 3 \cdot 3^m + 2 \cdot 2^m + 3$	15	$5_{(5)}, 3_{(3)}, 2_{(2)}, 1_{(3)}, 0_{(2)}$

Table 11.2: The Witten Index as a function of k, n, and m. The behavior of the Witten Index as a function of the number of quark and anti-quark pairs m allows us to calculate the eigenvalues λ_l of the operator $\mathcal{W}^{\dagger}\mathcal{W}$ and their multiplicities N_l . They are listed in the last column as $\lambda_{1(N_1)}, \lambda_{2(N_2)}, \ldots$

$v = \frac{\pi}{3}$	n = 1	$U(1)_1$
(k=1)	n=2	$U(2)_{2,1} \oplus U(2)_{2,-3}$
	n=3	$U(3)_{3,1} \oplus [U(1)_1 \times U(2)_{2,-3}] \oplus U(3)_{3,-2}$
	n=4	$U(4)_{4,1} \oplus 2[U(2)_{2,1} \times U(2)_{2,-3}] \oplus [U(1)_1 \times U(3)_{3,-2}] \oplus \mathcal{H}_{(2,2)}$
	n=5	$U(5)_{5,1} \oplus U(5)_{5,1} \oplus 2[U(3)_{3,1} \times U(2)_{2,-3}] \oplus [U(1)_1 \times \mathcal{H}_{(2,2)}] \oplus$
		$[U(2)_{2,1} \times U(3)_{3,-2}] \oplus [U(2)_{2,-3} \times U(3)_{3,-2}]$
$v = \frac{\pi}{2}$	n = 1	$U(1)_2$
(k=2)	n=2	$U(2)_{4,2} \oplus U(2)_{4,-2}$
	n = 3	$U(3)_{6,2} \oplus [U(1)_2 \times U(2)_{4,-2}] \oplus U(3)_{6,-1}$
$v = \frac{2\pi}{3}$	n = 1	$U(1)_3$
(k=3)	n=2	$U(2)_{6,3} \oplus U(2)_{6,-1}$

Table 11.3: The results of [1] regarding the equivalence of the Hilbert spaces of Tr-S and Chern-Simons theory on T^2 as representations of the mapping class group $SL(2,\mathbb{Z})$ together with \mathcal{U}, \mathcal{V} . The notation $U(n)_{k',k''}$ corresponds to a Chern-Simons theory where the U(1) part is at level k' and the SU(n) part is at level k''. One of the sectors (4 = 2 + 2) for n = 4 and k = 1 could not be matched with a Chern-Simons theory and is therefore written explicitly as $\mathcal{H}_{(2,2)}$. It also appears in the 5 = 2 + 2 + 1 decomposition of the n = 5 theory.

k	$n = n_1 + \dots + n_p$	Hilbert space
1	2 = 1 + 1	$U(2)_{2,1}$
1	2 = 2	$U(2)_{2,-3}$
2	2 = 1 + 1	$U(2)_{4,2}$
2	2 = 2	$U(2)_{4,-2}$
3	2 = 1 + 1	$U(2)_{6,3}$
3	2 = 2	$U(2)_{6,-1}$

Table 11.4: The n = 2 results of [1], sector by sector. Each Hilbert space of a Tr-S sector is equivalent, as a representation of the mapping class group $SL(2,\mathbb{Z})$ and \mathcal{U}, \mathcal{V} , to a corresponding Hilbert space of Chern-Simons theory. The notation $U(2)_{k',k''}$ corresponds to a Chern-Simons theory where the U(1) part is at level k' and the SU(2) part is at level k''.

Chapter 12 Discussion

We have computed the Witten Index of Tr-S theory on T^2 with charges, and we have used the results to calculate the eigenvalues of simple Wilson loop operators in the theory. The Witten Index of the U(n) theory with parameter k and 2m charges is listed in Table 11.2. We found that for gauge group U(2) and for the k = 2 case (the basic S-duality twist corresponding to $\tau \to -1/\tau$) the results are consistent with a conjecture put forward in [1] relating the theory to two U(2) Chern-Simons theories with $U(1) \subset U(2)$ at level 4 and the SU(2) at levels ± 2 . This would imply that the low-energy theory has two "superselection" sectors. On the other hand, we saw that in most other cases of n and k, the simple decomposition into superselection sectors labeled by a conjugacy class in the permutation group S_n (as conjectured in [1]) is inconsistent with the form of the Witten Index results, and several sectors have to be combined together to yield a consistent theory. What this theory is we do not know, but we were able to show that it is inconsistent with pure Chern-Simons theory, at least at low levels.

A physical perspective for understanding the discrepancy is plausibly as follows. Well-defined Wilson loops in the four-dimensional $\mathcal{N} = 4$ SYM theory with twisted boundary conditions flow, in the low-energy limit, to a S-duality invariant operator \mathcal{W}_{inv} that is a linear combination of Wilson loops and dual monopole operators. For example, as first briefly discussed in Section 6.6 of [1], for k = 2, the relevant operator supported on any curve C flows as¹

$$\mathcal{W}_{\text{inv.}}(C, x_3) \longrightarrow \mathcal{W}(C, x_3) + \mathcal{M}(C, x_3) + \mathcal{W}(C, x_3)^{\dagger} + \mathcal{M}(C, x_3)^{\dagger}$$

where \mathcal{M} is the dual 't Hooft operator. It is thus possible that Wilson loops in Tr-S theory correspond to an appropriate dimensional reduction of $\mathcal{W}_{inv.}$. As a simple check, we note that for the abelian case, computing $tr[(\mathcal{W}_{inv.}^{\dagger}\mathcal{W}_{inv.})^m]$ yields also the index to be k. Moreover, for the only nonabelian case which agrees with Chern-Simons theory, namely k = n = 2, the expectation values of monopole operators

¹For other values of k, \mathcal{W}_{inv} involves mixed Wilson-'t Hooft operators. See, for example, [58] for an illuminating discussion.
and Wilson loops are identical as first explained in [59]. To phrase it simply, the discrepancy between Tr-S and Chern-Simons theory may be understood physically as coming from non-trivial electromagnetic boundary conditions that descend from the four-dimensional twisted theory.

We conjecture that Tr-S is a topological theory for $n < \mathbf{r}$, and we presented some arguments in favor of this in §8.4. Another possible test of this could be to look for BPS states that carry nonzero momentum along T^2 . If the low-energy theory is topological we would expect to find only states with energies of the order of 1/R. In the type-IIA dual the momentum quantum numbers become D0-brane and D2-brane charge (where the D2 branes wrap directions 1, 10). It would be interesting to study the bound states of D0-branes with the *n* fundamental strings. In the limit $R \to 0$ this system can be mapped to a sector of a U(1) dipole-theory [13, 14]. We hope to explore this further in a separate work.

To calculate the Witten Index we divided the Hilbert space into "sectors" according to the pattern of closed and open strings of the dual type-IIA system. We saw that only a subset of sectors contributes to the Witten Index — the "decongested" sectors. The remaining (congested) sectors have fermionic zero modes, and they do not contribute to the Index. It would be interesting to explore these sectors further. For example, we noted that the supersymmetric system of charges has a global U(1) symmetry that is generated by the element \mathcal{J} of rotations in transverse directions that acts on spinors as $\frac{i}{2}(\Gamma^{45} + \Gamma^{67})$. Since \mathcal{J} commutes with all the surviving supersymmetry generators it is possible to generalize the Index to

$$I(u) \equiv \operatorname{tr}\left[(-)^F e^{iu\mathcal{J}}\right].$$

This modified index receives contributions only from ground states, but can get contributions from some congested sectors as well. At the end of §11.2 we gave an example of a congested sector with 2 complex fermionic zero modes that are all charged under \mathcal{J} . Quantizing these gives a Hilbert space with 4 states with \mathcal{J} charges -1, 0, 0, +1 and which contributes a term proportional to $(2 - 2\cos u)$ to the index. It is possible, however, that the fermionic zero modes interact with the bosonic modes and calculating I(u) therefore requires a separate treatment and will not be pursued here.

Taking a different approach, it would be interesting to construct Tr-S directly in terms of the duality-generating theories T(U(n)) defined in [41]. For example, Gaiotto and Witten argued that S-duality for SU(2) is generated by starting with T(SU(2)), which they identified with the strongly-coupled low-energy limit of the $2 + 1D \mathcal{N} = 4$ theory of two equally charged hypermultiplets coupled to a U(1) vector-multiplet. This theory has a manifest $SU(2) \times U(1)$ global symmetry, but as conjectured in [43] and further explained in [60], the low-energy limit has an enhanced $SU(2) \times SU(2)$ symmetry. The S-duality twist, according to Gaiotto and Witten, is then realized by gauging one SU(2) with the original gauge field (at $x_3 = 0$ in our context) and the other SU(2) with the dual gauge field (the one at $x_3 = 2\pi R$). It would be interesting to derive our results for the Witten Index directly from this construction. The computation is not so trivial, of course, because the T(SU(2)) theory is strongly coupled.

Recently, Terashima and Yamazaki [44] studied a related compactification with an S-duality twist but only $\mathcal{N} = 2$ supersymmetry in 2 + 1D. They computed the partition function of the theory on S^3 and related it to $SL(2, \mathbb{R})$ Chern-Simons theory. It would be interesting to understand if this construction can be modified to provide information on the $\mathcal{N} = 6$ setting that we studied in this paper.

In [1] another way to reproduce Tr-S from the T(SU(n)) theories was also offered. This made use of the low-energy limit of a D3-brane boundary on a (p,q) 5-brane, as constructed by Gaiotto and Witten using T(SU(n)) [41]. The starting point for [1] in this context was the (2, 0)-theory wrapping the three-dimensional submanifold of the space (8.5) that is defined by $\zeta_1 = \zeta_2 = \zeta_3 = 0$. Recently, a beautiful picture of the low-energy limit of the (2, 0)-theory compactified on a general three-dimensional manifold has emerged [45]-[61]. It would be interesting to analyze Tr-S from that perspective as well.

If Tr-S is topological then correlation functions of the low-energy limits of Wilson loops, discussed in §11.4, construct knot and link invariants. The general question, to which this paper provides only partial answers in special cases, is what are these invariants. Recently, there have been exciting new developments in the realization of knot invariants in terms of field theories and string theory (see [62] -[63] for a sample of the recent literature). A better understanding of Tr-S might provide new physical constructions of knot invariants. We hope to explore more general Wilson loops in future papers.

Chapter 13

Appendices

13.1 Supersymmetry and fermionic zero modes — details

In this appendix we expand on various statements made in §9.2, §9.9, and §10.3 about the amount of supersymmetry preserved by intersections of strings and branes and the fermionic modes that describe these systems at low-energy.

Our conventions are as follows. 10+1D directions are denoted by

$$I, J, K, \dots = 0, \dots, 10 \equiv \natural,$$

and we use $\natural \equiv 10$ in indices of Dirac matrices to avoid confusion with 1, 0. We work in Minkowski signature

$$\eta_{IJ}dx^Idx^J = -dx_0^2 + dx_1^2 + \dots + dx_{\natural}^2.$$

All our spinors, whether in 10+1D M-theory on 9+1D type-IIA/B are 32-component Majorana spinors on which the 11-dimensional Dirac matrices Γ^{I} can act. When we need type-IIA spinors, we will specify which direction is eliminated (as the "Mtheory direction"). For example, the 2^{nd} row of Table 8.1 is obtained from the 3^{rd} by eliminating direction 10, so the resulting type-IIA spinors ϵ are still 32-component Majorana spinors, but they can be decomposed into left-chirality and right-chirality spinors:

$$\epsilon = \epsilon_+ + \epsilon_-, \qquad \epsilon_{\pm} \equiv \frac{1}{2} (1 \pm \Gamma^{\natural}) \epsilon.$$

We will construct type-IIB spinors by performing T-duality on another direction. For example, the 1^{st} row of Table 8.1 is obtained from the 2^{rd} by T-duality on direction 1, so we can define the complex Weyl type-IIB SUSY parameters as

$$\epsilon_{\rm IIB} \equiv \epsilon_+ + i\Gamma^1 \epsilon_-$$

The Dirac matrices Γ^{I} are real and satisfy

$$\{\Gamma^{I}, \Gamma^{J}\} = 2\eta^{IJ}, \qquad \Gamma^{0123456789\natural} = 1.$$

Now take an M2-brane in directions $0, 9, \natural$. We denote

$$\mu, \nu, \dots = 0, 9, \natural; \qquad a, b, c, \dots = 1, \dots, 8.$$

The M2-brane low-energy fields are the scalars Φ^a (a = 1, ..., 8) and the spinors λ , which satisfy the chirality condition

$$\Gamma^{09\natural}\lambda = -\lambda\,.$$

Let ϵ be the 10+1D SUSY parameter and set

$$\epsilon_l \equiv \frac{1}{2}(1 - \Gamma^{09\natural})\epsilon, \qquad \epsilon_r \equiv \frac{1}{2}(1 + \Gamma^{09\natural})\epsilon$$

The SUSY transformations are

$$\delta\lambda = \epsilon_l + \partial_\mu \Phi^a \Gamma^\mu \Gamma_a \epsilon_r + (\overline{\lambda} \Gamma^\mu \epsilon_r) \partial_\mu \lambda \,, \qquad \delta\Phi^a = \overline{\lambda} \Gamma^a \epsilon_r + \partial_\mu \Phi^a \overline{\lambda} \Gamma^\mu \epsilon_r \,.$$

The spinors are real and $\overline{\lambda} \equiv \lambda^t \Gamma^0$. From the point of view of the 2 + 1*D* worldvolume theory, the ϵ_r parameters generate worldvolume supersymmetry transformations, while ϵ_l generate the κ -symmetry [64]. If we compactify this M2-brane on T^2 by making directions 1, 2 periodic, the spinors λ will have 16 linearly independent zero modes, which generate a multiplet of 256 states. These states are invariant under all supersymmetries with $\epsilon_l = 0$, but not invariant under supersymmetries with ϵ_r . As is customary, we refer to the supersymmetry transformation with $\epsilon_l = 0$ as the "unbroken supersymmetries."

Now consider an M2-brane stretched along directions 2, 3, which upon reduction to type-IIA on direction 2 will become an F1 in direction 3. At low-energy there are 8 scalar fields in the vector representation 8_v of the group SO(8) of rotations in transverse directions 1, 4, 5, 6, 7, 8, 9, 10, as well as their superpartners which are spinors in $(2, 8_s)$ of $SO(2, 1) \times SO(8)$. These spinors ψ satisfy

$$\psi = -\Gamma^{023}\psi = -\Gamma^{1456789\natural}\psi$$
.

Upon reduction to type-IIA we write

$$\psi = \psi_L + \psi_R$$

where

$$\psi_L = \frac{1}{2}(1+\Gamma^2)\psi, \qquad \psi_R = \frac{1}{2}(1-\Gamma^2)\psi,$$

satisfy

$$\Gamma^{03}\psi_L = \psi_L \,, \quad \Gamma^{03}\psi_R = -\psi_R \,.$$

100

which become left- and right-moving massless fields along the string. Note that in type-IIA both ψ_L and ψ_R are in 8_s of SO(8). Now, consider an M2-brane along directions 9, 10, which becomes a D2-brane in type-IIA, and compactify direction 10 as well. This M2-brane has low-energy fermions χ satisfying

$$\chi = -\Gamma^{09\natural}\chi = -\Gamma^{12345678}\chi \,.$$

Now compactify direction x_{10} . At low-energy, below the x_{10} compactification scale, the wrapped D2-brane looks like a string which has left- and right-moving massless fields $\chi_{L,R}$ along it. The are defined by:

$$\chi = \chi_R + \chi_L \,,$$

where

$$\chi_L = \frac{1}{2} (1 + \Gamma^{\natural}) \chi, \qquad \chi_R = \frac{1}{2} (1 - \Gamma^{\natural}) \chi, \qquad \Gamma^{09} \chi_L = \chi_L, \qquad \Gamma^{09} \chi_R = -\chi_R.$$

Next, we consider a configuration where such a D2-brane has two F1 strings (with the same orientation) emanating from a point on it: one string in the positive x_3 direction and one string in the negative x_3 direction. Note that the total charge at the point of origin is zero, since the charge of the endpoint of one string cancels the charge of the endpoint of the other string. Denote the low-energy fields on the string in the positive x_3 -direction by $\psi^{(>)}$, and denote the low-energy fields on the string in the negative x_3 -direction by $\psi^{(<)}$. Similarly, the wrapped D2-brane, at energies below the x_{10} compactification scale, has low-energy fields $\chi^{(<)}$ for the $x_9 < 0$ side and $\chi^{(>)}$ for the $x_9 > 0$ side. We are interested in the boundary conditions that connect the values of the 4 fields $\psi^{(<,>)}, \chi^{(<,>)}$ at the intersection point.

If we lift this system back to M-theory we get two M2-branes that intersect at a point. Perhaps the easiest way to derive the requisite boundary conditions is to deform this system to a smooth M2-brane that extends along a surface that, in appropriate complex coordinates described below, is a holomorphic curve. The lowenergy reduction, below the x_2, x_{10} compactification scales, looks like a (p, q)-web as in figure Figure 9.5(b) (see [65, 66] for some examples). The smooth geometry can be described by techniques similar to those developed in [53]. We define two complex coordinates

$$u \equiv e^{\frac{ix_2+x_3}{L_2}}, \qquad v \equiv e^{\frac{ix_{10}+x_9}{L_{10}}}$$

where L_2, L_{10} are the radii of directions 2 and 10. The smooth holomorphic curve is given by

$$(u-1)(v-1) = C \tag{13.1}$$

where $C \neq 0$ is a constant. Note that this is a deformation of the singular curve (u-1)(v-1) = 0. An M2-brane that wraps this holomorphic curve will have a lowenergy fermionic field λ on it. We are looking for zero modes of this field which have a finite limit at the 4 semi-infinite directions $x_3 \to \pm \infty$ and $x_9 \to \pm \infty$. Below, we explain how to make the connections:

$$\lambda(x_9 = -\infty) \to \chi^{(<)}, \qquad \lambda(x_9 = \infty) \to \chi^{(>)}, \lambda(x_3 = -\infty) \to \psi^{(<)}, \qquad \lambda(x_3 = \infty) \to \psi^{(>)}.$$
(13.2)

The linear algebraic relations among these four limit values will constitute the requisite boundary conditions.

Consider a part of the M2-brane near $x_3 \to \infty$. It is approximately at constant x_9, x_{10} and stretches in directions x_2, x_3 . The spinor can be decomposed according to the eigenvalue of Γ^{23} as

$$\lambda = \eta_{+} + \eta_{-}, \qquad \eta_{\pm} \equiv \frac{1}{2} (1 \pm i \Gamma^{23}) \lambda, \qquad \Gamma^{23} \eta_{\pm} = \mp i \eta_{\pm}, \qquad (\text{near } x_{3} \to \infty).$$
(13.3)

We then calculate the zero-mode equation

$$0 = (\Gamma^2 \partial_2 + \Gamma^3 \partial_3)\eta_{\pm} = \Gamma^2 (\partial_2 \mp i \partial_3)\eta_{\pm}$$

and so η_+ is holomorphic in $x_3 + ix_2$, and hence in u, while η_- is anti-holomorphic. When extending to the bulk of the holomorphic curve, we have to keep track of the tangent and normal bundles of the M2-brane surface given by (13.1). At an arbitrary point p on this surface the tangent plane T_p can be thought of as a subspace of \mathbb{R}^4 that is the constant tangent space in the x_2, x_3, x_9, x_{10} directions. As p varies the embedding $T_p \subset \mathbb{R}^4$ varies. Locally, we can pick a rotation $\Omega_p \in U(2) \subset \text{Spin}(4)$ that maps the tangent plane T_p to a common plane, which we choose to be the $x_2 - x_3$ plane, and also varies smoothly with p. At any fixed point p on the surface, this rotation Ω_p is unique up to $SO(2) \times SO(2)$ (rotations in the $x_2 - x_3$ and $x_9 - x_{10}$ planes separately). Near $x_9 \to \infty$, for example, T_p is the $x_9 - x_{10}$ plane and we can take the rotation in spinor representation to be

$$\Omega = e^{\frac{\pi}{4}(\Gamma^{2\natural} + \Gamma^{93})} = e^{\frac{\pi}{4}(1 + \Gamma^{239\natural})\Gamma^{93}} = \frac{1}{2}(1 + \Gamma^{2\natural})(1 + \Gamma^{93}).$$
(13.4)

If we decompose the fermionic field near $x_9 \to \infty$ as

$$\chi^{(>)} = \chi_R^{(>)} + \chi_L^{(>)}, \qquad \chi_R^{(>)} = \frac{1}{2} (1 + \Gamma^{239\natural}) \chi^{(>)}, \qquad \chi_L^{(>)} = \frac{1}{2} (1 - \Gamma^{239\natural}) \chi^{(>)}$$
(13.5)

the components $\chi_R^{(>)}$ and $\chi_L^{(>)}$, after rotation of the $x_9 - x_{10}$ plane into the $x_2 - x_3$ plane, are

$$\Omega\chi_R^{(>)} = \chi_R^{(>)}, \qquad \Omega\chi_L^{(>)} = \Gamma^{93}\chi_L^{(>)} = e^{\frac{\pi}{2}\Gamma^{93}}\chi_L^{(>)} = \Gamma^{2\natural}\chi_L^{(>)} = e^{\frac{\pi}{2}\Gamma^{2\natural}}\chi_L^{(>)}.$$
(13.6)

Thus, using Ω we can map chiral spinors at any point on the surface to a common space, and thus extend (13.3) by setting

$$\eta_{\pm} = \frac{1}{2} (1 \pm i\Gamma^{23})\Omega\lambda.$$
(13.7)

Let \mathcal{K} be the canonical bundle (i.e., the bundle whose sections are holomorphic (1, 0)forms on the curve), and let $\mathcal{N} = \mathcal{K}^{-1}$ be the normal bundle (where we embed the curve in \mathbb{C}^2 in directions 2,3,9,10). The modes η_+ transform as sections of $\mathcal{K}^{1/2} \otimes (\mathcal{N}^{1/2} \oplus \mathcal{N}^{-1/2}) = \mathcal{O} \oplus \mathcal{K}$, where \mathcal{O} is the trivial bundle. The relation (13.7) thus maps a spinor λ to a section of $\mathcal{O} \oplus \mathcal{K}$ (times a trivial spinor bundle in the transverse directions). So, altogether, we can decompose zero modes into

$$\lambda = \lambda_R + \lambda_L$$
, $\lambda_R \equiv \frac{1}{2} (1 + \Gamma^{239\natural}) \lambda$, $\lambda_L \equiv \frac{1}{2} (1 - \Gamma^{239\natural}) \lambda$

Then, zero modes λ_L are sections of $\mathcal{K}^{1/2} \otimes \mathcal{N}^{1/2} = \mathcal{O}$ which is the trivial bundle, while zero modes λ_R are sections of $\mathcal{K}^{1/2} \otimes \mathcal{N}^{-1/2} = \mathcal{K}$. Thus, λ_L is simply a holomorphic function of u, with finite limits at the 4 ends, while λ_R is a holomorphic 1-form with finite limits at the 4 ends.

In terms of the coordinate u, the curve (13.1) is mapped to the complex u-plane with 4 singular points: $u = 0, \infty$ correspond to the two ends of the F1-string, while u = 1 - C, 1 correspond to $v = 0, \infty$, which are the two ends of the D2-brane. Equation (13.2) becomes

$$\lambda(u = 1 - C) \to \chi^{(<)}, \qquad \lambda(u = 1) \to \chi^{(>)}, \qquad (13.8)$$
$$\lambda(u = 0) \to \psi^{(<)}, \qquad \lambda(u = \infty) \to \psi^{(>)}.$$

We denote

$$\chi_R^{(<,>)} \equiv \frac{1}{2} (1 + \Gamma^{239\natural}) \chi^{(<,>)}, \qquad \chi_L^{(<,>)} \equiv \frac{1}{2} (1 - \Gamma^{239\natural}) \chi^{(<,>)}.$$

and

$$\psi_R^{(<,>)} \equiv \frac{1}{2} (1 + \Gamma^{239\natural}) \psi^{(<,>)} , \qquad \psi_L^{(<,>)} \equiv \frac{1}{2} (1 - \Gamma^{239\natural}) \psi^{(<,>)} .$$

As λ_L modes are sections of the trivial line-bundle, and are therefore constant functions, their boundary conditions must be:

$$\chi_L^{(<)} = \chi_L^{(>)} = \psi_L^{(<)} = \psi_L^{(>)} .$$
(13.9)

On the other hand, the λ_R modes are sections of the canonical bundle. They correspond to holomorphic 1-forms which we denote by $\omega(u)du$. Being constant near $x_3 \to \infty$ means that ωdu is proportional at $u = \infty$ to $d \log u = du/u$, and so has a first-order zero there. Similar analysis of the behavior near the other three singular points u = 0, 1 - C, 1, shows that the 1-form needs to have at most a simple pole, and since it vanishes at $u = \infty$, the 1-form is of the form

$$\omega(u) = \left(\frac{\alpha}{u - (1 - C)} + \frac{\beta}{u - 1} + \frac{\gamma}{u}\right) du$$

Here $\chi_R^{(<)}$ is proportional to the constant α , $\chi_R^{(>)}$ is proportional to the constant β , $\psi_L^{(<)}$ is proportional to the constant γ , and $\psi_L^{(>)}$ is proportional to the constant $-(\alpha + \beta + \gamma)$. Converting back to spinors using (13.7), we find:

$$0 = \Gamma^{39} \chi_R^{(>)} - \Gamma^{39} \chi_R^{(<)} + \psi_R^{(>)} - \psi_R^{(<)} .$$
(13.10)

Equations (13.9)-(13.10) are the requisite boundary conditions!

103

13.2 Additional Combinatorics

13.2.1 The number of decongested binding matrices

In (11.29) we quoted the number f(n, m) of non-equivalent decongested binding matrices that have at least one nonzero entry in each of the *n* rows. We will now derive this expression. We can easily find a recursion formula for f(n, m) by noting that we can uniquely relabel the strings so that the m^{th} D2-brane is attached to the n^{th} string. Suppose there are $0 \leq l \leq m - n$ additional D2-branes attached to the n^{th} string, then the remaining (n-1) strings have f(n-1, m-l-1) configurations, and therefore

$$f(n,m) = \sum_{l=0}^{m-n} \binom{m-1}{l} f(n-1,m-l-1) = \sum_{j=n-1}^{m-1} \binom{m-1}{j} f(n-1,j). \quad (13.11)$$

Define the generating function

$$f_n(u) \equiv \sum_{m=n}^{\infty} f(n,m) u^{-m}$$
.

Then, (13.11) implies

$$f_n(u) = \sum_{m=n}^{\infty} \sum_{j=n-1}^{m-1} {\binom{m-1}{j}} f(n-1,j) u^{-m}$$
$$= \sum_{j=n-1}^{\infty} \sum_{m=j+1}^{\infty} {\binom{m-1}{j}} f(n-1,j) u^{-m}$$
$$= \sum_{j=n-1}^{\infty} (u-1)^{-(j+1)} f(n-1,j)$$
$$= \frac{1}{u-1} f_{n-1}(u-1) .$$

It follows that

$$f_n(u) = \frac{1}{\prod_{j=1}^n (u-j)} = \sum_{j=1}^n \frac{(-1)^{n-j}}{(j-1)!(n-j)!(u-j)},$$

and therefore

$$f(n,m) = \sum_{j=1}^{n} \frac{(-1)^{n-j}}{j!(n-j)!} j^{m}.$$

13.2.2 A generating function for the number of sectors and Fibonacci numbers

In the following, we will count the number of sectors for all $m \ge 1$. Remarkably, it turns out that the number of sectors for consecutive m's follows a generalized Fibonacci sequence = $\{3, 7, 18, 47, 123 \dots\}$. To solve this combinatorial problem, we begin by introducing another set of notations to describe the binding matrices \mathfrak{B} . We denote any continuous stretch of rows of \mathfrak{B} using letters [j] defined as:

$$\begin{pmatrix} 1\\0 \end{pmatrix} \equiv [1], \quad \begin{pmatrix} 1&1\\1&0 \end{pmatrix} \equiv [2], \dots \quad \underbrace{\begin{pmatrix} 1&1&1&\dots&1\\1&1&1&\dots&0 \end{pmatrix}}_{j \text{ rows}} \equiv [j] \tag{13.12}$$

In general we cannot recover \mathfrak{B} from the word, but note that different \mathfrak{B} 's can be made equivalent after relabeling of the open strings. In particular, whenever there is a column with two 1's, we can relabel the open strings that are to the right of that column. This will change the matrix \mathfrak{B} , and potentially also the accompanying permutation σ , but will give an equivalent physical sector. For example, if we relabel all strings after the 5th column of (13.13) we get [in the notation of (11.1) of §11.1]:

This means that up to changing σ we can always assume that after a string of $\begin{pmatrix} 1\\1 \end{pmatrix}$ columns there appears $\begin{pmatrix} 1\\0 \end{pmatrix}$. The letters [2], [3], ... thus translate back to a unique sequence of columns in \mathfrak{B} . But the letter [1] can translate back to either $\begin{pmatrix} 1\\0 \end{pmatrix}$ or $\begin{pmatrix} 0\\1 \end{pmatrix}$.

We denote the number of [1] letters in a given word by p. Then, there are 2^p ways to translate the word back to \mathfrak{B} . There are also two possibilities for σ , which gives 2^{p+1} possibilities, but now each sector is counted twice because we can exchange the entire two rows of \mathfrak{B} to get equivalent sectors. Altogether, we find that a word with p letters [1] corresponds to 2^p sectors.

We can now count the number of sectors for a generic m > 1 as follows. Let us consider words that have p letters [1] and $r \ge 0$ other letters, which makes (r + p)letters in total. The total number of different ways to fill in the [1]'s is then $\frac{(r+p)!}{r!p!}$. We need to compute also the total number of ways to write (m - p) as a sum of rnumbers from 2, 3, 4, This is calculated by subtracting 2 from each letter and then computing the number of ways to write (m - p - 2r) in this form as a sum of r non-negative integers, which is simply equal to $\frac{(m-p-r-1)!}{(m-p-2r)!(r-1)!}$. Putting these facts together, the total number of different configurations d_m represented by this class of words is then

$$d_m = \sum_{r,p} \frac{(r+p)!(m-p-r-1)!}{(m-p-2r)!(r-1)!r!p!} 2^p$$
(13.15)

Finally, we note that the set of \mathfrak{B} 's that can be represented by a word lacks those which end with a $\begin{pmatrix} 1 \\ 1 \end{pmatrix}$ column, i.e., $\mathfrak{B}_{1m} = \mathfrak{B}_{2m} = 1$. Including such configurations of which there are d_{m-1} , the total number of different sectors D_m is then found to be

$$D_m = d_m + d_{m-1} \,. \tag{13.16}$$

To find a closed form for D_m , we can sum over m, r, p to write down a rational generating function whose Taylor coefficients will yield D_m . It is convenient to consider first

$$G(t) = \sum_{m=0}^{\infty} d_m t^m = \sum_{r,p,m} \frac{(r+p)!(m-p-r-1)!}{(m-p-2r)!(r-1)!r!p!} 2^p t^m$$

=
$$\sum_{r,p,m} \frac{(r+p)!(m-p-r-1)!}{(m-p-2r)!(r-1)!r!p!} 2^p t^{m-p-2r} t^{p+2r} = \sum_{r,p} \frac{(r+p)!}{r!p!} 2^p (1-t)^{-r} t^{p+2r}$$

=
$$\sum_p (2t)^p \left(1 - \frac{t^2}{1-t}\right)^{-p-1} = \frac{1-t}{1-3t+t^2} .$$
 (13.17)

Note that $d_0 = 1$. Now, (13.16) and (13.17) allow us to construct the full generating function for D_m , defined for $m \ge 1$. This gives us

$$F(t) = \sum_{m=1}^{\infty} D_m t^m = \sum_{m=1}^{\infty} (d_m + d_{m-1}) t^m = \sum_{m=1}^{\infty} (d_m t^m) + t \sum_{m=0}^{\infty} d_m t^m$$

= $(1+t)G(t) - 1 = \frac{3t - 2t^2}{1 - 3t + t^2}$. (13.18)

We can thus compute D_m for all $m \ge 1$ easily from (13.18), and we get:

$$D_m = 3, 7, 18, 47, 123, \dots, \tag{13.19}$$

which is a sequence that is constructed by taking the even-numbered terms of a Fibonacci sequence L_n that starts with the first two seed values $L_0 = 2, L_1 = 1$. This sequence (Lucas numbers) is related to the standard Fibonacci sequence F_n , and we can write down an exact expression¹ for D_m :

$$D_m = L_{2m-1} + L_{2m-2} = F_{2m+1} + F_{2m-1} = \phi^{2m} + (1-\phi)^{2m}$$
(13.20)

where $\phi = (1 + \sqrt{5})/2$ is the Golden ratio.

Equation (13.20) can also be derived more directly. For this purpose, consider the family of sectors for (m-1) D2-branes. To enumerate the sectors for m D2-branes, we can add another column to the right² of \mathfrak{B} , i.e. any of $(1 \ 0)^{\top}$, $(0 \ 1)^{\top}$ and $(1 \ 1)^{\top}$. This gives a new set of \mathfrak{B} 's which includes all the sectors for m as a subset. Thus, we can write

$$D_m = 3D_{m-1} - \mathcal{O}_m \tag{13.21}$$

where \mathcal{O}_m counts the sectors that have been over-counted. It turns out that \mathcal{O}_m is exactly D_{m-2} .

To prove this, we observe that the family of sectors for m-1 D2-branes can always be divided into two classes: (i)those which are invariant under σ and (ii) those which are not. Class (i) matrices are bounded at both ends by at least one $(1 \ 1)^{\top}$, whereas for class (ii), no $(1 \ 1)^{\top}$ appears at either the left or right end. Now, for class (ii), when we add $(1 \ 1)^{\top}$ to the right end, the resulting \mathfrak{B} is now invariant under σ , and thus $3D_{m-1}$ over-counts by one for each distinct case. By removing the 1st column, each such over-counted matrix can be mapped to a matrix of class (i) with (m-2)D2-branes which end with either $(1 \ 0)^{\top}$ or $(0 \ 1)^{\top}$. Similarly, for class (i), consider each pair of terms generated by adding $(1 \ 0)^{\top}$ or $(0 \ 1)^{\top}$ at the right end. They can be easily seen to be equivalent, and thus each such matrix can be mapped to a matrix of class (i) with (m-2) D2-branes and which end with $(1 \ 1)^{\top}$. Taking into account the over-counting for both classes, we see that the total number of over-counts can be mapped to precisely D_{m-2} .

We conclude therefore that $\mathcal{O}_m = D_{m-2}$. This means that we have, from (13.21),

$$D_m = 3D_{m-1} - D_{m-2}. (13.22)$$

This is precisely the recurrence relation for the generalized Fibonacci sequence we have found in (13.20). Given D_1 and D_2 , we can generate the rest of the sequence.

Finally, we note that in the case of k = 2, the number of states in each sector is always 4. (This is not the case for k = 1, 3 as can be seen from our earlier computations.) For any m, we can verify this straightforwardly using the methods discussed in this section. Below, we present a short inductive derivation for $\sigma = 1$ sectors.

¹One can also use a binomial-Fibonacci identity to write $D_m = \sum_{k=0}^{m} {}^{2m-k}C_k + \sum_{k=0}^{m-1} {}^{2(m-1)-k}C_k$. Invoking (13.16), this gives us a simple closed form for (13.15). ²We can also add another column to the left, but to avoid over-counting, one can choose to add

in only one direction.

To be definite, let us consider an arbitrary k = 2 sector in m + 1 which begins with $\mathfrak{B}_{i1} = (1\,1)^{\mathsf{T}}$. Such a sector can be thought of as an (m-1) sector augmented by two more columns of \mathfrak{B} as represented in Figure 13.1.



Figure 13.1: The above decomposes the $\sigma = 1$ class of sectors (with (m + 1) D2branes) in a particular way. The circles on the last two D2-branes indicate various possibilities for the last two columns of \mathfrak{B} , giving possibly different sectors.

There are five different possibilities for the last two columns of \mathfrak{B} that we need to consider, the rest being related by symmetries. Below, we assume that $\mathfrak{B}_{i1} = 1$ for $\mathbf{i} = 1, 2$. Also, we denote the action that corresponds to the configuration before adding the last column by I_m , and the resulting action to be I_{m+1} . After some algebra, we simplify the various actions to be, in each case,

1. $\mathfrak{B}_{\mathfrak{i}m} = (1\,1)^{\top}, \mathfrak{B}_{\mathfrak{i}(m+1)} = (1\,1)^{\top} :$ $I_{m+1} = I_m + \int (\mathfrak{p}^{m+1} - \mathfrak{p}^m) \left(2d\mathfrak{q}^{1m} - d\mathfrak{q}^{1(m+1)} - dq^{2m}\right);$ 2. $\mathfrak{B}_{m-1} = (1\,1)^{\top}, \mathfrak{B}_{m-1} = (1\,0)^{\top}:$

2.
$$\mathcal{B}_{im} = (11)^{+}, \mathcal{B}_{i(m+1)} = (10)^{+};$$

 $I_{m+1} = I_m + \int (\mathfrak{p}^{m+1} - \mathfrak{p}^m) \left(d\mathfrak{q}^{1m} - d\mathfrak{q}^{1(m+1)} \right);$

3.
$$\mathfrak{B}_{\mathfrak{i}m} = (1\,0)^{\top}, \mathfrak{B}_{\mathfrak{i}(m+1)} = (1\,0)^{\top}:$$

 $I_{m+1} = I_m + \int (\mathfrak{p}^{m+1} - \mathfrak{p}^m) \left(d\mathfrak{q}^{1m} - d\mathfrak{q}^{1(m+1)} \right);$

4. $\mathfrak{B}_{im} = (1 \ 0)^{\top}, \mathfrak{B}_{i(m+1)} = (0 \ 1)^{\top}$: $I_{m+1} = I_m + \int (\mathfrak{p}^{m+1} - \mathfrak{p}^l) (d\mathfrak{q}^{1m} - d\mathfrak{q}^{1(m+1)}),$ where \mathfrak{B}_{2l} is the second rightmost column of $\mathfrak{B}_{2\mathfrak{a}}$ that is unity;

5.
$$\mathfrak{B}_{\mathfrak{i}m} = (1\,0)^{\top}, \mathfrak{B}_{\mathfrak{i}(m+1)} = (1\,1)^{\top}:$$

 $I_{m+1} = I_m + \int \left(\mathfrak{p}^l - \mathfrak{p}^{m+1}\right) \left(d\mathfrak{q}^{2m} - d\mathfrak{q}^{1m}\right) + \left(\mathfrak{p}^m - \mathfrak{p}^{m+1}\right) \left(d\mathfrak{q}^{1(m+1)} - d\mathfrak{q}^{1m}\right)$

Thus, we see that in all of these cases, the new action (for m + 1 branes) differs from the previous one (for m branes) by a term dependent on new conjugate pairs of $(\mathfrak{p}, \mathfrak{q})$. This implies that the determinant, and hence the number of states remains the same. Assuming other choices of \mathfrak{B}_{i1} and $\sigma = -1$ leads to a similar conclusion too, but we will leave details to the interested reader. Essentially, only in the k = 2case, the number of states = 4 for all sectors with m = 2 branes. The calculation above, together with similar ones for other choices of \mathfrak{B}_{i1} and $\sigma = -1$, then implies that the number of states = $4 \forall m$ by induction.



Figure 13.2: External quark and anti-quark sources are realized as endpoints of fundamental strings. 2m D3-branes (m = 2 in the picture) control the (x_1, x_2) coordinates of the sources. (x_1, x_2) are along the direction of the n D3-branes.

13.3 An alternative set of boundary conditions using tilted D3-branes

At the end of §9.9 we mentioned an alternative possibly useful set of boundary conditions for the fields $X_{\mathfrak{c}}^{\mu}, \psi_{\mathfrak{c}}$ ($\mathfrak{c} = 1, \ldots, 2m$) at the $x_9 = \Delta$ or $x_9 = -\Delta$ end of the 2m open strings, which we will now describe. The boundary conditions that we used in the main text are formally realized by D5-branes. Here we will instead realize the boundary conditions by D3-branes. By tilting the D3-branes, these boundary conditions can be made to preserve one real supercharge. They also have the advantage that they can be realized more comfortably in string theory, avoiding the complications mentioned below (9.33). However, they suffer from additional fermionic zero modes which render the Witten Index identically zero. We discuss this construction below.

In this alternative set-up, we realize the 2m sources as endpoints of 2m strings that end on the *n* D3-branes. The coordinates of one end of the j^{th} string are thus given by

$$(x_0 = t, x_1 = a_1^{(j)}, x_2 = a_2^{(j)}, x_3 = x_4 = x_5 = x_6 = x_7 = x_8 = x_9 = 0).$$

We control the coordinates (x_1, x_2) of the endpoint of the j^{th} string by letting its other endpoint lie on another D3-brane whose (x_1, x_2) position is fixed. Thus, for each j =

Brane	type	(number)	1, 4	2,7	3	5, 6	8	9
Original	D3	(n)	\longrightarrow	\longrightarrow	÷			
String	F1	(2m)						Η
additional	D3	(2m)	\nearrow	\nearrow			_	

Table 13.1: Open strings end on the original n D3 branes and additional 2m D3branes. \nearrow denotes a brane that extends along the diagonal of the corresponding plane (such as $x_1 - x_4 = \text{const.}$) and \longrightarrow denotes a brane that extends along the first direction ($x_4 = \text{const.}$).

 $1, \ldots, 2m$ we introduce a D3-brane which controls the j^{th} source (see Figure 13.2).³. We take the 2m D3-branes to be parallel to each other, and let the \mathfrak{c}^{th} one occupy the locus

$$x_1 - x_4 = a_1^{(j)}, \quad x_2 - x_7 = a_2^{(j)}, \quad x_3 = 0, \quad x_9 = \Delta_j, \quad x_5 = x_6 = 0.$$
 (13.23)

We will assume that

$$\Delta_{j+m} = -\Delta_j < 0, \qquad j = 1, \dots, m,$$

so that a D3-brane that controls a quark $(j \leq m)$ is at a positive x_9 and a D3-brane that controls an anti-quark (j > m) is at a negative x_9 , and all strings have a nonzero mass. We will also set $\Delta_1 = \cdots = \Delta_m = \Delta$ for simplicity. As we argued below (9.10), Δ will not affect the low-energy description, and in fact the mass of the string is an irrelevant operator in the IR.

Equation (13.23) describes a D3-brane that extends along the x_8 direction and along the diagonals of the $x_1 - x_4$ and $x_2 - x_7$ planes. The directions of the D-branes are summarized in Table 13.1. They are designed so that the combined system of original *n* D3-branes and the additional 2m D3-branes preserves some amount of supersymmetry. More precisely, we find 4 unbroken supersymmetries that are preserved by this combined system [68, 69, 70, 71]. Including the fundamental string and the S-R-twist we find that there is only one unbroken real supercharge left. Another way of saying this is that out of the original 12 supercharges that are preserved by the *n* D3-branes and the twist, 11 are broken by the 2m branes and the fundamental strings.

Recall that the original n D3-branes extend in directions x_1, x_2, x_3 and occupy the locus

$$x_4 = x_5 = x_6 = x_7 = x_8 = x_9 = 0. (13.24)$$

³This is reminiscent of the way Wilson loops were calculated in the topological string realization of Chern-Simons theory by Ooguri and Vafa [67]. We are grateful to Kevin Schaeffer for pointing out to us the connection with that work.

Brane	1	3	4	5	6	7	8	9	10
F1		÷							
D2								\vdash	_
NS5	—		_			—	—		—

Table 13.2: In the type IIA dual of Table 13.1, the D2-branes end on NS5-branes. Appropriate low-energy background fields on the NS5-brane worldvolumes control the position of the Wilson loop in the type-IIB picture.

Therefore, an open string with one endpoint on the original D3-branes and the other endpoint on one of the 2m D3-branes will have minimal length (of Δ) if and only if all its coordinates except x_9 are constant:

$$x_1 = a_1^{(c)}, \quad x_2 = a_2^{(c)}, \quad x_3 = x_4 = x_5 = x_6 = x_7 = x_8 = 0.$$

The positions of the 2m D3-branes therefore control the positions of the 2m quarks and anti-quarks.

Now we transform the system to type-IIA by applying the U-duality transformation described in Table 8.1. After the series of dualities of Table 8.1 the 2m D3-branes turn into type-IIA NS5-branes that wrap directions $x_1, x_4, x_7, x_8, x_{10}$. The parameters $(a_1^{(c)}, a_2^{(c)})$ that enter into the conditions (13.23) are encoded in the compact scalar Φ and 2-form *B* that are part of the low-energy tensor multiplet of the NS5-brane. We have

$$B = (x_4 + a_1^{(j)})dx_1 \wedge dx_{10} - x_7 dx_0 \wedge dx_8, \qquad \Phi = x_7 + a_2^{(j)}. \tag{13.25}$$

The type-IIA system is described in Table 13.2.

Similarly to the set-up in the main text, we have to connect each D2-brane that corresponds to a quark with a D2-brane that corresponds to an anti-quark and glue them into a smooth D2-brane that ends on one NS5-brane at $x_9 = \Delta$ (j = 1, ..., m) and another NS5-brane at $x_9 = -\Delta$.

For the specific purpose of computing the Witten Index however, this configuration is not so useful because, in an analogous computation as was done in §10.3, we found that there are fermionic zero modes that will make the contribution to the Witten Index vanish. Nonetheless, we also found that this configuration preserves one real supercharge, and thus it may turn out to be useful in understanding other aspects of the problem.

Part IV

Abelian Self-duality and Gravitational Chern-Simons Theory

Chapter 14 Motivation

There exists an intriguing connection between three-dimensional Chern-Simons theory and S-duality of four-dimensional $\mathcal{N} = 4$ super Yang-Mills (SYM) gauge theory. For abelian gauge groups the connection is simple and can be stated as follows [1]. Compactification of SYM on a circle with boundary conditions that are twisted by an S-duality accompanied with an R-symmetry twist in order to eliminate zero modes and preserve supersymmetry, leads to a 2 + 1D low-energy theory that is pure Chern-Simons theory at a level k determined by the complex coupling of SYM given by

$$\tau \equiv \frac{4\pi i}{g_{\rm YM}^2} + \frac{\theta}{2\pi} \,,$$

with θ and $g_{\rm YM}$ being a theta angle and gauge coupling constant.

In particular, the self-dual coupling constant $\tau = i$ is invariant under $S: \tau \to -1/\tau$ and leads to Chern-Simons level k = 2. In [1] the compactification of SYM was on a flat three-dimensional manifold $(T^2 \times \mathbb{R})$. Since the resulting low-energy theory is topological, it is independent of the metric. But this assertion is not completely obvious from the outset. In particular, the symmetries of the problem appear to allow a gravitational Chern-Simons term, which is defined by analogy with the Yang-Mills Chern-Simons term in the familiar form

$$S_{GCS} = \frac{1}{4\pi} \int_{M_3} (\omega \wedge d\omega + \frac{2}{3}\omega \wedge \omega \wedge \omega)$$

where ω is the Levi-Civita connection one-form on the spin bundle of M_3 . This term can be rewritten on a four-manifold M_4 with boundary $\partial M_4 = M_3$ like

$$S_{GCS} = \frac{1}{2\pi} \int_{M_4} \operatorname{tr}(R \wedge R)$$

as in §15.1 and must be studied on M_4 with non-vanishing curvature $R = d\omega + \omega \wedge \omega$.

In fact, we can expect that adding the gravitational Chern-Simons term is required for regularizing this theory [30]. Chern-Simons partition function, at least for large k, can be defined as a topological invariant of the oriented, framed three manifold M_3 (a framed three manifold being one that is endowed with a homotopy class of trivializations of the tangent bundle). This can be done only after adding the gravitational Chern-Simons term which is necessary for anomaly cancellation. However, the Lorentz Chern-Simons term is of a higher-order in derivatives, so only the Yang-Mills Chern-Simons term is a part of the low-energy effective supergravity theory, as was found in the study of the ground states [1].

The U(1) gauge theory on a four-manifold M_4 has S-duality symmetry which is an electric-magnetic duality that inverts the gauge coupling constant and extends to an action of full $SL(2,\mathbb{Z})$ on the complex coupling. However, the partition function of abelian SYM theory is not a modular-invariant function under this transformation. It instead transforms as a modular form, with weights that depend on the Euler characteristic, χ , and the signature, σ , of the four-manifold which can be written as

$$\chi = \frac{1}{2\pi} \int_{M_4} \operatorname{tr}(R \wedge *R), \qquad \sigma = \frac{1}{2\pi} \int_{M_4} \operatorname{tr}(R \wedge R)$$

in terms of the curvature two-form [31]. Thus the partition function includes the gravitational Chern-Simons term contribution.

Indeed, we do find such a contribution to be, at least to the lowest order in the deviation from flat four-dimensional space, in agreement with the S-duality of the abelian gauge theory at the self-dual point. We realize the S-duality geometrically starting with the six-dimensional description of the worldvolume of an M5-brane in M-theory, called (2, 0) theory, which gives $\mathcal{N} = 4$ SYM when compactified on a two-torus [38], and introduce a twist on this T^2 . The one-loop perturbative contribution in this theory is in accord with low-energy approximation of the gravitational Chern-Simons theory.

14.1 A Review of (2, 0) Theory

The wonderfully thing about supersymmetry is that it makes supersymmetric theories amenable to exact treatment allowing holomorphic quantities to be exactly computed. Like with any other symmetry, the more supersymmetry a theory has, the more constrained the field content and interactions are. The largest number of supercharges possible in free field theory is sixteen. With a greater number of supercharges than that, the free multiplet would include fields with spin larger than one which result in inconsistent theories without gravity.

The most symmetric classical field theory with sixteen supercharges is supersymmetric Yang-Mills (SYM) theory in ten dimensions. And the simplest such theory is an Abelian gauge theory. Its supermultiplet includes only a massless photon and a massless fermion. The nonabelian extension of this theory exists as a classical field theory, but its quantum version is anomalous and therefore inconsistent. We can construct theories with sixteen supercharges in less than ten dimensions by taking a toroidal compactification and retaining only the states of zero compact momentum. The theory in d dimensions, which is obtained by dimensional reduction of the abelian classical theory in ten dimensions, is anomaly free. Its Lorentz symmetry is Spin(d-1,1) while its R-symmetry group, Spin(10-d), originates from the rest of ten dimensional Lorentz group.

The most widely known theory with sixteen supercharges is the $\mathcal{N} = 4$ SYM theory in four dimensions. But in six dimensions, there is also a supersymmetry algebra with sixteen supercharges [72]. It includes four spinors of the same chirality and it is usually called the (2, 0) algebra and thus the corresponding class of theories are (2, 0) theories. Upon compactification on a two-torus down to four-dimensions these theories become $\mathcal{N} = 4$ SYM [38]. The complex coupling parameter τ of the four-dimensional U(1) gauge theory is simply the complex structure τ of the T^2 .

These theories appeared first in the study of K3 compactications of the Type IIB theory at an ADE singularity [38] and later in the context of coincidental 5-branes in M-theory [73, 74]. They are expected to be non-trivial fixed points of the renormalization group in six dimensions and thus not to have a dimensionful parameter. Furthermore, since these fixed points are isolated, they have no dimensionless parameter.

In six dimensions, (2, 0) supersymmetry is maximal for a non-gravitational theory, and there is a unique matter multiplet with this supersymmetry. The (2, 0)supersymmetry constrains the metric on the moduli space to be locally flat. Along this moduli space there is a tensor multiplets of (2, 0) supersymmetry which includes five scalars and a two-form B, whose field strength three form is selfdual

$$H \equiv dB = *dB. \tag{14.1}$$

Even without supersymmetry or fermions the theory needs a spin structure for its definition. With supersymmetry, the mater multiplet also contains four chiral fermions. This tensor multiplet reduces to a vector multiplet of four-dimensional $\mathcal{N} = 4$ supersymmetry when compactified on a two-torus.

The theory with a single matter multiplet can be realized as the low energy limit of the worldvolume theory of a 5-brane in M-theory. Since the only parameter of the 5-brane theory is the eleven-dimensional Planck scale, we again see that, in this limit, there are no adjustable parameters, dimensionful or dimensionless. The presence of a self-dual field makes a covariant Lagrangian description tricky [75], but a noncovariant Lagrangian can be written [76] and it is unique, with no free parameters. However, known results about the low-energy description of the theory will be sufficient for understanding our result.

This six-dimensional theory has string-like excitations of an unusual kind. From the point of view of the construction via type IIB theory at an ADE singularity the lightest scale in the theory is set by the tension of the strings. However, (2, 0)theories are scale-invariant just as are the four-dimensional $\mathcal{N} = 4$ gauge theories. And thus the strings in a (2, 0) theories are entirely different from other strings we have studied: they live in six dimensions, they are not associated with gravity, and they have no adjustable coupling constant — their interactions in fact are of order one. This is why (2, 0) theories are also called tensionless string theories. Of all the new phases of gauge and string theories that have been discovered this is perhaps the most mysterious, and may be a key to understanding many other things.

14.2 (2, 0) Theory on a Twisted Geometry

We are interested in studying S-duality of the abelian $\mathcal{N} = 4$ SYM theory in four dimensions, which can be realized as a geometrical twist of the six-dimensional (2, 0) theory [38]. This section describes the construction of this theory by twisting of the worldvolume of the coincident M5-branes while applying appropriate boundary conditions to the associated fields. This may well be the proper setting for understanding the S-duality of $\mathcal{N} = 4$ gauge theory.

Let the coordinates on the six-dimensional Minkowski manifold of the (2, 0) theory be x_{μ} with $\mu = 0, \ldots, 5$ while we will use x_a with $a = 0, \ldots, 3$ for coordinates on M_4 . Let us compactify this six-dimensional manifold to $M_4 \times T^2$ with the coordinates on the two-torus, x_4 and x_5 , given in terms of complex $z = x_4 + ix_5 \in \mathbb{C}$. We regard the T^2 with structure constant τ as this complex plane \mathbb{C} modded out by the lattice, $\mathbb{C}/(\mathbb{Z} + \tau\mathbb{Z})$, so that $z \sim z + 1 \sim z + \tau$. After shrinking this torus to zero size the leftover gauge theory on M_4 has manifest S-duality coming from the diffeomorphisms of the T^2 . This is similar to type IIB $SL(2,\mathbb{Z})$ strong-weak coupling duality viewed as a lower dimensional manifestation of M-theory diffeomorphism invariance [77].

Indeed, $\mathcal{N} = 4 \ U(1)$ super Yang-Mills theory with coupling constant τ is the low-energy limit of six-dimensional (2, 0) theory compactified on a T^2 , with τ being the complex structure parameter of the torus, where S-duality can be realized as the $SL(2,\mathbb{Z})$ transformation. If we denote a general $SL(2,\mathbb{Z})$ -element by

$$\mathbf{g} \equiv \begin{pmatrix} \mathbf{a} & \mathbf{b} \\ \mathbf{c} & \mathbf{d} \end{pmatrix}, \qquad \mathbf{g} : \tau \to \frac{\mathbf{a}\tau + \mathbf{b}}{\mathbf{c}\tau + \mathbf{d}}$$
 (14.2)

is the action of the S-duality transformation on the coupling τ where $\mathbf{ad} - \mathbf{bc} = 1$.

We set this complex coupling to the self-dual value $\tau = i$ and introduce S-duality twist into the compactification of the $\mathcal{N} = 4$ SYM on a circle by introducing an unusual boundary conditions on the fields along this S^1 , which we choose to be in the direction x_3 . This is achieved by inserting a strong-weak coupling transformation that realizes $\tau \to -1/\tau$ and corresponds to

$$\mathbf{g} = \begin{pmatrix} 0 & -1\\ 1 & 0 \end{pmatrix} \,. \tag{14.3}$$

This choice of **g** is permissible as it leaves the coupling $\tau = i$ invariant. Thinking of the circle as the segment $x_3 \in [0, 2\pi R]$ with endpoints 0 and $2\pi R$ identified, we require the field configuration at $2\pi R$ to be an S-dual of that at 0. This kind of boundary condition is what I have been referring to as an S-twist.

In particular, on the gauge field two-form B of the (2, 0) theory, the boundary conditions that results from this S-twist are

$$B_{az}(x_3 + 2\pi R) = iB_{az}(x_3), \qquad B_{a\bar{z}}(x_3 + 2\pi R) = -iB_{a\bar{z}}(x_3), \qquad (14.4)$$

while other components are not effected. This is exactly the action of the S-duality on the electric and magnetic fields $[\vec{E} \pm i\vec{B}](x_3 + 2\pi R) = \pm i[\vec{E} \pm i\vec{B}](x_3) = [-\vec{B} \pm i\vec{E}](x_3)$.

We recall [40] that $\mathbf{g} \in SL(2, \mathbb{Z})$ acts nontrivially on the supercharges

$$\mathbf{g}: Q_{a\alpha} \to \left(\frac{c\tau+d}{|c\tau+d|}\right)^{-\frac{1}{2}} Q_{a\alpha} = e^{-\frac{i\pi}{4}} Q_{a\alpha} \,. \tag{14.5}$$

In order to get a supersymmetric theory, we therefore need to supplement the S-twist with an R-symmetry twist so that the phase in (14.5) is canceled and supersymmetry restored. However, only the S-twist affects the gauge field, and only the R-twist affects the scalars and fermions. So we will not need to consider details of the R-twist for our calculation.

This setup has been studied in detail on a flat space (with $M_3 = T^2 \times \mathbb{R}$) in [1] as well as Part III of this dissertation and it was found to correspond at low-energy to the pure Chern-Simons theory at level k = 2. Here, I extend that study of the ground states to include the first order quantum effect arising as a consequence of curvature on M_4 . This effect was calculated to one-loop and presented in §15.2. It turns out that this contribution agrees with the lowest order effect of the gravitational Chern-Simons term as in §15.1.

Chapter 15 Calculation

In this Chapter I present details of the calculations that compare the low-energy effective gravitational Chern-Simons theory with the one-loop correlation function in (2, 0) theory twisted as described above. A reader that does not wish to get bogged down in these can find a summary of the results in Chapter 16.

15.1 Low-Energy Expansion of Gravitational Chern-Simons Theory

Although the Chern-Simons term was first introduced in three-dimensional gauge field and gravitational models [78], it can also deform physical theories in fourdimensional space-time, where it modifies conventional kinematics and dynamics in a Lorentz violating fashion. This possibility has been investigated for Maxwell electrodynamics [79] and in the present calculation we study a similar deformation of Einstein's general relativistic gravity theory.

On a four-manifold M_4 gravitational Chern-Simons Lagrangian density is given by the Pontryagin density which is the divergence of topological Chern-Simons current in analogy with the Yang-Mills case

$$S_{GSC} = \frac{1}{2\pi} \int_{M_4} \operatorname{tr}(R \wedge R) \tag{15.1}$$

where R is the curvature two-form. We want to expand this Lagrangian in terms of a small deviation h from the flat metric on a Minkowski manifold M_4

$$g_{\mu\nu} = \eta_{\mu\nu} + h_{\mu\nu} \,. \tag{15.2}$$

For this low-energy approximation we will be keeping just the lowest order in h so that, for example, Christoffel symbol to the first order in h looks like

$$\Gamma^{\nu}{}_{\mu\lambda} = \frac{1}{2} \delta^{\nu\sigma} (\partial_{\mu} h_{\lambda\sigma} + \partial_{\lambda} h_{\sigma\mu} - \partial_{\sigma} h_{\lambda\mu}) \,. \tag{15.3}$$

To complete this calculation we are switching to the non-coordinate basis

$$e_a = e_a{}^{\mu}\partial_{\mu} \tag{15.4}$$

where $e_a{}^{\mu}$ are valued in $GL(4, \mathbb{R})$ and ∂_{μ} are the coordinate basis. We start by writing the matrix valued connection one-form as

$$\omega^a{}_b = e^a{}_\nu (\partial_\mu e_b{}^\nu + e_b{}^\lambda \Gamma^\nu{}_{\mu\lambda}) dx^\mu \,. \tag{15.5}$$

When expanded to the first order in h the connection looks like

$$\omega^a{}_b = \frac{1}{2} \delta^{a\sigma} \delta_b{}^\lambda \partial_{[\lambda} h_{\sigma]\mu} dx^\mu \,. \tag{15.6}$$

And the curvature two-form

$$R^a{}_b = d\omega^a{}_b + \omega^a{}_c \wedge \omega^c{}_b \tag{15.7}$$

to the first order in h is given by

$$R^{a}{}_{b} = \frac{1}{2} \delta^{a\sigma} \delta_{b}{}^{\lambda} \partial_{\nu} \partial_{[\lambda} h_{\sigma]\mu} dx^{\nu} \wedge dx^{\mu}.$$
(15.8)

The gravitational Chern-Simons Lagrangian density to the lowest order in the deviation from flat metric h is

$$\mathcal{L}_{GSC} = \frac{1}{2\pi} R^a{}_b \wedge R^b{}_a = \frac{1}{4\pi} \epsilon^{\mu\nu\alpha\beta} \partial_\alpha \partial_{[\sigma} h_{\tau]\beta} \partial_\nu \partial^\sigma h^\tau_\mu dV$$
(15.9)

which is in agreement with [80].

For a plane monochromatic gravitational waves with wave-vector p_{α} satisfying $p_{\alpha}p^{\alpha} = 0$ we can rewrite $\partial_{\alpha}\partial_{\sigma}h_{\tau\beta} = -p_{\alpha}p_{\sigma}h_{\tau\beta}$ and the above approximation becomes

$$\mathcal{L}_{GSC} = \frac{1}{2\pi} R^a{}_b \wedge R^b{}_a = \frac{1}{4\pi} \epsilon^{\mu\nu\alpha\beta} p_\alpha p_{[\sigma} h_{\tau]\beta} p_\nu p^\sigma h^\tau_\mu.$$
(15.10)

This is the result we are comparing with the one-loop amplitude calculation of the twisted abelian gauge theory at the self-dual coupling constant $\tau = i$, presented in the next section.

15.2 Field theory loop calculation

As described in Chapter 14, we start with the six-dimensional (2, 0) theory compactified first on a two-torus down to four dimensional $\mathcal{N} = 4$ SYM and then on a circle with an S-duality twist at self dual SYM coupling $\tau = i$ so that the twist is a symmetry of the theory. We are shrinking the T^2 in directions x_4 and x_5 to zero size and retaining only the states of zero compact momentum by setting $p_4 = p_5 = 0$. However, momentum along the S^1 direction is not vanishing, but quantized in an unusual way. As a result of the geometrical twist introduced along the circle the Kaluza-Klein modes along S^1 are shifted by $\pm 1/4$.

This effect of the S-twist can be seen as follows. Recall the action of the twist on the *B*-field two-form (14.4) and let us first focus on B_{az} (with $a = 0, \ldots, 3$, while $z = x_4 + ix_5$ and $x_3 \in [0, 2\pi R]$) which S-twist multiplies by *i*. Starting at $x_3 = 0$ and making a full circle in the x_3 -direction shifts momentum eigenstates $e^{ip_3x_3/R}$ of B_{az} by $e^{2\pi i p_3}$. But they must be the same at $x_3 = 0$ after the S-twist by $i = e^{i\pi/2}$. It follows that $p_3 \in \mathbb{Z} - 1/4$ for the modes of B_{az} . Equivalently, the Kaluza-Klein modes of $B_{a\bar{z}}$ are $p_3 \in \mathbb{Z} + 1/4$ instead as S-twist multiplies $B_{a\bar{z}}$ by -i.

The bosonic field content of the (2, 0) consists of five scalars, corresponding to five broken translations, and a self dual two-form potential, B, with self-dual field strength H = dB. There are also the Fermionic superpartners, symplectic Majorana Spinors, all together making up the tensor multiplet. Due to the self-duality constraint on H, it is not possible to write a simple action for the tensor multiplet even at the linearized level [77]. This may be achieved, however, by introducing an auxiliary scalar that effectively allows one to gauge away the anti-self-dual degrees of freedom. This is often referred to as the PST approach [75].

While it is difficult to write the full action for (2, 0) theory [76], as we are looking for the highest order effect of spacetime curvature on this theory, using an incomplete low-energy approximation of the action will suffice

$$S_0 = \int \mathcal{L}_0 = -\frac{1}{12g^2} \int d^6 x H_{\mu\nu\lambda} H^{\mu\nu\lambda}$$
(15.11)

where the three-form field strength $H_{\mu\nu\lambda}$, in terms of two-form gauge field $B_{\nu\lambda}$, is given by

$$H_{\mu\nu\lambda} = 3\partial_{[\mu}B_{\nu\lambda]}$$

Only the self-dual part $H^{(+)}_{\mu\nu\lambda}$ contributes to the (2, 0) theory action, where

$$H^{(\pm)}_{\mu\nu\lambda} = \frac{1}{2} \left(H_{\mu\nu\lambda} \pm \frac{1}{6} \epsilon_{\mu\nu\lambda\alpha\beta\gamma} H^{\alpha\beta\gamma} \right) \,. \tag{15.12}$$

After gauge-fixing, the correlation function of the three-form can be written in momentum space as

$$\left\langle H_{\mu\nu\sigma}(\mathbf{p})H^{\alpha\beta\gamma}(\mathbf{q})\right\rangle = \frac{(2\pi)^6(2g^2)\delta(\mathbf{p}+\mathbf{q})}{p^2}p_{[\mu}p^{[\alpha}\delta^{\beta}_{\nu}\delta^{\gamma]}_{\sigma]}.$$
 (15.13)

Separating the T^2 coordinates and using that $g_{\sigma z} = 0$ except for $g_{z\bar{z}} = 1$ as well as that $p_z = p_{\bar{z}} = 0$ in low-energy limit gives correlation

$$\left\langle H_{mnz}(\mathbf{p})H^{abz}(\mathbf{q})\right\rangle = \left(\frac{2}{9}\right) \frac{(2\pi)^6 (2g^2)\delta(\mathbf{p}+\mathbf{q})}{p^2} p_{[m}p^{[a}\delta^{b]}_{n]}$$
 (15.14)

and a similar result for $\langle H_{mn\bar{z}}(\mathbf{p})H^{ab\bar{z}}(\mathbf{q})\rangle$ where Greek indices run over six-dimensions $(\alpha, \beta = 0, \ldots, 5)$ while Roman indices are reserved for M_4 $(a, b = 0, \ldots, 3)$ and z is the complex coordinate on the T^2 .

To calculate first order correction to this theory coming from curvature, let the metric be given by its deviation from flat $g_{\phi\chi} = \eta_{\phi\chi} + h_{\phi\chi}$, as before, and expand the Lagrangian in powers of $h_{\phi\chi}$ as

$$\mathcal{L} = \mathcal{L}_0 + h_{\phi\chi} T^{\phi\chi} + h_{\phi\chi} h_{\dot{\phi}\dot{\chi}} T^{\phi\chi} T^{\phi\dot{\chi}} + \mathcal{O}(h^3)$$

in order to get a low-energy effective description. We will calculate one-loop amplitude



Figure 15.1: One-loop diagram computed in twisted (2, 0) theory with vertices $h_{\phi\chi}T^{\phi\chi}$ and $h_{\dot{\phi}\dot{\chi}}T^{\dot{\phi}\dot{\chi}}$ where $h_{\phi\chi}$ are deviations from flat metric and $T^{\phi\chi}$ is the low-energymomentum tensor.

contribution with interactions $h_{\phi\chi}T^{\phi\chi}$ where $T^{\phi\chi}$ is the energy-momentum tensor of the incomplete Lagrangian \mathcal{L}_0

$$T^{\phi\chi} = -\frac{1}{2g^2} H^{\chi}{}_{\nu\sigma} H^{\phi\nu\sigma} + \frac{1}{12g^2} \eta^{\phi\chi} H_{\mu\nu\sigma} H^{\mu\nu\sigma}.$$
 (15.15)

We also must separate self-dual and anti-self-dual contributions to energy-momentum tensor $T^{\phi\chi} = T^{(+)\phi\chi} + T^{(-)\phi\chi}$. Here $T^{(+)\phi\chi}$ comes from the self-dual part of the field strength, $H^{(+)}_{\mu\nu\lambda}$, only while $T^{(-)\phi\chi}$ has only $H^{(-)}_{\mu\nu\lambda}$ components. Rewritten in terms of full $H_{\mu\nu\lambda} = H^{(+)}_{\mu\nu\lambda} + H^{(-)}_{\mu\nu\lambda}$ this gives

$$T^{(\pm)\phi\chi} = \mp \frac{1}{48g^2} \epsilon^{\phi\nu\sigma\alpha\beta\gamma} H^{\chi}{}_{\nu\sigma} H_{\alpha\beta\gamma} \mp \frac{1}{48g^2} \epsilon^{\chi\nu\sigma\alpha\beta\gamma} H^{\phi}{}_{\nu\sigma} H_{\alpha\beta\gamma} - \frac{1}{4g^2} H^{\chi}{}_{\beta\gamma} H^{\phi\beta\gamma} + \frac{1}{24g^2} \eta^{\phi\chi} H^{\alpha\beta\gamma} H_{\alpha\beta\gamma}$$
(15.16)

where just $T^{(+)\phi\chi}$ shows up in our calculation as (2, 0) theory has only a self-dual three-form field strength.

The correlation function of the energy-momentum tensor vanishes in the flat spacetime. But we expect even the one-loop amplitude contribution

$$\int \frac{d^4 \mathbf{p}}{(2\pi)^4} \frac{d^4 \mathbf{q}}{(2\pi)^4} h_{\phi \chi} h_{\dot{\phi} \dot{\chi}} \left\langle T^{(+)\phi \chi}(\mathbf{p}) T^{(+)\dot{\phi} \dot{\chi}}(\mathbf{q}) \right\rangle$$
(15.17)

to be non-zero in our setup. We find that, after the S-twist, a non-vanishing effect of $h_{\phi\chi}$ is a consequence of the modification in the treatment of loop momentum. In particular, the momentum component in the S^1 direction x_3 only takes discrete values which, instead of being simply integers, are shifted by the S-twist as $p_3 \in \mathbb{Z} \pm 1/4$.

The non vanishing term comes from replacing

$$\int \frac{d^4 \mathbf{p}}{(2\pi)^4} = \frac{1}{R} \int \frac{d^3 \mathbf{p}}{(2\pi)^3} \sum_{n \pm \frac{1}{4}}$$

and after regularizing this sum. This indeed is the effect of the S-twist itself which results in the $\pm 1/4$ shifts. In fact, this contribution is in exact agreement with the gravitational Chern-Simons term (15.10).

Chapter 16

Interpretation

It was shown in [1] that the low-energy limit of $\mathcal{N} = 4$ SYM theory on a flat fourdimensional manifold compactified on a shrinking circle with an S-duality twist gives a topological field theory, whose correlation functions do not depend on the metric of the spacetime. However, we know that Chern-Simons theory on a manifold with curvature is not quite invariant but requires a gravitational Chern-Simons counterterm [30]. Also, [31] shows that the partition function of Chern-Simons theory is not a modular invariant but transforms as a modular form with weight quadratic in curvature. These results are suggesting that an S-duality twist should be accompanied with a gravitational Chern-Simons term.

To check this, we geometrically realize S-duality in SYM as a twist of (2, 0) theory of coincident M5-branes compactified on a two-torus. We find that the effective lowenergy contribution is in agreement with the $\mathcal{O}(h^2)$ contribution of the gravitational Chern-Simons term

$$\mathcal{L}_{GSC} = \frac{1}{2\pi} R^a{}_b \wedge R^b{}_a = \frac{1}{4\pi} \epsilon^{\mu\nu\alpha\beta} p_\alpha p_{[\sigma} h_{\tau]\beta} p_\nu p^\sigma h^\tau_\mu.$$
(16.1)

This is the first step toward proving the conjecture that adding S-duality twist to $\mathcal{N} = 4$ super Yang-Mills gauge theory gives Chern-Simons theory with gravitational Chern-Simons contributions.

Bibliography

- [1] Ori J. Ganor, Yoon Pyo Hong, and H.S. Tan. Ground States of S-duality Twisted N=4 Super Yang-Mills Theory. *JHEP*, 1103:099, 2011.
- [2] Ichiro Satake. On the compactification of the Siegel space. J. Indian Math. Soc. (N.S.), 20:259–281, 1956.
- [3] William P Thurston. The geometry and topology of three-manifolds. *Geometry* And Topology, 1(March):217–2269, 2002.
- [4] E. Ghys, A. Haefliger, and A. Verjovsky, editors. Group theory from a geometrical viewpoint. Proceedings of a workshop, held at the International Centre for Theoretical Physics in Trieste, Italy, 26 March to 6 April 1990. Reprint of the 1991 ed. Singapore: World Scientific, reprint of the 1991 ed. edition, 1993.
- [5] Lance J. Dixon, Jeffrey A. Harvey, C. Vafa, and Edward Witten. Strings on Orbifolds. 2. Nucl. Phys., B274:285–314, 1986.
- [6] Lance J. Dixon, Jeffrey A. Harvey, C. Vafa, and Edward Witten. Strings on Orbifolds. Nucl. Phys., B261:678–686, 1985.
- [7] M.A. Melvin. Pure magnetic and electric geons. *Phys.Lett.*, 8:65–70, 1964.
- [8] J.G. Russo and Arkady A. Tseytlin. Magnetic flux tube models in superstring theory. Nucl. Phys., B461:131–154, 1996.
- [9] Danny Dhokarh, Sheikh Shajidul Haque, and Akikazu Hashimoto. Melvin Twists of global AdS(5) x S(5) and their Non-Commutative Field Theory Dual. JHEP, 0808:084, 2008.
- [10] Louise Dolan and Chiara R. Nappi. Noncommutativity in a time dependent background. *Phys.Lett.*, B551:369–377, 2003.
- [11] Akikazu Hashimoto and Keith Thomas. Dualities, twists, and gauge theories with non-constant non-commutativity. JHEP, 0501:033, 2005.
- [12] Akikazu Hashimoto and Keith Thomas. Non-commutative Gauge Theory on D-branes in Melvin Universes. JHEP, 0601:083, 2006.

- [13] Aaron Bergman and Ori J. Ganor. Dipoles, twists and noncommutative gauge theory. JHEP, 0010:018, 2000.
- [14] Aaron Bergman, Keshav Dasgupta, Ori J. Ganor, Joanna L. Karczmarek, and Govindan Rajesh. Nonlocal field theories and their gravity duals. *Phys.Rev.*, D65:066005, 2002.
- [15] Ori J. Ganor and Uday Varadarajan. Nonlocal effects on D-branes in plane wave backgrounds. *JHEP*, 0211:051, 2002.
- [16] Oleg Lunin and Juan Martin Maldacena. Deforming field theories with U(1) x U(1) global symmetry and their gravity duals. JHEP, 0505:033, 2005.
- [17] Ori J. Ganor. A New Lorentz violating nonlocal field theory from string-theory. *Phys. Rev.*, D75:025002, 2007.
- [18] Ori J. Ganor, Akikazu Hashimoto, Sharon Jue, Bom Soo Kim, and Anthony Ndirango. Aspects of Puff Field Theory. JHEP, 0708:035, 2007.
- [19] H. A. Kramers and G. H. Wannier. Statistics of the two-dimensional ferromagnet. part i. *Phys. Rev.*, 60:252–262, Aug 1941.
- [20] P. Goddard, J. Nuyts, and David I. Olive. Gauge Theories and Magnetic Charge. Nucl. Phys., B125:1, 1977.
- [21] C. Montonen and David I. Olive. Magnetic Monopoles as Gauge Particles? *Phys.Lett.*, B72:117, 1977.
- [22] Edward Witten and David I. Olive. Supersymmetry Algebras That Include Topological Charges. *Phys.Lett.*, B78:97, 1978.
- [23] John L. Cardy and Eliezer Rabinovici. Phase Structure of Z(p) Models in the Presence of a Theta Parameter. Nucl. Phys., B205:1, 1982.
- [24] John L. Cardy. Duality and the Theta Parameter in Abelian Lattice Models. Nucl. Phys., B205:17, 1982.
- [25] Alfred D. Shapere and Frank Wilczek. Selfdual Models with Theta Terms. Nucl. Phys., B320:669, 1989.
- [26] Ori J. Ganor and Yoon Pyo Hong. Selfduality and Chern-Simons Theory. 2008.
- [27] Ori J. Ganor, Yoon Pyo Hong, Ruza Markov, and Hai Siong Tan. Static Charges in the Low-Energy Theory of the S-Duality Twist. 2012. 73 pages, two paragraphs added, one to the introduction and one to the discussion.

- [28] J. Wess and B. Zumino. Consequences of anomalous ward identities. *Physics Letters B*, 37(1):95 97, 1971.
- [29] Edward Witten. Global Aspects of Current Algebra. Nucl. Phys., B223:422–432, 1983.
- [30] Edward Witten. Quantum Field Theory and the Jones Polynomial. Commun. Math. Phys., 121:351, 1989.
- [31] Edward Witten. On S duality in Abelian gauge theory. Selecta Math., 1:383, 1995.
- [32] K. Becker, M. Becker, and J.H. Schwarz. String theory and M-theory: A modern introduction. 2007.
- [33] Hugh Osborn. Topological Charges for N=4 Supersymmetric Gauge Theories and Monopoles of Spin 1. *Phys.Lett.*, B83:321, 1979.
- [34] Ashoke Sen. Dyon monopole bound states, selfdual harmonic forms on the multi - monopole moduli space, and SL(2,Z) invariance in string theory. *Phys.Lett.*, B329:217–221, 1994.
- [35] Cumrun Vafa and Edward Witten. A Strong coupling test of S duality. Nucl. Phys., B431:3–77, 1994.
- [36] M. Bershadsky, A. Johansen, V. Sadov, and C. Vafa. Topological reduction of 4-d SYM to 2-d sigma models. *Nucl. Phys.*, B448:166–186, 1995.
- [37] Jeffrey A. Harvey, Gregory W. Moore, and Andrew Strominger. Reducing S duality to T duality. *Phys.Rev.*, D52:7161–7167, 1995.
- [38] Edward Witten. Some comments on string dynamics. 1995.
- [39] Cumrun Vafa. Geometric origin of Montonen-Olive duality. Adv. Theor. Math. Phys., 1:158–166, 1998.
- [40] Anton Kapustin and Edward Witten. Electric-Magnetic Duality And The Geometric Langlands Program. 2006.
- [41] Davide Gaiotto and Edward Witten. S-Duality of Boundary Conditions In N=4 Super Yang-Mills Theory. 2008.
- [42] Davide Gaiotto and Edward Witten. Supersymmetric Boundary Conditions in N=4 Super Yang-Mills Theory. 2008.
- [43] Kenneth A. Intriligator and N. Seiberg. Mirror symmetry in three-dimensional gauge theories. *Phys.Lett.*, B387:513–519, 1996.

- [44] Yuji Terashima and Masahito Yamazaki. SL(2,R) Chern-Simons, Liouville, and Gauge Theory on Duality Walls. *JHEP*, 1108:135, 2011. 53+1 pages, 14 figures/ v2: typos corrected, references added.
- [45] Tudor Dimofte and Sergei Gukov. Chern-Simons Theory and S-duality. 2011.
- [46] Tudor Dimofte, Davide Gaiotto, and Sergei Gukov. Gauge Theories Labelled by Three-Manifolds. 2011.
- [47] Jonathan Bagger and Neil Lambert. Modeling Multiple M2's. Phys.Rev., D75:045020, 2007. Dedicated to the Memory of Andrew Chamblin.
- [48] Jonathan Bagger and Neil Lambert. Comments on multiple M2-branes. JHEP, 0802:105, 2008.
- [49] Ofer Aharony, Oren Bergman, Daniel Louis Jafferis, and Juan Maldacena. N=6 superconformal Chern-Simons-matter theories, M2-branes and their gravity duals. JHEP, 0810:091, 2008.
- [50] Joel Scherk and John H. Schwarz. How to Get Masses from Extra Dimensions. Nucl. Phys., B153:61–88, 1979.
- [51] Soo-Jong Rey and Jung-Tay Yee. Macroscopic strings as heavy quarks in large N gauge theory and anti-de Sitter supergravity. *Eur.Phys.J.*, C22:379–394, 2001.
- [52] Juan Martin Maldacena. Wilson loops in large N field theories. *Phys.Rev.Lett.*, 80:4859–4862, 1998.
- [53] Edward Witten. Solutions of four-dimensional field theories via M theory. Nucl. Phys., B500:3–42, 1997.
- [54] Joseph Polchinski and Edward Witten. Evidence for heterotic type I string duality. Nucl. Phys., B460:525–540, 1996.
- [55] Cumrun Vafa. Evidence for F theory. Nucl. Phys., B469:403–418, 1996.
- [56] Erik P. Verlinde. Fusion Rules and Modular Transformations in 2D Conformal Field Theory. Nucl. Phys., B300:360, 1988.
- [57] J.M.F. Labastida, P.M. Llatas, and A.V. Ramallo. Knot operators in Chern-Simons gauge theory. *Nucl. Phys.*, B348:651–692, 1991.
- [58] Anton Kapustin. Wilson-'t Hooft operators in four-dimensional gauge theories and S-duality. *Phys. Rev.*, D74:025005, 2006.

- [59] Gregory W. Moore and Nathan Seiberg. Taming the Conformal Zoo. Phys.Lett., B220:422, 1989.
- [60] Amihay Hanany and Edward Witten. Type IIB superstrings, BPS monopoles, and three-dimensional gauge dynamics. *Nucl. Phys.*, B492:152–190, 1997.
- [61] Sergio Cecotti, Clay Cordova, and Cumrun Vafa. Braids, Walls, and Mirrors. 2011.
- [62] Sergei Gukov, Albert S. Schwarz, and Cumrun Vafa. Khovanov-Rozansky homology and topological strings. *Lett. Math. Phys.*, 74:53–74, 2005.
- [63] D.E. Diaconescu, V. Shende, and C. Vafa. Large N duality, lagrangian cycles, and algebraic knots. 2011.
- [64] Katrin Becker, Melanie Becker, and Andrew Strominger. Five-branes, membranes and nonperturbative string theory, volume B456. 1995.
- [65] Ofer Aharony, Amihay Hanany, and Barak Kol. Webs of (p,q) five-branes, five-dimensional field theories and grid diagrams. JHEP, 9801:002, 1998.
- [66] Oren Bergman and Barak Kol. String webs and 1/4 BPS monopoles. Nucl. Phys., B536:149–174, 1998.
- [67] Hirosi Ooguri and Cumrun Vafa. Knot invariants and topological strings. Nucl. Phys., B577:419–438, 2000.
- [68] Micha Berkooz, Michael R. Douglas, and Robert G. Leigh. Branes intersecting at angles. Nucl. Phys., B480:265–278, 1996.
- [69] Shamit Kachru and John McGreevy. Supersymmetric three cycles and supersymmetry breaking. *Phys. Rev.*, D61:026001, 2000.
- [70] M. Mihailescu, I.Y. Park, and Tuan A. Tran. D-branes as solitons of an N=1, D = 10 noncommutative gauge theory. *Phys.Rev.*, D64:046006, 2001.
- [71] Edward Witten. BPS Bound states of D0 D6 and D0 D8 systems in a B field. JHEP, 0204:012, 2002.
- [72] Nathan Seiberg. Notes on theories with 16 supercharges. Nucl. Phys. Proc. Suppl., 67:158–171, 1998.
- [73] Andrew Strominger. Open p-branes. *Phys.Lett.*, B383:44–47, 1996.
- [74] Edward Witten. Five-branes and M-theory on an orbifold. Nucl. Phys., B463:383–397, 1996.

- [75] Paolo Pasti, Dmitri P. Sorokin, and Mario Tonin. Covariant action for a D = 11 five-brane with the chiral field. *Phys.Lett.*, B398:41–46, 1997.
- [76] Mina Aganagic, Jaemo Park, Costin Popescu, and John H. Schwarz. World volume action of the M theory five-brane. *Nucl. Phys.*, B496:191–214, 1997.
- [77] David S. Berman. M-theory branes and their interactions. *Phys.Rept.*, 456:89– 126, 2008.
- [78] Stanley Deser, R. Jackiw, and S. Templeton. Topologically Massive Gauge Theories. Annals Phys., 140:372–411, 1982.
- [79] Sean M. Carroll, George B. Field, and Roman Jackiw. Limits on a Lorentz and Parity Violating Modification of Electrodynamics. *Phys. Rev.*, D41:1231, 1990.
- [80] R. Jackiw and S.Y. Pi. Chern-Simons modification of general relativity. *Phys.Rev.*, D68:104012, 2003.
- [81] Andreas Gustavsson. Algebraic structures on parallel M2-branes. Nucl. Phys., B811:66-76, 2009.
- [82] Jonathan Bagger and Neil Lambert. Gauge symmetry and supersymmetry of multiple M2-branes. *Phys. Rev.*, D77:065008, 2008.
- [83] V.G. Kac. A Sketch of Lie Superalgebra Theory. Commun.Math.Phys., 53:31– 64, 1977.
- [84] W. Nahm. Supersymmetries and their Representations. Nucl. Phys., B135:149, 1978.
- [85] Jeong-Hyuck Park. Superconformal symmetry in three-dimensions. J.Math.Phys., 41:7129–7161, 2000.
- [86] Michael B. Green, J.H. Schwarz, and Edward Witten. Superstring Theory. Vol. 1: Introduction. 1987.
- [87] Michael B. Green, J.H. Schwarz, and Edward Witten. Superstring Theory. Vol. 2: Loop Amplitudes, Anomalies And Phenomenology. 1987.
- [88] Daniel S. Freed. Remarks on Chern-Simons Theory. 2008. dedicated to MSRI on its 25th anniversary.
- [89] Hitoshi Murayama. Explicit Quantization Of The Chern-simons Action. Z.Phys., C48:79–88, 1990.
- [90] Edward Witten. Bound states of strings and p-branes. Nucl. Phys., B460:335– 350, 1996.

- [91] Michael R. Douglas. Branes within branes. 1995.
- [92] Edward Witten. Branes, Instantons, And Taub-NUT Spaces. JHEP, 0906:067, 2009.
- [93] V.G. Kac and M. Wakimoto. Modular and conformal invariance constraints in representation theory of affine algebras. Adv. Math., 70:156, 1988.
- [94] Atish Dabholkar and Jeffrey A. Harvey. String islands. JHEP, 9902:006, 1999.
- [95] Ori J. Ganor. U duality twists and possible phase transitions in (2+1)dimensions supergravity. Nucl. Phys., B549:145–180, 1999.
- [96] Jessie Shelton, Washington Taylor, and Brian Wecht. Nongeometric flux compactifications. JHEP, 0510:085, 2005.
- [97] David Vegh and John McGreevy. Semi-Flatland. JHEP, 0810:068, 2008.
- [98] C.M. Hull and P.K. Townsend. Unity of superstring dualities. Nucl. Phys., B438:109–137, 1995.
- [99] Edward Witten. String theory dynamics in various dimensions. Nucl. Phys., B443:85–126, 1995.
- [100] Atish Dabholkar and Jeffrey A. Harvey. Nonrenormalization of the Superstring Tension. *Phys. Rev. Lett.*, 63:478, 1989.
- [101] Krzysztof Gawedzki and Antti Kupiainen. SU(2) Chern-Simons Theory at Genus Zero. Commun.Math.Phys., 135:531–546, 1991.
- [102] Fernando Falceto and Krzysztof Gawedzki. Chern-Simons states at genus one. Commun. Math. Phys., 159:549–580, 1994.
- [103] (Ed.) Deligne, P., (Ed.) Etingof, P., (Ed.) Freed, D.S., (Ed.) Jeffrey, L.C., (Ed.) Kazhdan, D., et al. Quantum fields and strings: A course for mathematicians. Vol. 1, 2. 1999.
- [104] Ofer Aharony, Jacob Sonnenschein, and Shimon Yankielowicz. Interactions of strings and D-branes from M theory. Nucl. Phys., B474:309–322, 1996.
- [105] Csaba Csaki, Joshua Erlich, and Christophe Grojean. Gravitational Lorentz violations and adjustment of the cosmological constant in asymmetrically warped space-times. *Nucl. Phys.*, B604:312–342, 2001.
- [106] Yeuk-Kwan E. Cheung and Zheng Yin. Anomalies, branes, and currents. Nucl.Phys., B517:69–91, 1998.

- [107] Michael B. Green, Jeffrey A. Harvey, and Gregory W. Moore. I-brane inflow and anomalous couplings on d-branes. *Class. Quant. Grav.*, 14:47–52, 1997.
- [108] Robert C. Myers. Dielectric branes. JHEP, 9912:022, 1999.
- [109] J. Polchinski. String theory. Vol. 1: An introduction to the bosonic string. 1998.
- [110] J. Polchinski. String theory. Vol. 2: Superstring theory and beyond. 1998.